'VIBRATIONS OF THICK PLATES AND SHELLS''

by MICHAEL D J BRIDLE, B.Sc

Thesis submitted to the University of Nottingham for the degree of Doctor of Philosophy, October 1973.

LIBRARY HSILING

IMAGING SERVICES NORTH

Boston Spa, Wetherby West Yorkshire, LS23 7BQ www.bl.uk

BEST COPY AVAILABLE.

VARIABLE PRINT QUALITY

Illus	trations
Table	n an an an an an an an an ann an ann an
Abstr	\mathbf{act} , the state of the st
HAPTER 1.	INTRODUCTI CN
1.1.	Problem
1.2.	Shell Theory
1.3.	Shell Vibrations
1.4.	Outline of Thesis
HAFTER 2.	SHELL THEORY
2.1.	Introduction
2.2.	General Theory
2.3.	Fundamental Shell Equations
2.4.	Displacement Functions
2.5.	Cylindrical Shell
2.6.	Cylindrical Shell - Displacement Functions
2.7.	Twisted Plate
2.8.	Twisted Plate - Displacement Functions
HAPTER 3.	NUMERICAL ANALYSIS AND COMPUTING
3.1.	Introduction
3.2.	Numerical Analysis
3.3.	Computer Program - General Outline
3.4.	Strain and Kinetic Energy Matrices
3.5.	Boundary Conditions
3.6.	Solution of the Eigenvalue Problem
3.7.	Discussion.

•

•

.

CHAPTER 4.	RUSULTS AND CONCLUSIONS	Paga
4.1.	Introduction	77
4.2.	Convergence	78
4.3.	Comparison of Theoretical with Experimental Results	7 9
4.4.	Twisted Plate Results	82
4.5.	Cylindrical Shell Results	85.
4.6.	Conclusions	68
BIBLIOGRAFH	Ω ·····	112
ACKNOWLEDGE	LENTS	116
APPENDIX 1.	Basis of the Variational Nethod	117
APPENDIX 2.	Thick Twisted Plate Computer Program	122
APPENDIX 3.	Thick Twisted Flate Energy Data	136
APPENDIX 4.	Reduction Parts of Thin Cylinder Computer Program	144
APPENDIX 5.	Input Data for Computer Programs	149
APPENDIX 6.	Library Procedures used in Computer Programs	150
• • • •		

11

ILLUSTRATIONS

ILLUSTRATIC	<u>NS</u>	Paga
Fig 1.	Cylindrical Shell	20
Fig 2.	Twisted Plate	30
Fig 3.	Program Flow Diagram	49
Fig 4.	SESET Flow Diagram	53
Fig 5.	EVENAB Flow Diegrom	59
Fig 6.	BCMAT Flow Diagrom	63
Fig 7.	BCRED Flow Diegram	68
Fig 8.	SOLVE Flow Diegram	71
Fig 9.	Lindberg-Olsen Fan blade	80
Fig 10.	Cylindrical Shell Convergence Tests, Thin Shell Theory	99
Fig 11.	Twisted Plate Convergence Tests, Thick Shell Theory	100
Fig 12.	Lindberg-Olsen Fan Blade Nodes	101
Fig 13.	Carnegie Beam Nodes and a set find the set of the set o	102
Fig 14.	Dawson Beam Modes	103
Fig 15.	Nodes for 5:1 L/W Ratio Twisted Plate	104
Fig 16.	Effect of Twist on Modes of 1:1 L/W Batio Twisted Plate, Thin Shell Theory	105
Fig 17.	Effect of Twist on Modes of 1:1 L/W Ratio Twisted Plate, Thick Shell Theory	106
Fig 18.	Effect of Thickness on Modes of 1:1 L/W Ratio Twisted Plate, Thin Shell Theory	107
Fig 19.	Effect of Thickness on Modes of 1:1 L/W Ratio Twisted Plate, Thin Shell Theory	108
Fig 20.	Comparison of Thin and Thick Cylindrical Shell Theory. Effect of Thickness on Modes for 5:1 L/W Ratio Shell.	109
Fig 21.	Comparison of Thin and Thick Cylindrical Shell Theory. Modes for 1:1 L/W Ratio Shell	110
Fig 22.	Effect of Thickness on Modes of 1:1 L/W Ratio Cylindrical Sholl. Thin Shell Theory	111

TABLES Page Computer Times 1. 50 2. Thin Cylinder Convergence Tests 88 3. Thick Twisted Plate Convergence Tests 89 Lindberg-Olsen Cylinder Comparison 4. 90 Carnegie Twisted Beam Comparison 5. 91 6. Dawson Twisted Beam Comparison 92. 7. Effect of Twist on Twisted Plate Frequencies 5:1 L/W 93 Ratio Shell 8. Effect of Twist on Twisted Plate Frequencies 1:1 L/W 94 Ratio Shell Effect of Thickness on Twisted Plate Frequencies 5:1 9. 95 L/W Ratio Shell 10. Effect of Thickness on Twisted Plate Frequencies 1:1 L/W 96 Ratio Shell 11. Effect of Thickness on Cylindrical Shell Frequencies 97 5:1 L/W Ratio Shell Effect of Thickness on Cylindrical Shell Frequencies 12. 98 1:1 L/W Ratio Shell

J.V.

VIBRATIONS OF THICK PLATES AND SHELLS

M D J BRIDLE B.Sc.

ABSTRACT:

Using an asymptotic series approach, a thick shell theory is proposed for doubly curved shells with variable thickness. This theory includes the effects of transverse shear stresses and rotatory inertia. The displacement functions are designed to give non-zero transverse shear stresses internal to the shell, which satisfy the stress-free boundary conditions on the upper and lower surface. Use of the stress-free conditions makes the displacement functions, which vary through the thickness of the shell, dependent only on the middle surface displacements. This theory is applied to the twisted plate. A similar approach is applied to the cylindrical shell, but the effects of transverse normal stress are also included.

The theory is applied to the problem of free vibrations of shells clamped along one edge with the other three edges free. The results obtained are compared with practical and theoratical results of other researchers, and with those obtained from thin shell theory. The twisted plate results show the answers that are expected from a thick shell theory, in that it predicts lower frequencies than thin shell theory for modes in which the wavelength/ thickness ratio is less than ten.

The results for the cylindrical shell show that the inclusion of transverse normal stress to the order essumed is not warranted.

The numerical techniques used for the solution of the free vibration problem are based on variational methods in which the Hamiltonian for the shell is minimized, subject to the constraints of the displacement boundary conditions.

CHAPTER 1 .- INTRODUCTION

1.1. Problem

The subject of this thesis is thick shell vibrations. The need for a theory of thick shells to predict natural frequencies is highlighted by the failures of turbine and compressor blading in such projects as the Queen Elizabeth II and the RE211 Jet engine. For certain blading, adequate results can be obtained using twisted beam or thin shell theory, dependent on the geometry of the blade. However, many modern blades are of such a shape that these theories are inadequate. A thick shell theory is required as a half-way stage between thin shell theory, and a full three dimensional analysis.

ì ...

The major difference between thin and thick shall theories is the inclusion of transverse shear stresses in the latter case. These are neglected in thin shall theory, which therefore fails to give sufficiently accurate results as soon as these effects become important. In general the transverse shear modes do not appear by themselves, but are coupled with other modes. This coupling generally effects the frequencies of vibration for modes in which the wavelength to thickness ratio is less than ten.

1.2. Shell Theory

The basic property of shell theory is that one dimension, the thickness, is small compared with the other two. In shell theory this property is used to reduce the problem to two rather than three dimensions. To achieve this, certain approximations have to be made. The different approximations used, and the variation in the rigour applied in the analysis lead to the wide variety of shell theories to be found in the literature. Further references to and discussion of these theories can be found in Love (1), Nach (2), Flügge (3), Novechilov (4), Goldenveiser (5), Naghdi (6), Koiter (7), and on vibrations of shells in Kalnins (8), Gros and Forsberg (9) and Hu (10).

In this section, an outline of the basic approximations is given, and the differences between thin and thick shell theories shown. Traditionally thin shell theories are based on the Kirchoff-Love assumptions:

- a. Points which lie on one and the same normal, to the undeformed middle surface, also lie on one and the same normal to the deformed middle surface.
- b. The effect of normal stress on surfaces parallel to the middle surface may be neglected in the stress-strain relation.
- c. The displacements in the direction of the normal to the middle surface are approximately equal for all points on the same normal.

Assumptions a. and c. imply the displacement functions for the shell take the following form

 $U_{1} = U_{1}^{\circ}(\Theta_{1}, \Theta_{2}) + \Theta_{3} U_{1}^{\prime}(\Theta_{1}, \Theta_{2})$ $U_{2} = U_{2}^{\circ}(\Theta_{1}, \Theta_{2}) + \Theta_{3} U_{2}^{\prime}(\Theta_{1}, \Theta_{2})$ $U_{3} = U_{3}^{\circ}(\Theta_{1}, \Theta_{2})$

where U_1' and U_2' are given in terms of the middle surface displacements U_1'' , U_2'' and U_3'' by the equations for the transverse shear strain implied by assumption a.

Assumption b. effects the stress-strain relations; the fact that the transverse normal stress \mathcal{T}_{33}^{33} is zero is used to obtain an expression for the transverse normal strain \mathcal{S}_{33} which is substituted in the other stress-strain relations. This gives rise to one of the contradictions of thin shell theory in that assumption c. implies that $\mathcal{S}_{33} = O$

If assumption a. is relaxed, so that points on normals to the undeformed middle surfaces remain straight lines during deformation, but are not necessarily normal to the deformed middle surface, then the displacement functions are of the form 1.2.1. In this case, however, equations 1.2.2. do not apply, and so U_1' and U_2' are not dependent on the middle surface displacements. In this case substitution of these displacement functions in the strain-

2.

1.2.1.

displacement and stress-strain relations gives non-zero transverse shear stresses and strains.

To include the effects of transverse normal stress, \mathcal{T}^{33} , the expansion for \mathcal{U}_3 can be expanded to include terms of $\mathcal{O}(\mathcal{O}_3)$ or $\mathcal{O}(\mathcal{O}_3^2)$ i.e.

 $U_{3} = U_{3}^{\circ}(\Theta_{1}, \Theta_{2}) + \Theta_{3} U_{3}^{\prime}(\Theta_{1}, \Theta_{2})$ or $U_{3} = U_{3}^{\circ}(\Theta_{1}, \Theta_{2}) + \Theta_{3} U_{3}^{\prime}(\Theta_{1}, \Theta_{2}) + \frac{\Theta_{3}^{2}}{2} U_{3}^{2}(\Theta_{1}, \Theta_{2})$ 1.2.3.

The second of these cases has been considered by Reissner (11) and Naghci (12).

In thick shell theory substitution of displacement functions 1.2.1. into the standard three dimensional strain-displacement and the stress-strain relations modified by the transverse normal stress assumption discussed earlier, gives transverse shear stresses which are constant through the thickness of the shell. For free vibration problems, the boundary conditions on the upper and lower surfaces of the shell imply that these stresses are zero on these boundaries. Hence direct application of this boundary condition would lead to thin shell theory. To surnount this difficulty changes have to be made to the standard three dimensional equations. This problem has been approached in two ways. In the first the stress-strain relations are modified by a shear constant x^2 so that

 $T^{\alpha_3} = K^2 G \chi_3 \qquad (\alpha = 1, 2)$

where G is the shear modulus of the material. This is the approach usually used in thick shell vibration problems. K^2 is evaluated so that the contribution of the transverse shear terms to the strain energy, matches the contribution of the true stress field. It is evaluated by matching frequencies obtained by the shell theory with those obtained by three dimensional theories. These methods are discussed by Mindlin (13) for plates and Herrmann and Mirsky (14) for shells. For the use of this technique in shell vibration problems see Herrmann and Mirsky (14), Tsui (15) and Zeinkiewicz et al (16, 17). The other approach, which is often used in forming equations for shell theory, but has apparently not been applied to evaluate natural frequencies of shells is that presented by Naghdi (6). He assumes that the transverse stresses and strains are of the form:

 $\chi_{\alpha \langle 3} = \chi_{\alpha \langle 3}^{\circ} \left(\Theta_{1}, \Theta_{2} \right) \left(I - \frac{\Theta_{2}^{2}}{L^{2}} \right)$ $\mathcal{T}^{3} = \mathcal{T}^{3}(\Theta_{1},\Theta_{2})\left(1-\frac{\Theta_{3}^{2}}{C^{2}}\right)$

Higher order thick shell theories can be obtained by expanding the displacements U_1 , U_2 , U_3 to higher orders in Θ_3 . This approach has been taken by Martinez-Marquez (18) and Abe (19).

The discussion so far has only considered shell theories obtained by substituting assumed displacement functions in the three dimensional strain-displacement and stress-strain relations, with slight modifications to take account of stress assumptions made in the theories. The equations of motion or equilibrium can similarly be obtained by substituting the expressions for stresses, strains and displacements in the appropriate three dimensional equations, or by considering the forces acting on an element of the shell.

Another approach to obtain not only the equations of equilibrium but the stress-strain and the strain-displacement relations for the shell is by use of a variational theorem. This is done by Naghdi (6) who obtains these equations for a thick shell theory, and then shows the approximations introduced into various thin shell theories.

More recently, since the work of Johnson and Reissner (20) and Reiss (21) a great deal of interest has been shown in the use of asymptotic expansions for the solution of shell problems. In this case the displacements, strains and stresses are all expanded as series in a small parameter based on h/L whore h is the thickness, and L some characteristic length of the shell. These expressions are substituted into the three dimensional equations, and a system of equations for varying orders of the parameter h/L extracted. This system can then be solved to give a solution to any order required. Here again as in the traditional approach to shall theory a variety of theories are obtained dependent on the asymptotic expansion used. The asymptotic technique was first applied by Johnson and Reissner (20) and Reiss (21) to static cylindrical shell problems.

5.

Since then this has been extended by Green (22, 23) and Green and Naghdi (24) to general shells, and dynamic problems. Green and Naghdi (24) by this technique obtain equations of motion for thin shells.

The variational techniques used (as discussed in section 1.3 and Chapter 3), for the solution of the free vibration of shells, require that the kinetic and strain energy are expressed in terms of the displacement functions for the shell. For thin shell theories the displacements of the shell can be completely defined in terms of the three middle surface displacements, no matter which theory is applied. For the thick shell theories considered, five or more functions of the two middle surface coordinates are needed to define the displacements for the shell.

1.3. Vibrations of Shalls

Only for shells with very special geometric properties and boundary conditions can the frequencies of natural vibration be found by exact methods. In order to obtain the solutions, in many cases, extra assumptions are made in the shell theory used. The most common of these are

a. The neglect of rotatory inertia.

- b. The neglect of in-plane inertia effects. This is used most in the theory of shallow shells.
- c. Bending stiffness of the shell is zero (Extensional vibrations).
- d. Extensional stiffness of the shell is infinite (Inextensional vibrations).

The use of exact solution techniques and the effects of the above assumptions, as well as those in section 1.2, are discussed by Kalnins (8).

To obtain solutions for more general shall geometries and boundary conditions a large variety of approximate techniques have been developed. The most commonly used are those based on energy principles, such as the Rayleigh-Ritz method. In this method the displacements are expressed as a series of functions, each of which satisfies the displacement boundary conditions A minimal energy condition is then used to find the values of the relative amplitudes of each member of the series, and the frequencies of vibration. The main drawback of this technique is that the displacement functions must each satisfy the displacement boundary conditions. This can be surmounted by finding stationary values of the Hamiltonian, subject to the constraints of the displacement boundary conditions. This technique is outlined in Chapter 3 and the basis of the method established in Appendix 1. For further details of the use of variational methods see Mikhlin (25) and Washizu (26).

For the application of these energy techniques, expressions are required for the strain and kinetic energy of the shell in terms of the displacement functions. For thin shell theory the displacements can be defined in terms of three functions of the two middle surface coordinates, but for the thick shell theories in use, five or more functions are involved. The energy techniques require sufficient terms in the approximating functions for convergence of the frequencies and mode shapes of vibration to be obtained. With the techniques available for solving the resultant eigenvalue problems, this convergence can be obtained for thih shell theory. For thick shell theory the number of functions involved does not allow adequate terms in each function to ensure convergence, except in special cases, such as axisymmetric shells where the problem is reduced to one dimension.

This problem has been overcome to some extent by the use of finite element techniques, Zeinkiewicz (27), which are very powerful in obtaining solutions to arbitrarily shaped shells, but in order to obtain solutions to thick shell problems assumptions must be introduced, usually involving the neglect of in-plane inertia terms. So although the results obtained using those techniques are very impressive, care must be taken over the effects of the other approximations introduced, especially for shells of high curvature. 1.4. Outline of Thesis

In this thesis a thick shell theory is proposed making use of the asymptotic series approach, and using the three dimensional strain-displacement and stress-strain equations. Displacement functions are proposed which satisfy the stress-free surface conditions on the shell, but which give non-zero

transverse shear stresses in the interior. The use of the stress-free surface conditions make these displacement functions dependent only on the three middle surface displacements as opposed to the five or more functions normally encountered in thick shell theory. The resultant stress field satisfies the surface boundary conditions, and therefore no shear constant is required to modify the stress-strain relations. The stresses and strains are obtained by direct substitution of the displacements in the three dimensional equations, thus no extra assumptions have to be made as in equations 1.2.4. The only assumptions made in this theory are on the form of the displacement functions, and, in the resultant energy expression, the neglect of terms of $O(h^3)$. This theory is then applied to the problem of free vibrations of shells.

In Chapter 2 the thick shell theory is derived. The three-dimensional and shell equations in tensor form for classical infinitesimal elasticity are outlined in sections 2.2 and 2.3. The displacement functions are then derived in section 2.4. for a general variable thickness double curvature shell. This is for a theory where transverse shear stresses are included, but transverse normal stress neglected. The equations are also formed for the derivation of a theory including transverse normal stress, but these equations can only be solved in very special case's.

In sections 2.5 and 2.6 the equations are applied to the cylindrical shell. In this case transverse normal stress is included. In sections 2.7 and 2.8 the twisted plate is considered. This is a double curvature shell and so the theory in which transverse normal stress is neglected is applied.

In Chapter 3 the numerical methods employed in the solution of the free vibration problem are outlined. The method used was to minimize the Hamiltonian for the shell subject to the constraints of the displacement boundary conditions. The displacement functions are expanded as double power series. In order to obtain convergence of the frequencies and mode shapes of vibration the size of the system of equations had to be reduced. To achieve this two methods were applied, symmetry, and the use of the stress free edge conditions. The computer program developed to solve the problem is given, and the problems encountered in this discussed.

The results obtained for the two shells are presented in Chapter 4. The problem considered in each case was the shell clamped along one edge, the other three being stress free. The results obtained were compared with theoretical and experimental results obtained by other researchers, and with thin shell theory results obtained using Flügge's (3) equations.

.

CHAPTER 2 SHELL THEORY

2.1. Introduction

In this chapter displacement functions are proposed, which satisfy the stress free surface condition, but give non-zero interior transverse stresses. These displacement functions, which vary through the thickness of the shell, are made dependent on the middle surface displacements, by the use of the stress free surface condition. The functions are applied to two cases: the cylindrical shell and the twisted plate.

9.

Use is made of the fundamental equations for the shell in tensor form obtained from Green and Zerna (28). These and the appropriate three dimensional equations and surface geometric properties are outlined in sections 2.2 and 2.3. Motivated by the works of Johnson and Reissner (20) and Reiss (21) for static shell problems, asymptotic series are introduced for the displacement function in section 2.4. The stress free surface conditions are then used to make these displacement functions dependent on the middle surface displacements This then gives for a particular shell only three functions to expand in series for the Reyleich Ritz numerical solution (Chapter 3), as egainst the five normally encountered in thick shell theory.

These displacement functions are then applied in section 2.5 and 2.6 to the cylindrical shell, and in section 2.7 and 2.8 to the twisted plate. In each case giving expression for displacements, strains and stresses, to be used later in solving the free vibration problem.

For the twisted plate the theory makes the usual thick shell assumption that the transverse normal stress is zero. For the cylindrical shell the transverse normal stress is included. The effects of these assumptions are discussed in chapter 4. Also included in sections 2.6 and 2.8 are the relevant thin shall theory expressions for the cylinder and twisted plate respectively. These are used for comparison with the thick shell theory in the numerical work of chapter 4.

2.2. General Theory

In this section the relevant formulas for classical infinitesimal elasticity in general curvilinear coordinates Θ^i are summarised. The notation used is that adopted by Green and Zerna (28). The covariant base vector \mathcal{G}_i is defined by the equation

$$g_i = \frac{\partial R}{\partial \Theta^i}$$

where R is the position vector of a point in the body

$$\mathcal{R} = \mathcal{R}(\Theta', \Theta^2, \Theta^3) \qquad 2.2.2.$$

The covariant and contravariant metric tensors, and the contravariant base vector are then given by the equations.

The displacement vector u of a point in space is written as

$$\underline{\mathcal{U}} = \mathcal{U}^{i} \underline{g}_{i} = \mathcal{U}_{j} \underline{g}^{j} \qquad 2.2.4.$$

from which the covariant strain tensor X_{ij} is obtained as

$$\begin{aligned} &\delta_{ij} = \frac{1}{2} \left\{ \underbrace{g_{i}}_{\partial ej} + \underbrace{g_{j}}_{\partial ej} + \underbrace{g_{j}}_{\partial j} \underbrace{\frac{\partial u}{\partial e^{j}}}_{\partial e^{j}} \right\} \\ &2.2.5. \end{aligned}$$

For an isotropic material the stress strain relations are given by the equation

$$T^{ik} = \mathcal{M}\left\{g^{ir}g^{ks} + g^{is}g^{kr} + \frac{2\eta}{1-2\eta}g^{ik}g^{rs}\right\} \forall rs \quad 2.2.6.$$

where \mathcal{T}^{1k} is the contraveriant stress tensor, and μ and η are elastic constants for the material, the shear modulus and Poisson's ratio respectively.

The stress tensor \mathcal{T}^{ik} is related to a stress vector \underline{T}_{i} , acting on the surface Θ^1 = constant at any point, by the equation

$$T_i = T_{\frac{j}{2}} \frac{g_j}{g_j} \qquad 2.2.7.$$
where $g = dot |g_{i_1}|$

The physical components of displacement, strain and strass are given in terms of the tensor quantities by the following equations

2.2.1

 $V^i = u^j \sqrt{g^j}$ lij = Dik/ att

2.3. Fundamental Shell Equations

Foints of the shell are defined by the position vector

$\underline{R} = \underline{r}(\Theta, \Theta^2) + \Theta_3 \underline{a}_3(\Theta, \Theta^2)$ 2.3.1. ***

where \underline{r} is the position vector of points on the middle surface of the shell. This surface M is defined by a system of curvilinear coordinates $\Theta'_{,} \Theta^2 \cdot \underline{a_3}$ is a unit normal to the surface M at each point \underline{r} . The outer surfaces of the shell are given by

$$\Theta_3 = -t$$
 2.3.2.

where for a constant thickness shell t is a constant, but in general it is a function of $\Theta'_{i}\Theta^{2}$.

For the surface M, the covariant base vector is given by

$$\underline{a}_{oi} = \frac{\partial \underline{r}}{\partial \theta^{i}}$$
2.3.3. **

The covariant and contravariant metric tensors, and the contravariant base vector then being given by the equations

$$a_{\mathcal{A}\mathcal{B}} = \underline{a}_{\mathcal{A}} \cdot \underline{a}_{\mathcal{B}} \qquad a^{\mathcal{A}\mathcal{B}} a_{\mathcal{A}\mathcal{B}} = \delta_{\mathcal{B}}^{\mathcal{A}\mathcal{B}} \qquad 2.3.4.$$

$$\underline{a}^{\mathcal{A}} = a^{\mathcal{A}\mathcal{B}} \underline{a}_{\mathcal{A}}$$

Other surface quantities required later are the second fundamental forms, and the Christoffel symbols. The second fundamental forms are defined by

* Note Indices repeated more than twice are not summed.

**<u>Note</u> From here on the middle surface coordinate system Θ_{i}, Θ_{j} is denoted by Greek indices, the three dimensional system by latin indices.

*** Note
$$\theta^3 = \theta_3$$

2.2.8. *

 $b_{\alpha\beta} = -\underline{a}_{\alpha} \cdot \underline{a}_{\beta,\beta} = -\underline{a}_{\beta} \cdot \underline{a}_{\alpha,\beta}$ $b_{\beta}^{\alpha} = a^{\alpha \cdot \lambda} b_{\beta\lambda} = a_{\beta\lambda} b^{\alpha \cdot \lambda}$

where β denotes the partial derivative w.r.t. Θ^{β} The Christoffel symbols are given by

 $\Gamma_{ax}^{\dagger\alpha} = \underline{\alpha}^{\alpha} \cdot \frac{\partial \underline{a}_{\beta}}{\partial \Theta^{\alpha}} = \frac{1}{2} \alpha^{\alpha} \lambda \left[a_{\beta\lambda,\beta} + a_{\lambda\lambda,\beta} - a_{\beta\lambda,\lambda} \right] 2.3.6.$

Applying the equations of this section to those of section 2.2. gives the following equations for the shell base vectors and metric tensors.

 $\begin{array}{l} \begin{array}{l} \begin{array}{l} g_{ex} = \left(\int_{a}^{A} - \Theta_{3} \int_{a}^{A} \right) g_{a} \\ g_{3} = g_{3} \\ g_{a'\beta} = a_{a'\beta} - 2\Theta_{3} \int_{a'\beta} + \Theta_{3}^{2} \int_{\beta}^{\lambda} \int_{a'\lambda} \\ g_{a'\beta} = g^{a'3} = 0 \\ g_{a'3} = g^{a'3} = 0 \\ g_{33} = g^{33} = 1 \\ g^{a'\beta} = a^{a'\beta} + 2\Theta_{3} \int_{a'\beta}^{a'\beta} + 3\Theta_{3}^{2} \int_{a'\beta}^{a'\beta} \int_{b}^{\beta} + 0\left(\Theta_{3}^{3}\right) \end{array}$

where $\mathcal{I}^{,\beta}$ is an infinite power series in \mathcal{E}_3 , provided that t is sufficiently small.

12.

2.3.5.

2.3.7.

For the shell the displacement vector is defined in terms of the middle surface base vector as

 $\underline{U} = u^{\alpha} \underline{a}_{\alpha} + u^{3} \underline{a}_{3} = U_{\alpha} \underline{a}^{\alpha} + U_{3} \underline{a}^{3}$ where $u^{\alpha} = a^{\alpha\beta} u_{\beta}$

which gives expression for the strain tensor

 $2Y_{\alpha\beta} = U_{\alpha} |_{\beta} + U_{\beta} |_{\alpha} - 2b_{\beta} U_{3}$ $-\Theta_{3}\left[b_{\beta}^{\lambda}\left(\mathcal{U}_{\lambda}\left[-b_{\lambda\alpha}^{\alpha\beta}\mathcal{U}_{3}\right]+b_{\alpha}^{\lambda}\left(\mathcal{U}_{\lambda}\right]_{\beta}-b_{\lambda\beta}\mathcal{U}_{3}\right)\right]$ $2 \aleph_{12} = U_3 , + b^{\lambda} U_{\lambda} + U_{1,3} - \theta_3 b^{\lambda} U_{\lambda,3}$ $\chi_{33} = U_{3,3}$

where $\mathcal{U}_{\alpha}/\beta$ denotes the covariant derivative of \mathcal{U}_{α} w.r.t. \mathcal{O}^{β} . In terms of the partial derivatives

 $U_{\alpha}|_{\beta} = U_{\alpha}, \beta - \int_{\alpha\beta}^{-i\lambda} u_{\lambda}$

 $u^{\prime}|_{\beta} = u^{\prime}_{,\beta} + \Gamma^{\prime}_{\lambda\beta} u^{\lambda}$

2.3.10.

2.3.8.

Substituting equation 2.3.7. into 2.2.6. gives for the shell stress-

strain relations

$$\begin{aligned} \mathcal{T}^{\alpha\beta} &= \mathcal{M}\left\{ g^{\alpha} g^{\beta\rho} + g^{\alpha\prime} g^{\beta\lambda} + \frac{2\eta}{l-2\eta} g^{\alpha\beta} g^{\lambda\rho} \right\} \delta_{\lambda\rho} \\ &+ \frac{2\mathcal{M}\eta}{l-2\eta} g^{\alpha\beta} \delta_{33} \\ \mathcal{T}^{\alpha3} &= 2\mathcal{M} g^{\alpha\lambda} \delta_{\lambda3} \\ \mathcal{T}^{33} &= \frac{2\mathcal{M}}{l-2\eta} \left\{ (l-\eta) \delta_{33} + \eta g^{\rho\lambda} \delta_{\rho\lambda} \right\} \end{aligned}$$

A shell stress tensor $\mathcal{G}^{\mathcal{I}\mathcal{J}}$ is defined by

2.3.12. *

2.3.14.

 $T_i = \tau^{ij} g_i \sqrt{g} = (\sigma^{i\lambda} a_\lambda + \sigma^{i3} a_3) \sqrt{a}$

which implies that

Shell stress and couple resultants are defined in terms of $\mathcal{G}^{\mathcal{I}}$ by



2.4. Displacement Functions

Using the general displacement-strain end stress-strain equations 2.3.9. and 2.3.11, displacement functions are proposed which satisfy the stress free conditions on the surface of the shell, but which give non zero transverse stresses interior to the shell.

Applying the stress free conditions to the stress-strain relations 2.3.8. implies that

 $\begin{aligned} & \chi_{13} = \chi_{23} = 0 \\ & (1-\eta)\chi_{33} + \eta g^{-\lambda}\chi_{\beta\lambda} = 0 \end{aligned} \qquad \Theta_3 = \pm t \\ \end{aligned}$

2.4.1.

* Note a = det/aii/

2.3.11.

An asymptotic series for the displacement functions is to be used. Therefore a new normal coordinate $\int f$ is defined, where

$$\int = \frac{\Theta_3}{t}$$

In general t is a function of the middle surface variables Θ_1 and Θ_2 , this is now redefined by

$$t = h X(\Theta', \Theta^2)$$
2.4.3

where h is a constant equal to half the maximum thickness of the shell, and X defines the variation in thickness.

Therefore
$$\Theta_3 = h X (\Theta', \Theta^2) S$$

-1 ≤ ∫ ≤ 1

 $-1 \leq X(\Theta', \Theta^2) \leq 1$

whore

end

With the adoption of the new normal coordinate S. derivatives w.r.t. the old coordinate system are given by

$\frac{\partial V}{\partial \nabla} =$	<u>JV</u>	-(NG.	•	
20	00	JV	X			2.4.5
263	hX	28				

The only condition on X is that it is a smooth function with continuous first derivatives of the same order of magnitude as the function itself.

The displacement functions \mathcal{U}_{i} are now defined as asymptotic series in t.

$$\mathcal{U}_{i} = \underset{T=0}{\overset{N}{\leq}} t^{\mathcal{J}\mathcal{J}}\mathcal{U}_{i}\left(\Theta, \Theta^{2}, S\right)$$

This series, unlike a Taylor's series expansion, can be truncated after any term, and still satisfy the stress free boundary conditions with nonzero transverse stresses. Therefore the series is truncated after the first term to give

2.4.6.

 $U_{i} = U_{i}\left(\Theta, \Theta^{2}, \Gamma\right) + t'U_{i}\left(\Theta, \Theta^{2}, \Gamma\right)$ 2.4.7.

Substituting 2.4.7. into the displacement-strain relations 2.3.9. and the stress-strain relations 2.3.11 gives rise to in-plane stresses of O(1) and transverse stresses of O $(\frac{1}{t})$. All stresses can be made of the same order, O(1) by making \mathcal{U}_{i} a function of the middle surface coordinates Θ' and Θ^{2} only. With this assumption, and expanding \mathcal{U}_{i} as a polynomial in \int_{i}^{i} gives

$$u_i = u_i(\Theta, \Theta^2) + t \underset{J=0}{\overset{M}{\leq}} {}^{J}u_i S^J \qquad 2.4.8.$$

Substitution of these displacement functions 2.4.8. into the stress free boundary conditions 2.4.1. and the displacement-strain relations 2.3.8, neglecting terms of $O(t^2)$, gives twolve equations in $3\mathbb{X} + 6$ unknowns. In order to give a set of equations to solve for $\int_{\mathcal{T}} \mathcal{U}_{\mathcal{A}}$ in terms of the middle surface displacements $\mathcal{U}_{\mathcal{A}}$, M is taken as three, to leave twelve equations in fifteen unknowns. This gives displacements with terms up to and including $\int_{\mathcal{A}}^{3}$, this being the least number of terms required to give non-zero transverse stresses, which satisfy the stress free boundary conditions The truncation of the polynomial after the $\int_{\mathcal{A}}^{3}$ term is made as a Rayleigh-Ritz approximation to the true solution, and it does not imply that higher order terms are negligible.

The twelve equations imply that

2.4.3.i.e. no $\int^2 \text{ term.}$ Also all the \mathcal{U}_i terms are independent of all other terms but themselves, therefore they are incorporated into the $\mathcal{O}_i(I)^O \mathcal{U}_i$ term.

Taking account of these facts, the indicial notation is now dropped, and the displacement functions redefined as

> $U_{1} = A + hX \int (D + \int^{2} E)$ $U_{2} = B + hX \int (F + \int^{2} G)$ $U_{3} = C + hX \int (H + \int^{2} J)$

2.4.10.

The six remaining stress free boundary conditions are now

 $\frac{\partial C}{\partial a} + b_i A + b_i^2 B + D + 3E = 0$ $\frac{\partial C}{\partial a^2} + b'_A + b^2 B + F + 3G = 0$ $X(\underbrace{\partial H}_{\partial G}, + \underbrace{\partial J}_{\partial G},) - 2 \underbrace{\partial X}_{\partial G}, J - 2X(b, E + b^{2}, G) = 0$ $X\left(\frac{\partial H}{\partial \theta^2} + \frac{\partial T}{\partial \theta^2}\right) - 2\frac{\partial X}{\partial \theta^2} \mathcal{J} - 2X(b_2'E + b_2^2G) = 0$ $\eta P(A, \frac{\partial A}{\partial \Theta^{x}}, B, \frac{\partial B}{\partial \Theta^{x}}, C) + (1-\eta)(H+3J) = O$ $Q(A, \frac{\partial A}{\partial a}, B, \frac{\partial B}{\partial a}, C, D, \frac{\partial D}{\partial a}, E, \frac{\partial E}{\partial a}, F, \frac{\partial F}{\partial a},$ G, == H, J: X, =) = 0

where P and Q are linear functions of the displacements and their derivatives. In the second case each displacement term being multiplied by X or one of its derivatives. These functions P and Q are dependent only on the shell middle surface geometry.

In general these equations are impossible to solve in terms of A, B and C, the middle surface displacements, but in certain special cases solution is possible, i.e. the cylindrical shell, section 2.6.

Now the more usual thick shell theory assumption is made, that the transverse normal stress is zero throughout the shell. This implies that J = 0. With this result, the first five equations of 2.4.11. are considered, the equation for the O(h) term in T^{53} being neglected. The Rayleigh Ritz solution procedure will take care of this (see Chapter 3). Thus the equation 2.4.11. become

 $\frac{\partial C}{\partial \theta'} + b_1 A + b_1^2 B + D + 3E = 0$ $\frac{\partial C}{\partial \theta^2} + b_2 A + b_2^2 B + F + 3G = 0$ $\frac{\partial H}{\partial \theta'} - 2(b_1' E + b_1^2 G) = 0$ $\frac{\partial H}{\partial \theta^2} - 2(b_2' E + b_2^2 G) = 0$ $\frac{\partial H}{\partial \theta^2} - 2(b_2' E + b_2^2 G) = 0$ $\frac{\partial H}{\partial \theta^2} - 2(b_2' E + b_2^2 G) = 0$

It should be noted that these equations are now independent of variable thickness, as all the derivative terms of X disappear with the assumption that J = 0. From the fifth equation H is obtained as a function of A, B, C

2.4.11.

2.4.12.

and their derivatives. Substituting for H in the first four equations, then gives four linear equation for D, E, F and G in terms of the middle surface displacements.

The condition on these equations having a solution being that

$$K = b_1 b_2^2 - b_2^2 b_1^2 \neq 0$$
 2.4.13.

i.e. that the shell has double curvature.

For a single curvature shell it is possible to construct displacement functions to give a non zero transverse shear stress, dependent on the middle surface displacements, in the curved direction. In this case Θ' and Θ^2 must be principal coordinates. Assuming Θ^2 to be the curved principal coordinate, gives, for a single curvature shell, b_2^2 as the only non zero mixed 2nd fundamental form. To obtain a non-zero transverse stress in the curved direction it is assumed that $T^{'3} = O$ throughout the chell, this implies that E = 0. Also as in the $T^{'3} = O$ assumption the $O(t)T^{13}$ equation is neglected, leaving the four equations

A Constraint State State State State
 A Market State State State
 A Market State State
 A Market State State

2.4.14.

 $\frac{\partial C}{\partial \Theta'} + D = 0$

 $\frac{\partial C}{\partial G^2} + b^2 B + F + 3G = 0$

 $\frac{\partial H}{\partial \theta^2} - 2b_2^2 G = 0$

 $\eta P(A, \frac{\partial A}{\partial \Theta}, B, \frac{\partial B}{\partial \Theta}, C) + (1-\eta)H = 0$

Thus displacement functions can be obtained in terms of the middle surface displacement only, which give non-zero transverse shear stresses in any curved coordinate direction of the shell, and yst satisfy the stress free surface condition. Rotatory Inertia effects are taken account of by including $O(h^2)$ terms in the kinetic energy expression.

2.5. Cylindrical Sholl

Here the equations of section 2.3 are applied to a cylindrical shell in order to obtain the displacement-strain and stress-strain relations. The shell is defined in terms of a middle surface M, taking the form of a cylinder of radius R. Thus in terms of cylindrical coordinates \propto and β , see fig 1, the position vector of the middle surface is given by

19.

2.5.1.

$$\underline{r} = (R \cos \alpha, R \sin \alpha, R_{\beta})$$

where β and \ll correspond to the curvilinear coordinates of section 2.3 Θ' and Θ^2 respectively.

The length coordinates involved are scaled by the radius of the cylinder R. The normal to the surface is given by

$$a^3 = (\cos \alpha, \sin \alpha, 0) \qquad 2.5.2.$$

Substituting equations 2.5.1 and 2.5.2 its equations 2.3.3., 2.3.4, and 2.3.5 gives the following expressions for the middle surface base vector, metric tensors and 2nd fundamental forms.

 $a_{1} = (0, 0, 1)$ 2.5.3.

 $a_2 = (-sind, Casa, 0)$ $a_{\alpha\beta} = a^{\alpha\beta} = \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}$ $b_{\alpha\beta} = \begin{pmatrix} 0 & 0 \\ 0 & -1 \end{pmatrix}$

 $b_{1}' = b_{1}^{2} = b_{2}' = 0$

 $b_{2}^{2} = -1$



Since the $\mathcal{A}_{\mathcal{A}\mathcal{B}}$ terms are constant, all the Christoffel symbols are zero, therefore

21.

2.5.4.

0.03

an an tha an

 $U_{\alpha}|_{\beta} = U_{\alpha}, \beta$

Substituting equations 2.5.2 and 2.5.3. into equation 2.3.7. gives expressions for the shell base vectors and metric tensors.

The geometric quantities for the middle surface 2.5.4. are substituted into equations 2.3.9. to give the following displacement-strain relations.

$$\begin{split} & \begin{cases} \chi_{11} = \mathcal{U}_{1,2,1} \\ & \chi_{12} = \frac{1}{2} \left\{ \mathcal{U}_{1,2} + \mathcal{U}_{2,1} \left(1 + \Theta_{3} \right) \right\} \\ & \chi_{22} = \left(\mathcal{U}_{2,2} + \mathcal{U}_{3} \right) \left(1 + \Theta_{3} \right) \\ & \chi_{13} = \frac{1}{2} \left(\mathcal{U}_{3,2,1} + \mathcal{U}_{1,3} \right) \\ & \chi_{23} = \frac{1}{2} \left\{ \mathcal{U}_{3,2,2} - \mathcal{U}_{2} + \mathcal{U}_{2,3} \left(1 + \Theta_{3} \right) \right\} \\ & \chi_{33} = \mathcal{U}_{3,3} \end{split}$$

Substituting equations 2.5.5. into equations 2.3.11. gives stressstrein relations, ignoring terms of $O(\Theta_i^3)$, of the form.

22.

 $\mathcal{T}'' = \frac{2\mu}{I-2\eta} \left\{ (1-\eta) \mathcal{Y}_{11} + \eta \left[(1-2\theta_3 + 3\theta_3^2) \mathcal{Y}_{22} + \mathcal{Y}_{33} \right] \right\}$ 2.5.7. $T^{12} = 2\mu (1 - 2\Theta_2 + 3\Theta_2^2) \mathcal{E}_{12}$ $\mathcal{T}^{22} = \frac{2M}{1-2\eta} \left\{ (1-\eta) \left(1-4\theta_3 + 10\theta_3^2 \right) \delta_{22} + \eta \left(1-2\theta_3 + 3\theta_3^2 \right) \left(\delta_{11} + \delta_{33} \right) \right\}$ $T^{\prime 3} = 2 \mathcal{M} \delta_{13}$ $T^{23} = 2 \mathcal{M} (1 - 2 \Theta_3 + 3 \Theta_3^2) \delta_{23}$ $T^{33} = \frac{2\mu}{1-2\eta} \left\{ \eta \left[\gamma_{11} + (1-2\theta_3 + 3\theta_3) \gamma_{22} \right] + (1-\eta) \gamma_{33} \right\}$

2.6. Cylindrical Shell - Displacement Functions

For the cylindrical shell three shell theories are considered. The thick shell theory using displacement functions 2.4.10 including non-zero transverse normal and shear stresses, secondly the thick shell theory with zero transverse normal stress \mathcal{T}^{33} , and thirdly the thin shell, Flügge theory.

The first case uses displacement functions of the form

 $\mathcal{U}_{,} = A + h X \, \mathcal{S}(D + \beta^{2} E)$

 $U_2 = B + h X g (F + g^2 G)$ 2.6.1.

23.

2.6.2.

 $U_3 = C + h X P(H + P^2 J)$

Substituting these equations into the stress free surface condition 2.4.1. and with the use of the geometrical cylindrical shell relations of section 2.5. gives

 $D + 3E + \frac{\partial C}{\partial B} = 0$ $F+3G+\frac{\partial C}{\partial x}-B=0$ $(I-\eta)(H+3J)+\eta(\frac{\partial A}{\partial s}+\frac{\partial B}{\partial \alpha}+C)=0$ $X\left(\frac{\partial H}{\partial B} + \frac{\partial J}{\partial B}\right) - 2\frac{\partial X}{\partial B}J = 0$ $X\left(\frac{\partial H}{\partial x} + \frac{\partial J}{\partial x}\right) - 2\frac{\partial X}{\partial x}J + 2XG = 0$ $X\left(\frac{\partial D}{\partial R} - \frac{\partial E}{\partial R}\right) - 2\frac{\partial X}{\partial R}E + X\left(\frac{\partial F}{\partial x} + \frac{\partial G}{\partial x}\right)$ $-2\frac{\partial X}{\partial G} + X(H+J) - X(\frac{\partial B}{\partial G} + C) = 0$

For general X it is impossible to solve these equations to obtain the displacement functions in terms of the middle surface displacements A, B end C. However, in certain special cases solution is possible. Considered here is the case when X is a function of \propto only. The equation 2.6.2. then solve to give

 $U_{1} = A + \frac{1}{2}h\chi_{S}^{2} \left((3 - S^{2}) (\vec{c} + \frac{3}{2}\vec{c}_{2} - \beta(\vec{T} - \eta) (T + \frac{d}{dx^{2}}(\chi T)) \right)$ $+ \left(\frac{1}{2}\right) \xrightarrow{2} \left[\frac{dX}{dX}\left(A + \frac{2}{3} + \frac{1}{2}\right)\right] + \left(1 - \frac{1}{3}\right) \xrightarrow{2} \frac{2}{3}$ 2.6.3.

 $U_{2} = \frac{\partial \overline{B}}{\partial \beta} + h \int X \left\{ \frac{\partial \overline{B}}{\partial \overline{\beta}} - \frac{\partial^{2} \overline{C}}{\partial x \partial \beta} + \frac{1}{2} \left(3 - \beta^{2} \right) \left(\frac{\eta}{1 - \eta} \right) \left(\frac{d}{dx} (XT) \right) \right\}$ $+\frac{dx}{dx}\left(\frac{\partial A}{\partial s}+\frac{\partial^2 B}{\partial x \partial s}+\frac{\partial \overline{c}}{\partial s}\right)$

$$\begin{split} \mathcal{U}_{3} &= \frac{\partial \overline{C}}{\partial \beta} + \frac{i}{2}h \int \chi \left(\frac{\eta}{1-\eta}\right) \left\{ \left(1-\int^{u}\right) \left(\frac{\partial A}{\partial \beta} + \frac{\partial^{2} \overline{g}}{\partial \chi \partial \beta} + \frac{\partial \overline{C}}{\partial \beta} \right) \\ &+ \left(3-\int^{u}\right) T_{1}^{2} \end{split}$$

where T = T(x), an arbitrary function,

 $\frac{\partial \bar{B}}{\partial B} = B$

and

 $\frac{\partial \bar{c}}{\partial k} = c$

If the shell is of constant thickness then these equations reduce

25.

2.6.4.

6.5.

to the form

 $U_{i} = A + \frac{1}{2}h \int \left(\left(3 - S^{2} \right) \left(\overline{C} + \frac{3^{2} \overline{C}}{2} - \beta \left(\frac{1}{2n} \right) \left(T + \frac{d^{2} T}{d \alpha^{2}} \right) \right)$ $+\left(1-\int^{2}\right)\frac{\partial^{2}C}{\partial R^{2}}$ $U_2 = B + h \int \left\{ B - \frac{\partial^2 \vec{c}}{\partial x \partial B} + \frac{i}{2} (3 - \beta^2) \left(\frac{\eta}{1 - \eta} \right) \frac{dT}{dx} \right\}$ $U_{3} = \frac{\partial \overline{C}}{\partial R} + \frac{i}{2}h \int \left(\frac{1}{1-n}\right) \left\{ \left(1-\int^{2}\right) \left(\frac{\partial A}{\partial B} + \frac{\partial B}{\partial x} + \frac{\partial \overline{C}}{\partial \beta} \right) \right\}$ $+ (3 - p^{1})T$

These equations for constant thickness give displacement strain relations of the form

 $\mathcal{Y}_{II} = \frac{\partial A}{\partial \beta} + \frac{i}{2}h \int \left[(3 - \beta^{*}) \left(\frac{\partial \bar{C}}{\partial \beta} + \frac{\partial^{*} \bar{C}}{\partial \chi^{*} \partial \beta} - \left(\frac{\eta}{1 - \eta} \right) \left(T + \frac{d^{*} T}{d \chi^{*}} \right) \right]$ $+(1-\beta^{1})\frac{\partial^{2} \sigma}{\partial \sigma}$

 $\delta_{12} = \frac{1}{2} \left\{ \frac{\partial A}{\partial \alpha} + \frac{\partial B}{\partial \beta} + h \int \left[2 \frac{\partial B}{\partial \beta} - \frac{1}{2} \left(1 + f^2 \right) \frac{\partial^2 \overline{c}}{\partial \alpha \partial \beta} \right] \right\}$ $+\frac{1}{2}(3-\beta^{2})\left[\frac{2c}{2}+\frac{2c}{2}-\beta(\frac{7}{2})(\frac{4}{2}+\frac{4^{3}T}{2})\right]$ $+h^{2}\int_{a}^{a}\left[\frac{2B}{2C}-\frac{2C}{2C}\right]_{a}^{a}$

 $\chi_{22} = \frac{2}{2}\frac{B}{2} + \frac{2}{3}\frac{C}{B} + \frac{1}{3}\int \frac{2}{2}\frac{2}{3}\frac{B}{2} + \frac{2}{3}\frac{C}{B} - \frac{2}{3}\frac{C}{2}\frac{1}{3}\frac{C}{3}$ $+\frac{1}{2}\left(\frac{7}{1-1}\right)\left[\left(3-5^{2}\right)\left(T+\frac{7}{1-1}\right)+\left(1-5^{2}\right)\left(\frac{2}{3}+\frac{2}{3}+\frac{2}{3}+\frac{2}{3}\right)\right]$

 $\mathcal{Y}_{13} = \frac{1}{4} \left(1 - \beta^2 \right) \left\{ 3 \left(\overline{C} + \frac{\partial^2 \overline{C}}{\partial x^2} + \frac{\partial^2 \overline{C}}{\partial \beta^2} - \beta \left(\frac{\eta}{1 - \eta} \right) \left(T + \frac{d^2 T}{d x^{\prime 1}} \right) \right\}$

 $+h_{f}\left(\frac{p}{1-h}\right)\left(\frac{\partial A}{\partial h^{2}}+\frac{\partial^{2}B}{\partial k^{2}}+\frac{\partial^{2}C}{\partial h^{2}}\right)$

26. $\chi_{23} = \pm \left(\frac{2}{1-\eta}\right)\left(1-\int^{2}\right)\left\{3\frac{dT}{dx}+h\int\left[\frac{dT}{dx}+\frac{\partial^{2}A}{\partial x\partial \beta}+\frac{\partial^{2}B}{\partial x^{2}}+\frac{\partial^{2}\overline{C}}{\partial x\partial \beta}\right]\right\}$ $\chi_{33} = \frac{1}{2} \left(\frac{\eta}{1-\eta} \right) \left\{ \left(1-3 \int_{-\infty}^{2} \right) \left(\frac{\partial A}{\partial \beta} + \frac{\partial B}{\partial \alpha} + \frac{\partial \overline{C}}{\partial \beta} \right) + 3 \left(1-\int_{-\infty}^{2} \right) T \right\}$ 2.6.5.

The displacement strain relations are used with the stress strain relations 2.5.7 to completely define the state of the shell in terms of the middle surface displacements and the arbitary function of \propto (T and the terms to make up \overline{C}).

It should be noted from these equations that the transverse shear terms are quadratics to the first order, but that the $\hbar \int$ terms mean that these strains and hence the stresses are not symmetric about the middle surface. The in plane strains are all, to the first order, the same as in the thin shell theory presented later, but the inclusion of the J term in χ_{33} means that the in plane stresses, to the first order, are quadratics in \int .

Secondly consider the thick shell theory with the assumption that to the first order $T^{33} = 0$, thus the displacements are of the form 2.6.1, but with J = 0.

The cylindrical shell being a single curvature shell means that, with the displacement functions considered, a non-zero transverse shear stress satisfying the surface conditions can only occur in the curved direction. The coordinates $\alpha'_{\beta}\beta'$ considered for the cylindrical shell are principal coordinates, therefore assuming also that $T'^3=0$ equation 2.4.14 become

 $\frac{\partial C}{\partial \beta} + D = 0$ $\frac{\partial C}{\partial \alpha} - B + F + 3G = 0$ $\frac{\partial H}{\partial \alpha} - 2G = 0$ $\eta \left(\frac{\partial A}{\partial \beta} + \frac{\partial B}{\partial \alpha} + C \right) + (I - \eta)H = 0$

2.6.6.

with solution

 $u_{i} = A - h \times \int \frac{\partial C}{\partial R}$ $U_{2} = B + h \times S \left\{ B - \frac{\partial C}{\partial x} - \frac{1}{2} \left(\frac{p}{p} \right) (3 - S') \left(\frac{\partial^{2} A}{\partial x \partial \beta} + \frac{\partial^{2} B}{\partial x'^{2}} + \frac{\partial C}{\partial x'} \right) \right\}$ $U_3 = C - h \times \int \left(\frac{1}{1-n}\right) \left(\frac{\partial A}{\partial B} + \frac{\partial B}{\partial \alpha} + C\right)$

27.

Substituting in the displacement strain relations 2.5.6 gives

 $\chi_{II} = \frac{\partial A}{\partial B} - h \times \int \frac{\partial^2 C}{\partial A^2}$ $\chi_{12} = \frac{1}{2} \left\{ \frac{\partial A}{\partial \alpha} + \frac{\partial B}{\partial \beta} + \frac{\partial B}{\partial \beta} + \frac{\partial B}{\partial \beta} - 2 \frac{\partial C}{\partial \alpha \partial \beta} \right\}$ $-\frac{1}{2}\left(\frac{1}{1-\eta}\right)\left(3-\frac{1}{2}\right)\left(\frac{\partial^{3}A}{\partial x^{2}\beta^{2}}+\frac{\partial^{3}B}{\partial x^{2}\beta}+\frac{\partial^{3}C}{\partial x^{2}\beta}\right)$ $+ h_{1}^{2} \chi^{2} \int_{a}^{2} \left[\frac{\partial B}{\partial A} - \frac{\partial^{2} C}{\partial x^{2} \partial A} - \frac{1}{2} \left(\frac{H}{1-1} \right) \left(3 - \int_{a}^{2} \right) \left(\frac{\partial^{3} A}{\partial x^{2} \partial A} + \frac{\partial^{2} B}{\partial x^{2} \partial A} + \frac{\partial^{2} C}{\partial x^{2} \partial A} \right) \right] \\ - h_{\frac{\partial X}{\partial B}} \left(1 + h_{X} \int_{a}^{a} \int_{a}^{b} \left(\frac{\partial^{2} A}{\partial x^{2} \partial A} + \frac{\partial^{2} B}{\partial x^{2} \partial A} + \frac{\partial^{2} C}{\partial x^{2} \partial A} \right) \right]$ 2.6. $Y_{32} = (1+hx_{5}) \{ \frac{2B}{3x} + C + hx_{5} \{ \frac{2B}{3x} - \frac{2C}{3x} \}$ $-\left(\frac{\eta}{1-\eta}\right)\left(\frac{\partial}{\partial t}+\frac{\partial}{\partial t}B_{+}+C\right)-\frac{i}{2}\left(\frac{\eta}{1-\eta}\right)\left(3-\frac{\delta^{2}}{2}\right)\left(\frac{\partial^{2}}{\partial t}A_{+}+\frac{\partial^{2}}{\partial t}B_{+}+\frac{\partial^{2}}{\partial t}C_{+}\right)\right]$ $-\frac{1}{2} + \frac{1}{2} + \frac{1$ $\chi_{13} = -\frac{1}{2}h\chi_{S}(\frac{1}{1-\eta})(\frac{\partial A}{\partial \beta} + \frac{\partial B}{\partial \beta} + C)$ $\mathcal{Y}_{23} = -\frac{1}{4} \left(\frac{\eta}{1-\eta} \right) \left(1 - g^2 \right) \left\{ \left(3 - 2h \times g \right) \left(\frac{\partial^2 A}{\partial \times \partial \beta} + \frac{\partial^2 B}{\partial \times^2} + \frac{\partial C}{\partial \times} \right) \right\}$ $\mathcal{Y}_{33} = -\left(\frac{7}{1-\eta}\right)\left(\frac{\partial A}{\partial \beta} + \frac{\partial B}{\partial \kappa} + C\right)$

The stress-strain relations to be used being 2.5.7.

Also included are the equations for Flugge's cylindrical shell theory. For this displacement functions are assumed of the form

$$U_{1} = A - \Theta_{3} \frac{\partial C}{\partial \beta}$$
$$U_{2} = B + \Theta_{3} \left(B - \frac{\partial C}{\partial \alpha} \right)$$
$$U_{3} = C$$

The displacement-strain relations are

$$\begin{split} & \chi_{II} = \frac{\partial A}{\partial \beta} - \Theta_{3} \frac{\partial^{2} C}{\partial \beta^{2}} \\ & \chi_{I2} = \frac{1}{2} \left\{ \frac{\partial A}{\partial \alpha} + \frac{\partial B}{\partial \beta} + 2\Theta_{3} \left(\frac{\partial B}{\partial \beta} - \frac{\partial^{2} C}{\partial \alpha^{2}} \right) \right\} \\ & \chi_{I2} = \left\{ \frac{\partial B}{\partial \alpha} + C + \Theta_{3} \left(\frac{\partial B}{\partial \alpha^{2}} - \frac{\partial^{2} C}{\partial \alpha^{2}} \right) \right\} (I + \Theta_{3}) \\ & \chi_{I3} = \chi_{I3} = \chi_{I3} = 0. \end{split}$$

and the stress-strain relations

$$\begin{aligned} \mathcal{T}'' &= \frac{2 \,\mathcal{U}}{1 - \eta} \left\{ \mathcal{Y}_{11} + \eta \left(1 - 2 \theta_3 + 3 \theta_3^2 \right) \mathcal{Y}_{22} \right\} \\ \mathcal{T}'^2 &= 2 \,\mathcal{U} \left(1 - 2 \theta_3 + 3 \theta_3^2 \right) \mathcal{Y}_{12} \\ \mathcal{T}^{22} &= \frac{2 \,\mathcal{U}}{1 - \eta} \left\{ \eta \left(1 - 2 \theta_3 + 3 \theta_3^2 \right) \mathcal{Y}_{11} + \left(1 - 4 \theta_3 + 10 \theta_3^2 \right) \mathcal{Y}_{22} \right\} \end{aligned}$$
 2.6.11.
$$\mathcal{T}^{13} &= \mathcal{T}^{23} = \mathcal{T}^{33} = \mathcal{O} \end{aligned}$$

2.6.9.

2.6.10.

2.7. Twisted Plate

The equations of section 2.3 are now applied to the geometry of a twisted plate. The middle surface of the twisted plate is defined by two curvilinear coordinates α' and β' , see fig 2. The characteristic length, a, is defined for the shell so that if β' is the angle of twist at any point, then that point is a length $\alpha \beta'$ in the 2 direction. α' is defined as the angle such that any point on the surface is a length $\alpha \beta' \alpha' \alpha'$ from the exis, CC^1 , of the shell.

The position vector of the middle surface is given by

 $Y = (a \tan \alpha \cos \beta, a \tan \alpha \sin \beta, a \beta)$

The length, a, is now scaled to unity and β and ζ are taken as the curvilinear coordinates Θ' and Θ^2 respectively. The normal to the middle surface is given by

$$Q_3 = (\cos \alpha \sin \beta, -\cos \alpha \cos \beta, \sin \alpha) \qquad 2.7.2.$$

Substitution of equations 2.7.1 and 2.7.2 into 2.3.3., 2.3.5 and 2.3.6 gives for the middle surface base vectors, metric tensors and 2nd fundamental forms the following expressions.

$$\begin{array}{l} \underline{a}_{1} = \left(0, 0, 1\right) \\ \underline{a}_{2} = \left(sec^{2}\alpha \cos\beta, sec^{2}\alpha \sin\beta, 0\right) \\ \underline{a}_{2} = \left(sec^{2}\alpha \cos\beta, sec^{2}\alpha \sin\beta, 0\right) \\ \underline{a}_{3} = \left(sec^{2}\alpha & 0 \\ 0 & sec^{4}\alpha\right), \quad \underline{a}^{\alpha} \stackrel{\beta}{=} \left(cs^{2}\alpha & 0 \\ 0 & cs^{4}\alpha\right) \\ 2.7.3. \end{array}$$

29.

2.7.1.


a



a -06

Shell Viewed Along Axis CC

2.7.3.

$$b_{\alpha\beta} = \begin{pmatrix} 0 & -ie(\alpha) \\ -se(\alpha) & 0 \end{pmatrix}$$

$$b_{1}' = b_{2}^{2} = 0, \quad b_{1}'^{2} = -\cos^{3}\alpha, \quad b_{2}' = -\cos^{3}\alpha$$

In this case the metric tensors are functions of α and therefore the non-zero Christoffel symbols are given by

$$\Gamma_{112}^{4} = -Sec^{2} \tan \alpha, \ \Gamma_{121} = \Gamma_{211}^{4} = Sec^{2} \tan \alpha$$

$$\Gamma_{122}^{4} = 2\tan \alpha Sec^{4}, \ \Gamma_{1}^{422} = -\sin \alpha \cos \alpha$$

$$\Gamma_{12}^{42} = \Gamma_{21}^{4} = \tan \alpha, \ \Gamma_{22}^{42} = 2\tan \alpha$$

$$2.7.4.$$

The shell base vectors and metric tensors are obtained by substituting equations 2.7.3 and 2.7.4 into 2.3.7 to give

$$\begin{array}{l}
g_{1} = a_{1} + \theta_{3} (as^{3} \times a_{2}, g_{2} = a_{2} + \theta_{3} (as \times a_{1}, g_{3} = a_{3}) \\
g_{1j} = \begin{pmatrix} st(i+(1+i)s^{4} \times e_{3}^{2}) & 2e_{3} st(x & 0) \\
2e_{3} st(x & st(i+(1+i)s^{4} \times e_{3}^{2}) & 0) \\
0 & 0 & 1 \end{pmatrix} \\
2\cdot7\cdot5. \\
0 & 0 & 1 \end{pmatrix} \\
g_{1j}^{ij} = \begin{pmatrix} cus^{2} (1+3 (us^{4} \times \theta_{3}^{2}) & -2\theta_{3} (us^{5} \times \theta_{3}^{2}) & 0 \\
-2\theta_{3} (us^{5} \times \theta_{3}^{2}) & -2\theta_{3} (us^{5} \times \theta_{3}^{2}) & 0 \\
0 & 0 & 1 \end{pmatrix} \\
\end{array}$$

The geometric properties of the shell are applied to the displacementstrain equation 2.3.9, with the use of equation 2.3.8, to give the strains in terms of the contravariant displacement as

 $\mathcal{Y}_{11} = \operatorname{sec}^{2} \mathcal{U}_{1} + \operatorname{sec}^{2} \operatorname{tan} \mathcal{U}_{1}^{2} + \Theta_{3}(\operatorname{seca} \mathcal{U}_{1}^{2} - \operatorname{sin} \mathcal{U}_{1} + \operatorname{cos}^{2} \mathcal{U}_{1}^{3})$ δ₁₂ = ½ { Seco U'₂ + Sec & U'₂ + Sec & U³ $+\Theta_3(S(\alpha u'_1) + S(\alpha u'_2) + 3 S(\alpha tan \alpha u'_2))$

822 = seca U2 +2 seca tana u2 + 03 (seca U2 + seca tana U+ U3) 2.7.6. $\delta_{13} = \frac{1}{2} \int U_{31}^{3} - se(\alpha U^{2} + sec^{2} U_{33}' + \Theta_{3} se(\alpha U_{33}^{2}) \Big]$ $Y_{23} = \frac{1}{2} \left\{ U_{32}^{3} - Se(\alpha U' + Se(\frac{4}{2} U_{32}^{2} + \Theta_{3} Se(\alpha U_{32}) \right\}$

These equations are given in terms of the contravariant displacement functions as they give simpler equations in later applications, see section 2.8.

Substitution of equations 2.7.5 into 2.3.11 gives the following stress-strain relations, terms of $\mathcal{O}(\Theta_3^3)$ being ignored.

 $T'' = \frac{2}{1-2\eta} \left\{ (1-\eta) c_{4} s_{4}^{2} (1+b c_{4} s_{4}^{2} - \theta_{3}^{2}) \delta_{11} - 4(1-\eta) c_{4} s_{4}^{2} - \theta_{3} \delta_{12} + \left[\eta c_{4} s_{4}^{2} (1+b c_{4} s_{4}^{2} - \theta_{3}^{2}) + 4(1-2\eta) c_{5} s_{4}^{2} - \theta_{3}^{2} \right] \delta_{22} + \eta c_{5} s_{4}^{2} (1+3c_{5} s_{4}^{2} - \theta_{3}^{2}) \delta_{23} \right\}$

 $T^{2} = \frac{2M}{1-2\eta} \left\{ -4(1-\eta)\cos^{3} \alpha \Theta_{3} \delta_{11} + \left[(1-2\eta)\cos^{4} \alpha (1+6\cos^{4} \Theta_{3}^{2}) + 8\eta\cos^{4} \Theta_{3}^{2} \right] \delta_{12} - 4(1-\eta)\cos^{4} \alpha \Theta_{3} \delta_{32} - 4\eta\cos^{5} \alpha \Theta_{3} \delta_{33} \right\}$

 $T^{22} = \frac{2\pi i}{1-2\eta} \left\{ \left[\eta \cos^{3} \left(1+6\cos^{4} \Theta_{3}^{2} \right) + 4\left(1-2\eta \right) \cos^{2} \Theta_{3}^{2} \right] \delta_{11} - 4\left(1-\eta \right) \cos^{2} \Theta_{3} \delta_{12} + \left(1-\eta \right) \cos^{2} \left(1+6\cos^{4} \Theta_{3}^{2} \right) \delta_{22} + \eta \cos^{4} \left(1+3\cos^{4} \Theta_{3}^{2} \right) \delta_{33} \right\}$

 $T^{13} = 2\mu \left\{ \cos^{2} \left(1 + 3\cos^{4} \Theta_{3}^{2} \right) \delta_{13} - 2\cos^{5} \Theta_{3} \delta_{23} \right\}$

 $T^{23} = 2M \left\{ -2\cos^{2} \theta_{3} \delta_{13} + \cos^{2} (1+3\cos^{4} \theta_{3}) \delta_{23} \right\}$

 $T^{33} = \frac{2 1}{1-2 \eta} \left\{ \eta c_{15} \times (1+3 c_{15} \times \Theta_{3}^{2}) \chi_{11} - 4 \eta c_{15} \times \Theta_{3} \chi_{12} \right\}$ + $\eta \cos \frac{1}{2} (1+3 \cos \frac{1}{2} \Theta_3^2) \delta_{22} + (1-\eta) \delta_{33}$

2.8. Twisted Plate - Displacement Functions

For the twisted plate only the thick shell theory with zero T^{33} is considered. The six equations 2.4.11 for the theory with non-zero T^{33} being impossible to solve for the displacement functions in terms of the middle surface displacement, even for special cases is constant thickness.

Thus displacements functions are considered of the form

u' =	$A + h \times S(D + S^2 E)$
$u^2 =$	$B + h \times f(F + fG)$
$u^3 =$	CthxSH

2.8.1.

2.7.7.

Substituting these expressions into the stress free boundary equations 2.4.1., using the geometric properties of the shell from section 2.7, and the relations between contravariant and covariant displacements 2.3.8. gives the set of equations

$$\frac{\partial C}{\partial \beta} - se(\alpha B + sec^{2}(D+3E) = 0 \qquad 2.8.2.$$

$$\frac{\partial C}{\partial \alpha} - se(\alpha A + sec^{4}(F+3G) = 0$$

$$\frac{\partial H}{\partial \beta} + 2se(\alpha G = 0$$

$$\frac{\partial H}{\partial \alpha} + 2se(\alpha E = 0$$

$$\eta \left(\frac{\partial A}{\partial \alpha} + \frac{\partial B}{\partial \alpha} + tan\alpha B\right) + (1-\eta)H = 0$$

These equations solve to give displacements functions of the form

U'= A + hxg cosx B- cos ~ 30 $-\frac{1}{2}\left(\frac{\eta}{1-\eta}\right)\left(3-\frac{\beta}{2}\right)\left(\cos\left(\frac{\partial^{2}A}{\partial x^{2}\beta}+\cos\left(\frac{\partial^{2}B}{\partial x^{2}}+3\sin\left(\frac{\partial^{2}B}{\partial x}+3\sin\left(\frac{\partial^{2}B}{\partial x}+3\cos\left(\frac{\partial^{2}B}{\partial x}+3\cos\left(\frac{$ $u^2 = B + h \times \int \left(c_0 s_{\mathcal{A}}^3 - c_0 s_{\mathcal{A}}^4 - \frac{\partial c}{\partial x} \right)^2$ $-\frac{1}{2}\left(\frac{7}{7-\eta}\right)\left(3-\int^{2}\right)\left(\cos \alpha \frac{\partial^{2}A}{\partial\beta^{2}}+\cos \alpha \frac{\partial^{2}B}{\partial\beta^{2}}+3\sin \alpha \frac{\partial^{2}B}{\partial\beta^{2}}\right)^{2}$ $u^{3} = C-h \times \int \left(\frac{7}{7-\eta}\right)\left(\frac{\partial A}{\partial\beta}+\frac{\partial B}{\partial\alpha}+3\tan \alpha B\right)$

Substitution into the displacement-strain relations 2.7.6. gives the equations

 $Y_{\mu} = sei \stackrel{2}{\sim} \stackrel{2}{\rightarrow} \stackrel{4}{\rightarrow} sei \stackrel{2}{\sim} tan \propto B + h \times S \left\{ 2 sei \propto \frac{2}{3} + cos \propto C - \frac{2}{3} \frac{C}{3} \right\}$ $-\sin\alpha\cos\alpha \frac{\partial C}{\partial \alpha} - \frac{1}{2} \left(\frac{1}{1-\eta}\right) \left(3 - \int^{2} \left(\frac{1}{3-\eta}\right) \left(3 - \int^{2} \left(\frac{1}{3-\eta}\right) \frac{1}{3-\eta} + 3\tan^{2} \frac{1}{3-\eta} + 3\tan^{2} \frac{1}{3-\eta} \right)$ +3 $\mathcal{L}(\alpha(2\mathcal{L}(\alpha-1)) \xrightarrow{\partial B} + 4 \mathcal{L}(\alpha \tan \alpha) \xrightarrow{\partial B} + \mathcal{L}(\alpha \xrightarrow{\partial^3 B})$ 2.8.4. + $h^2 \chi^2 \int -Cus \chi \left(\frac{1-2\eta}{1-\eta}\right) \frac{\partial A}{\partial B} - sin \chi cos \chi \left(\frac{1+2\eta}{1-\eta}\right) B - Cus \chi \left(\frac{\eta}{1-\eta}\right) \frac{\partial B}{\partial A}$ $+ \sin \alpha \cos \alpha \frac{\partial C}{\partial \beta} - \cos \alpha \frac{\partial^2 C}{\partial \alpha} - \frac{1}{2} \left(\frac{\eta}{1-\eta} \right) (3-\beta^2) \left(\frac{\partial^2 A}{\partial \beta^3} - \sin \alpha \cos \alpha \frac{\partial^2 A}{\partial \alpha} \right) \\ -3 \tan \alpha B + 3 \tan \alpha \frac{\partial^2 B}{\partial \beta^2} - 3 \sin^2 \alpha \frac{\partial B}{\partial \alpha} - \sin \alpha \cos \alpha \frac{\partial^2 B}{\partial \alpha^2} + \frac{\partial^3 B}{\partial \alpha^2} = 1 \\ \end{array}$ - h = x 5° (1)/suca = 1 + seca = + 3 seca + 3 seca + 3 seca B $-h^{2} \times \frac{\partial \times}{\partial \beta} \int^{4} \left(\frac{h}{1-\eta}\right) \left(\frac{\partial^{2} A}{\partial \beta^{2}} + \frac{\partial^{2} B}{\partial \omega \beta} + 3 \tan \omega \frac{\partial B}{\partial \beta}\right)$

 $\mathcal{Y}_{12} = \frac{1}{2} \left(\operatorname{sec}_{\mathcal{A}} \xrightarrow{\partial A}{\partial \mathcal{A}} + \operatorname{sec}_{\mathcal{A}} \xrightarrow{\partial B}{\partial \mathcal{A}} + 2\operatorname{sec}_{\mathcal{A}} C' \right) + h \times S \left(\operatorname{sec}_{\mathcal{A}} \left(\frac{1-2\eta}{1-\eta} \right) \xrightarrow{\partial A}{\partial \mathcal{A}} \right)$ + seca tand $\left(\frac{1-4\eta}{1-\eta}\right)B$ + seca $\left(\frac{1-2\eta}{1-\eta}\right)\frac{\partial B}{\partial \alpha}$ + tand $\frac{\partial C'}{\partial \beta} - \frac{\partial^2 C}{\partial \alpha \partial \beta}$ $-\frac{1}{4}(3-\beta^2)\left(\frac{\eta}{1-\eta}\right)\left[\sec^3\alpha\frac{\partial^3A}{\partial\beta^3}-\sec\alpha\tan\alpha\frac{\partial^2A}{\partial\alpha\partial\beta}+\sec\alpha\frac{\partial^3A}{\partial\alpha^2\partial\beta}\right]$ +3 secar tana B+ 3 secar tana 2B +3 secar (1+secar) 2B +2 secontana $\frac{\partial^2 B}{\partial \alpha^2}$ + Seca $\frac{\partial^3 B}{\partial \alpha^3}$ + Sec $^3 \frac{\partial^3 B}{\partial \alpha^2}$ $+\frac{1}{2}h^{2}x^{2}\int \cos^{2}x \frac{\partial A}{\partial \alpha} + \frac{\partial B}{\partial B} - \cos \alpha \frac{\partial^{2}c}{\partial \beta^{2}} + \sin \alpha \cos \alpha \frac{\partial c'}{\partial \alpha}$ $-\cos^{3}\alpha \frac{\partial^{2}C}{\partial \alpha^{2}} - (3-S^{2})(\frac{\eta}{1-\eta}) \left[\tan \alpha \frac{\partial^{2}A}{\partial B^{2}} + \frac{\partial^{3}A}{\partial B^{2}} + 3\left(2\sin^{2}\alpha - 1\right) \frac{\partial B}{\partial B^{2}} \right]$ $+4 \tan \alpha \frac{\partial^2 B}{\partial \alpha^2} + \frac{\partial^3 B}{\partial \alpha^2} \int -\frac{1}{2}h \int_{0}^{0} \left(\frac{\eta}{1-\eta}\right) \int_{0}^{0} \frac{\partial x}{\partial \alpha} \int_{-\infty}^{\infty} \int_{0}^{\infty} \frac{\partial^2 A}{\partial \alpha^2} + 3 \sec^2 \alpha \delta \int +\frac{\partial x}{\partial \beta} \int_{-\infty}^{\infty} \int_{0}^{\infty} \frac{\partial^2 A}{\partial \alpha^2} + 3 \sec^2 \alpha \delta \int +\frac{\partial x}{\partial \beta} \int_{0}^{\infty} \int_{0}^{\infty} \int_{0}^{\infty} \frac{\partial^2 A}{\partial \beta} + 3 \sec^2 \alpha \delta \int +\frac{\partial x}{\partial \beta} \int_{0}^{\infty} \int_{0}^$ + secar $\frac{\partial^{2}B}{\partial x \partial \beta}$ + 3 secar tan $\left(\frac{\partial B}{\partial \beta}\right)^{2} - \frac{1}{2}h^{2}X\int^{4}\left(\frac{n}{1.n}\right)\left(\frac{\partial X}{\partial x}\right)^{2}\frac{\partial^{2}A}{\partial \beta^{2}}$ $+\frac{\partial^2 B}{\partial x \partial \beta} + \frac{\partial A}{\partial x \partial \beta} + \frac{\partial A}{\partial \beta} + \frac{\partial^2 A}{\partial x \partial \beta} + \frac{\partial^2 B}{\partial x^2} + \frac{\partial^2 B}{\partial x^2} + \frac{\partial^2 B}{\partial x \partial \beta} + \frac{\partial^2 B}{\partial x^2} + \frac{\partial^2 B}{\partial x \partial \beta} + \frac{\partial^2 B}{\partial x \partial \beta}$ $\mathcal{Y}_{22} = 2 \operatorname{selar} tan \alpha \mathcal{B} + \operatorname{selar} \frac{\partial \mathcal{B}}{\partial \alpha} + h \times \int \left\{ 2 \operatorname{selar} \frac{\partial \mathcal{A}}{\partial \alpha} + \mathcal{C} + 2 \operatorname{tan} \alpha \frac{\partial \mathcal{C}}{\partial \alpha} - \frac{\partial^2 \mathcal{C}}{\partial \alpha^2} \right\}$ $=\frac{1}{2}(3-g^{2})(\frac{\mu}{1-\eta})\left[\operatorname{sic}^{3}(\tan n)^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(\sqrt{\frac{3^{2}}{3\beta^{2}}}+3\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(2\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}})^{2}\frac{1}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}(-1)\frac{2}{3\beta^{2}}+\operatorname{sic}^{3}+\operatorname{sic}^{3}+\operatorname{sic$ + 4 SEC + GIX 215 + SEC 2 23 - 4 X 5 { (1) 24 +3(1-1)tanx B- (1-21) 25 - sinx 20 + cow 200 + 2 (1) 3-52) (23-4 + 6sec x tanx 8 + 6sec 23 2.8.4. $+3 \tan \frac{\partial^2 \beta}{\partial x^2} + \frac{\partial^2 \beta}{\partial x^2} - \frac{\partial^2 \beta}{\partial x^2} + \frac{\partial^2$ +3 Sec x fan x 28] - h2 x 2x p4 (1) [2 + 2 28 + 3 tan ~ 3 + 3 se c 8 $\chi_{13} = -\frac{1}{2} \left(\frac{\eta}{1-\eta}\right) \left(1-\int_{-1}^{2}\right) \left(\frac{3}{2} \ln \alpha \frac{\partial^{2} A}{\partial \alpha \partial \beta} + \frac{2}{2} \ln^{2} \alpha \beta + \frac{2}{2} \ln \alpha + \frac{\partial^{2} B}{\partial \alpha} + \frac{3}{2} \ln \alpha \frac{\partial^{2} B}{\partial \alpha^{2}} + \frac{2}{2} \ln^{2} \alpha \frac{\partial^{2} B}{\partial \alpha^{2}} + \frac{2}{2} \ln^{$ + $h \times \int \left[\frac{\partial^2 A}{\partial A^2} + 3 \tan \alpha \frac{\partial B}{\partial B} + \frac{\partial^2 B}{\partial \alpha \partial \beta} \right]$

 $\chi_{23} = -\frac{1}{2} \left(\frac{1}{1 - \eta} \right) \left(1 - \zeta^2 \right) \left\{ \frac{3}{2} - \xi \varepsilon \right\}^2 \propto \frac{3^2 A}{\partial \beta^2} + \frac{9}{2} \xi \varepsilon \varepsilon^2 + \frac{1}{2} \xi \varepsilon \varepsilon^2 + \frac{3}{2} \xi \varepsilon^2 + \frac{3}{2$ $+h \times \left[\frac{\partial^2 A}{\partial \alpha \partial \beta} + 3 4 \cos^2 \beta + 3 \tan^2 \frac{\partial B}{\partial \alpha} + \frac{\partial^2 B}{\partial \alpha^2} \right]$ $\mathcal{X}_{33} = -\left(\frac{n}{1-n}\right)\left[\frac{\partial A}{\partial B} + 3 \tan \alpha B + \frac{\partial B}{\partial \alpha}\right]$

The stress strain relations for this theory are those in equation 2.7.7.

The equations for the thin shell theory for the twisted plate, equivalent to the Flügge Theory equation for the cylindrical shell, are also given here.

The displacement functions are $U' = A + \Theta_3 (US \propto B - GS^2 \prec \frac{\partial C}{\partial \beta})$ $U^2 = B + \Theta_3 (GS^3 \prec A - GS^4 \prec \frac{\partial C}{\partial \alpha})$ $U^3 = C'$

the displacement strain relations are

 $\mathcal{Y}_{II} = Sec_{\alpha}^{2} \frac{\partial A}{\partial B} + Sec_{\alpha}^{2} tana B + O_{3}\left(2Sec_{\alpha} \frac{\partial B}{\partial B} + COS_{\alpha}^{2}C'\right)$ $-\frac{\partial^2 C}{\partial a^2} - \sin \alpha \cos \alpha \frac{\partial C}{\partial \alpha} + \Theta_2^2 (\cos^2 \alpha \frac{\partial A}{\partial \beta} - \sin \alpha \cos \beta B$ + $\sin \alpha \cos \alpha \frac{\partial C}{\partial B} - \cos^3 \alpha \frac{\partial^2 C}{\partial \alpha \partial B}$) 2.8.6.

$$\begin{split} \mathcal{J}_{12} &= \frac{1}{2} \left\{ Sec_{\alpha}^{2} \frac{\partial A}{\partial \beta} + Sec_{\alpha}^{2} \frac{\partial B}{\partial \beta} + 2 Sec_{\alpha}^{2} C + \Theta_{s} \left(Sec_{\alpha} \frac{\partial A}{\partial \beta} + Sec_{\alpha}^{2} tan_{\alpha} B \right) \right. \\ &+ Sec_{\alpha} \frac{\partial B}{\partial \alpha} + tan_{\alpha} \frac{\partial C}{\partial \beta} - \frac{\partial^{2} C}{\partial \alpha \partial \beta} \right) + \frac{1}{2} \Theta_{3}^{2} \left(\cos_{\alpha}^{2} \frac{\partial A}{\partial \alpha} + \frac{\partial B}{\partial \beta} \right) \end{split}$$
- $\cos \left(\frac{\partial^2 C}{\partial R^2} + \sin \alpha \cos^2 \alpha \frac{\partial C}{\partial \alpha} - \cos^2 \alpha \frac{\partial^2 C}{\partial \alpha^2} \right)$

 $\mathcal{Y}_{22} = 2 \operatorname{sec}^{4} \operatorname{tand} \mathcal{B} + \operatorname{sec}^{4} \operatorname{\frac{\partial B}{\partial x}} + \mathcal{O}_{3}(2 \operatorname{sec} \operatorname{\frac{\partial A}{\partial x}} + C + 2 \operatorname{tand} \operatorname{\frac{\partial C}{\partial x}})$ $-\frac{\partial^2 C}{\partial \alpha^2} + \theta_3^2 \left(\frac{\partial B}{\partial \alpha} + \sin \alpha \frac{\partial C}{\partial B} - \cos \alpha \frac{\partial^2 C}{\partial \alpha \partial B} \right)$

 $\chi_{13} = \chi_{23} = \chi_{33} = 0$

2.8.5.

and finally the stress-strain relations are

 $T' = \frac{2M}{1-n} \left\{ \cos^4 \left(1 + 6\cos^4 \theta_3^2 \right) \delta_{11} - 4\cos^2 \delta_{12} \right\}$ + $[\eta \cos \alpha (1 + 6 \cos \alpha \theta_3^2) + 4 \cos \alpha \theta_3^2] \delta_{22}]$ $T^{12} = \frac{2M}{1-m} \left\{ -2\theta_3 \cos^2 \chi_{11} + \left[\cos^2 \chi (1 + 10\cos^2 \chi \theta_3^2) + 8\eta \cos^2 \chi \theta_3^2 \right] \chi_{12} - 2\theta_3 \cos^2 \chi \chi_{22} \right\}$ 2.8.7. $T^{22} = \frac{2M}{1-\eta} \left\{ \left[\eta \cos^{6} \chi \left(1 + \delta \cos^{4} \chi \Theta_{3}^{2} \right) + 4\cos^{6} \chi \Theta_{3}^{2} \right] \delta_{11} - 4\cos^{6} \chi \Theta_{3}^{2} \delta_{12} + \cos^{6} \chi \left(1 + \delta \cos^{4} \chi \Theta_{3}^{2} \right) \delta_{22} \right\}$ $T^{/3} = T^{23} = T^{33} = 0$

37.

CHAPTER 3 NUMERICAL ANALYSIS AND COMPUTING:

3.1. Introduction

The numerical techniques and computing methods used in this investigation for the solution of shell free vibration problems are presented in this chapter. The computer programs have been applied to the twisted plate and cylindrical shell, to both thin and thick theories for comparison, in every case to the problem of the shell clamped along one edge. The results obtained are presented in Chapter 4. The programs have been made as general as possible, so that they can be used with ease for any shell geometry and boundary condition.

The techniques used have been restricted due to the limitations of computing facilities available. The computer used was a KDF9, which is a 32K, 4S bit word machine with extensive disc and magnetic tape facilities. The machine was in heavy use, so the time available for running programs was rather restricted. Therefore the programs had to be made as efficient as possible. They were written in Egdon Algol (a specially modified version of Algol for use on the KDF9) and, where possible, standard library routines were used. The efficiency could have been improved by the use of a lower level language, but the use of Algol makes the program machine independent. The core store available for use was 26K. By careful chaining of the program, 22K was left for data space, this being the limit on the size of the system of equations that could be considered.

The numerical techniques applied to the problem are approximate solution techniques based on the Rayleigh-Ritz method. For this, strain and kinetic energy matrices are set up in terms of coefficients in the expansions of the middle surface displacements. Boundary conditions are then applied to reduce the size of these equations, so that all the displacement boundary conditions are satisfied. The resulting eigenvalue problem is then solved to give the frequencies and mode shapes for the shell. Other techniques used to reduce the size of the eigenvalue problem are symmetry, and the use of the stress free edge conditions.

The numerical techniques used are outlined in section 3.2, and the computer program in section 3.3. The particular sections of the program are then discussed in more detail in sections 3.4 to 3.6, and a general discussion of the program is presented in section 3.7.

3.2. Numerical Analysis

Exact solutions of free vibration problems for general shaped shells, with arbitrary boundary conditions, are not possible. For this problem a wide variety of approximate solutions can be used, most of these being relevant only to particular geometries or boundary conditions. The best general method, in the sense that it can be applied to any shell geometry and boundary conditions, is one using an energy principle. Essentially this is very similar to the Rayleigh-Ritz method, in that an approximation to the energy of the structure is minimised. In the Rayleigh-Ritz method the displacements of the shell are defined as a series of functions, each of which satisfy the displacement boundary conditions. The main drawback of the Rayleigh-Ritz method is that the displacement functions must be chosen to satisfy the displacement boundary conditions.

The solution technique used here is to find stationary values of the Hamiltonian:

 $H = \int_{t}^{t_{i}} (V - T) dt.$

3.2.1.

subject to the constraints imposed by the displacement boundary conditions. The functions used in the approximations to the displacements can now be general, and are chosen as double power series in the middle surface coordinates \propto and β . The basis of this variational method is given in Appendix 1.

For a general shell problem, the potential energy V and the kinetic energy T are defined as

 $V = \frac{1}{2} \int_{V} \delta_{ij} T^{ij} dV - \int_{V} F_{i} u^{i} - \int_{S} T_{i} u^{i} dS$

3.2.2.

 $T = \frac{1}{2} \int \rho \dot{u}_i \dot{u}^i dV.$

3.2.2.

40.

where \mathcal{U}_{i} and \mathcal{U}_{i} are the covariant and contravariant displacements, the dot representing the time derivatives. \mathcal{V}_{ij} is the covariant strain tensor, $\mathcal{T}^{\mathcal{U}}$ the contravariant stress tensor, \mathcal{F}_{i} the body forces, and \mathcal{T}_{i} the surface tractions. The first integral in the potential energy equation is known as the strain energy. For a free vibration problem the body force and surface traction integrals are both zero. For this reason only the displacement boundary conditions have to be satisfied; the stress free conditions are satisfied by the solution, which would otherwise give nonzero surface tractions \mathcal{T}_{i} .

The strain and stress tensors are known linear functions of the displacements \mathcal{U}_i and their derivatives.

 $\mathcal{T}^{ij} = \mathcal{D}_{\delta ij}$

3.2.3.

 $\delta_{ij} = \delta_{ij}(u_i)$

where D is the stress strain matrix.

The displacement functions \mathcal{U}_i are in turn defined in terms of the middle surface displacements A, B, C and their derivatives (see Chapter 2).

$$u_i = u_i(A, B, C)$$
$$u^i = a^{ij}u_j$$

3.2.4.

The middle surface displacements are defined as double power series in the scaled middle surface coordinates of the shell.

 $A = \sum_{I=0}^{M-1} \sum_{J=0}^{M-1} A_{IJ} \left(\frac{\alpha}{AL}\right)^{I} \left(\frac{\beta}{BL}\right)^{J} e^{-i\omega t}$ $B = \sum_{J=0}^{M-1} \sum_{J=0}^{M-1} B_{IJ} \left(\frac{\alpha}{AL}\right)^{I} \left(\frac{\beta}{BL}\right)^{J} e^{-i\omega t}$ $C' = \sum_{I=0}^{M-1} \sum_{T=0}^{N} C'_{IJ} \left(\frac{\alpha}{AL}\right)^{I} \left(\frac{\beta}{BL}\right)^{J} e^{-i\omega t}$ 3.2.5.

Substituting equations 3.2.5, 3.2.4., 3.2.3 and 3.2.2. into 3.2.1. gives a quadratic in the coefficients A_{IT} , B_{IJ} , C_{IJ} . To find the stationary values of H, it is differentiated by each of the coefficients to give a system of linear equations in the form of an eigenvalue problem

$$\left(K-\omega^2 M\right)\underline{x} = 0 \qquad 3.2.6.$$

where \underline{x} is a vector of the coefficients A_{II} , B_{II} , C_{II} . These equations are subject to the constraints implied by the displacement boundary conditions. The standard technique for this is to introduce Lagrange multipliers for each of the constraint equations. This, however, increases the number of equations to be solved, whereas the need is to decrease it as much as possible. The alternative method of achieving this is outlined below.

The constraint equations are set up in the form

$$G\underline{x} = 0$$

3.2.7.

41.

where, if there are R constraints, G is a R x 3MN matrix. These equations are used to reduce out R of the coefficients of x.

Partitioning 3.2.7. gives

 $G_1 q + G_2 p = 0$

3.2.8.

where p is a vector of the R coefficients to be reduced out and q the remaining 3MN - R coefficients.

Therefore,

$$p = -G_2 G_1 Q_2$$

From this a transformation matrix Q can be formed such that

$$\underline{x} = Q q$$
 3.2.10.

The strain and kinetic energy matrices are modified to become

Ñ	=	QKQ
M	=	$Q^T M Q$

leaving the eigenvalue problem

 $\left(\bar{K}-\omega^2\bar{M}\right)q=0$

This technique has been successfully spplied, by Webster (29), to the problem of a thin cylindrical panel clamped on four sides.

For the problems of shells clamped along one edge $\beta = O$, considered later, the displacement boundary conditions lead to equations of the form

$$\sum_{I=0}^{M-1} A_{IO} \left(\frac{\alpha}{AL}\right)^{I} = 0$$

$$3.2.13.$$

$$A_{IO} = 0$$

Thus, this boundary condition leads to certain coefficients being zero. For this case, it is not necessary to apply the boundary condition reduction above, as the conditions can simply be satisfied by leaving the appropriate terms out of the displacement functions. In this particular case, the displacement functions used then satisfy all the displacement boundary

42.

3.2.11.

3.2.12.

3.2.9.

conditions: and therefore the Rayleigh-Ritz technique is being applied. However, it is emphasized that this is true only for certain boundary conditions: in general, the Hamiltonian approach with reduction is required.

The eigenvalue problem is solved employing standard library routines, which can only be used to solve problems with up to 75 unknowns. For a general shell problem this restricts the (M, N) values used in the displacement functions to (5, 6) or (6,5), which are insufficient to ensure convergence of the frequencies.

Two techniques have been utilised to give larger (M, N) values, and thus better convergence: symmetry and stress-free boundary conditions.

Symmetry along one axis of a shell doubles the number of terms in the displacement functions, by solving two 75 x 75 problems. This increases the $(\underline{W}, \underline{N})$ values to (7, 8), (8, 7), which are high enough to give good convergence (Section 4.2). However, the use of symmetry restricts the shells and boundary conditions for which solutions can be found.

For the Hamiltonian solution it is only necessary that the displacement boundary conditions are satisfied, stress-free conditions are satisfied automatically. By utilizing these stress-free conditions the number of terms in the displacement functions can be increased, whilst the size of the eigenvalu problem remains the same. For example, consider the problem of a shell clamped along one edge. Putting the resultant forces and moments zero on the other three edges gives possible (M, N) values of (8, 9), (9, 8). Application of the force-conditions alone gives values (6, 8), (7, 7), (8, 6) similar to those which ensure adequate convergence in the symmetry problem. It should be noted that as the stress-free conditions are not necessary for application of the solution technique, only sufficient conditions for convergence need to be applied.

Standard library routines are used to solve the eigenvalue problem

 $(Z - \lambda I) Y = 0$

3.2.14.

to which equation 3.2.12 can be transformed.

Two solution techniques were used. The first should be sufficient to solve the free vibration problem. However, due to the build up of errors the second approach was also applied.

The first technique is to solve equation 3.2.16 by Householder's method, Wilkinson (30). In this Z is first transformed to tridiagonal form by a sequence of transformations.

 $Z_r = P_r Z_{r_1} P_r , \quad Z_0 = Z \qquad 3.2.15.$ $P_r = I - 2 \omega_r \omega_r', \quad \omega_r' \omega_r = I$

The elements of $W_{f'}$ are chosen so that $Z_{f'}$ will have zeros in all except the tri-diagonal positions of a particular column. Each transformation leaves unchanged the elements of the previous columns, so that for a system of equations of order J, J-2 transformations are required to reduce Z to its tri-diagonal form $Z_{f'}$. $P_{f'}$ has the property that it is orthogonal, therefore the eigenvalues of $Z_{f'}$ are the same as those of Z.

The eigenvalues of \mathbb{Z}_7 are then found by the method of Sturm sequences. Given a tri-diagonal eigenvalue problem

3.2.16. $\begin{array}{cccc} z_1 - \lambda & y_2 \\ y_2 & z_2 - \lambda & y_3 \end{array}$ $y_{n-1} = Z_{n-1} = \lambda - y_n$ $y_n = Z_n = \lambda$

let $f_n(\lambda)$ bd the determinant of the matrix formed from the first r rows and columns. It can then be proved that

 $f_{r+1}(\lambda) = (z_{r+1} - \lambda) f_r(\lambda) - y_{r+1}^2 f_{r-1}(\lambda)$ 3.2.17.

For a particular value of λ , say $\overline{\lambda}$, the sequence $f_r(\overline{\lambda})$ is a Sturm sequence, that is, the number of changes of sign in the sequence is equal to the number of roots of $f_r(\lambda)$ less than $\overline{\lambda}$ in algebraic value. This result can be used to compute all of the roots, or any particular one required, simply by bisecting the real axis and applying Newton's method to obtain the root to the accuracy required.

The eigenvectors are then evaluated by the method of inverse iteration. It can be shown that for the root nearest to S the eigenvector of the eigenvalue problem is given by the iteration.

This iteration is performed by successively solving the system of linear equations with the same matrix on the left, for varying right hand sides. This is performed by Gaussian elimination, the first step forms the upper triangular matrix, which is then used in all successive iterations.

When convergence is obtained the ratio of the components of $\frac{y'}{y'}(r_r) = \xi$ gives a correction to the eigenvalue S, so that the better approximation to the eigenvalue is given by

 $\lambda_r = S + E$

3.2.19.

This technique depends on Z being symmetric. This can be achieved by the transformation

3.2.20.

 $\frac{y}{\omega} = \frac{1}{2} \frac{y}{2}$ where $\overline{M} = LL^{T}$ and $\lambda = \omega^{2}$

This transformation can only be applied if \overline{M} is symmetric and positive definite. From the definition of \overline{M} this must be so, but as the number of terms in the displacement function are increased, error terms make the evaluation of L impossible. To evaluate the effects of these errors the second technique was used. For this the following transformation is applied.

Z = M'KY = 2 $\lambda = \omega^2$

 $Z = L' \overline{K} L^{-T}$

This Z is not symmetric, so Householder's method cannot be applied. Therefore the technique referred to as the H.Q.R. method is applied Wilkinson (30), Frances (31). Z is first reduced to Hessenberg form (an upper triangle matrix, with the first lower diagonal nonzero also). This is achieved by a sequence of transformations

$$Z_r = M_r Z_{r,1} M_r$$
, $Z_0 = Z$ 3.2.22.

The M_{fr} 's are chosen so that they reduce to zero all the terms in one column that need to be zero in the final Hessenberg form. In addition, the form of the M_{fr} 's is such that they do not affect previously zeroed columns. The order in which the columns are reduced is dependent on the largest term in the lower triangle, thus reducing the effects of error. By this method the matrix is reduced to Hessenberg form by J-2 transformations.

3.2.21.

The Q-R Algorithm is then applied to the resulting Z_{μ} so that

 $Z_{H1} = Q_1 R_1, \quad Z_{H2} = R_1 Q_1 = Q_2 R_2$ $Z_{H3} = R_2 Q_2 = Q_3 R_3$

such that

 $Z_{H2} = R_1 Z_{H1} R_1^{-1}$, $Z_{H3} = R_2 Z_{H2} R_2^{-1} = (R_2 R_1) Z_{H1} (R_2 R_1)^{-1} 3.2.24$.

where Qr is an orthogonal matrix, and R_ris an upper triangular matrix. This sequence converges to a matrix Z_Q . In the case of all real eigenvalues Z_Q is an upper triangular matrix, the diagonal elements of which are the eigenvalues of Z. For complex roots Z_Q has elements just below the diagonal. The solution of the 2 x 2 determinant of this term with the adjacent diagonal and upper triangular terms give a complex pair of roots of Z.

The eigenvectors are then formed by inverse iteration as in Householder's method.

The HCR method, in general gives complex eigenvalues and eigenvectors, and is. much slower than Householder's method. The use of the two techniques is discussed in section 3.7.

3.3. Computer Frogram. General Outline

The program has been made as general as possible, so that only certain blocks have to be changed dependent on the shell geometry, the boundary conditions applied, and the type of solution required. The flow diagram for the program is given in fig 3.

The program has been chained so that the maximum amount of core store is available for data storage. All the blocks in fig 3 are separate chains, except BCVAR which is performed manually, the program being split in two at this point. The first chain is very straightforward, involving only the

3.2.23.

setting up of control variables required later in the program. The other chains are discussed in more detail in the following sections. The computer program for the constant thickness thick twisted plate is given in Appendix 2. This program does not involve any boundary condition reduction. The appropriate parts of the program for boundary condition reduction applied to the thin cylindrical shell are given in Appendix 4. In Appendix 3 the data required to define the energy expressions for the constant thickness thick twisted plate is presented.

The approximate times for the various parts of the program are shown in table 1; first for the solution using symmetry and Householder's eigenvalue solution; second for a problem without symmetry, using the stress free boundary condition reduction and the HQR eigenvalue solution. The times given are for the maximum size problem by each method, so as to give the greatest accuracy possible. Fig 3. Frogram Flow Diagram



49•

Table 1 Computer Times

	Computer Core Time (sec)			
	Thin Theory Twisted Plate	Thick Theory Twisted Flate	Thin Theory Cylinder	Thick Theory Cylinder
(a) Symmetry & Householder				
MAIN	10	10	10	10
SESET	330	1030	100	250
BCSUB	10	10	10	10
SOLVE	350	350	350	350
TOTAL	700	1400	470	620
(b) Boundary Condition Reduction end HQR				
KAIN	10	10	10	10
SESET	650	2050	190	490
BCMAT	40	40 m m	ev i 10 40 av de €	40
BCRED	180	180	180	180
SCLVE	300	300	300	300
TOTAL	1180	2580	720	1020

3.4. Strain and Kinetic Energy Katrices

Applying equations 3.2.1 to 3.2.7 gives rise to a set of linear equations in the coefficients of the displacement function. The matrices, K and M, of these terms are referred to as the strain energy and kinetic energy matrices (or in the engineering literature as the "stiffness" and "mass" matrices respectively).

The energy matrices are set up separately, both being symmetric matrices of the form below -



where AB refers to coefficients arising from A_{IJ} B_{KL} terms, similarly for the other partitions. To keep the core requirements to a minimum the matrices are partitioned as above, each partition being worked out separately. Only six matrices have to be evaluated, because of symmetry.

The flow diagram for this part of the program, referred to as SESET is given in fig 4.

For this section of the program, data is required to define the energy expressions. For the thick shell theories these can be very complicated, in fact for the variable thickness twisted plate theory there are over 600 terms, each being of the form

 $\int \int CE(IC)X \sin^{35} \cos^{3c} \frac{\int^{10} + \int^{p} F}{\partial \alpha^{10} \partial \beta^{10}} \frac{\int^{K0+H0} G}{\int \chi^{K0} \partial \beta^{H0}} \bar{X}(X, \frac{\partial X}{\partial \alpha}, \frac{\partial X}{\partial \beta}) dx d\beta$

3.4.1.

where F and G can be either A, B or C; CE(IC) is a constant, dependent on shell geometry, and material properties; and \overline{X} is a quadratic function of X and its first derivatives (X defines the thickness variation of the shell. See section 2.4). In this expression the integration over the thickness has already been performed. Thus, for this shell, every term in the energy expressions can be defined in terms of a set of numbers

IS IA ID JD HD KD JS JC IC JX

and a set of expressions defining CE(IC)

IS defines whether the term is strain or kinetic energy,

 $IS = \begin{cases} 1 & Strain Energy \\ 2 & Kinetic Energy \end{cases}$

IA defines F and G in terms of A and B.

3.4.4.

3.4.5.



IA	F	G. G.
1	A A	A
2	B B	$\mathbf{B}^{\mathrm{res}} = \begin{bmatrix} \mathbf{B} \\ \mathbf{B} \end{bmatrix}$
3	C	С
4	Α	В
5	Α	C e
6	B B	C

and JX defines the quadratic \overline{X}_{\bullet}

JX	Ī
1	1
2	x ²
3	$\left(\frac{\partial X}{\partial x}\right)^2$
44 19 19 19 19 19 19 19 19 19 19 19 19 19	$\left(\frac{\partial X}{\partial \beta}\right)^2$
5	· X 3x
6 6	X 3x
7	2x 2x

For a constant thickness twisted plate JX is not required. For a cylinder, JS and JC are not required.

The number of terms in the energy expressions make it inefficient to store all these values at the same time. Thus, the terms are divided into sub-sets for IS and IA, only certain sub-sets being used at any one time. Also for IA = 1, 2 and 3, the numbers are divided into further sub-sets so those where ID = KD and JD = HD are separate from the remaining terms.

54.

3.4.6.

3.4.7.

The symmetry of these terms means that the evaluation and storage of their resultant energy terms can be performed twice as efficiently as for the general terms.

Arrays of the set of numbers 3.4.4, excepting IS and IA, are set up on disc for each value of IS, for three distinct cases.

(a) IA = 1, 2, 3 ID = JD and JD = KD

(b) IA = 1, 2, 3 Terms not included in (a)

(c)
$$IA = 4.5.6$$

Reference is made to these arrays in the program as A1, A2, A3. For each case, arrays I1, I2, I3 are formed, which define the position of the terms in A1, A2 and A3 respectively, for the different values of IA. This division of the arrays makes the program more efficient as different operations are performed for each of the cases (a), (b), (c) above, and in addition saves storage space as the arrays A1, A2, A3 and CE need only be declared so as to contain the largest sub-set of values. The coefficients CE are set up using four routines, which define CE for the separate cases.

(i)	COEFF 1	AA,	BB,	CC	Strain Energy terms
(ii)	COEFF 2	AA,	BB,	cc	Kinetic Energy terms
(iii)	COEFF 3	AB,	AC,	BC	Strain Energy terms ·
(iv)	COEFF 4	AB,	AC,	BC	Kinetic Energy terms

The remainder of this section will be devoted to discussion of the constant thickness twisted plate, with comments on the changes to be made for other shells. When referring in general to the arrays and routines defined above, AN will be used to refer to A1, A2 and A3, IN for I1, I2 and I3 and COEFFN for COEFF 1, 2, 3 and 4.

All the information is now readily available for the evaluation of the general energy expression 3.4.2. After taking out all constant terms and substituting the power series for the displacements the only terms left to be evaluated are of the form

 $\int_{a}^{b} \int_{a}^{AL} \left(\frac{\alpha}{AL}\right)^{R} \left(\frac{\beta}{BL}\right)^{S} \sin \frac{JS}{\alpha} \cos \frac{JC}{\alpha} d\alpha d\beta$

3.4.8.

For constant AL and BL, the α' and β' terms in this integral are independent, and so the integrals with respect to each can be evaluated separately. The integral with respect to β' can be carried out analytically, but the ∞' integral has to be evaluated numerically. This is done by Simpson's rule in the procedure INTEG. In the program, both integrations are carried out for all terms to be encountered in the energy expressions and the results stored in arrays. This means that each integration has only to be carried out once. Two arrays are used to store this data.

 $SC(I) = \int_{a}^{b} \left(\frac{\beta}{BL}\right)^{T} d\beta$

 $COEF(I, J, K) = \int_{a}^{AL} \sin \alpha \cos \alpha \left(\frac{\alpha}{AL}\right)^{K} d\alpha$

Tests were made on the accuracy of the numerical integration, and finally an accuracy to the sixth figure was used. This had no noticeable effect on the resulting frequencies.

For the constant thickness cylinder, the \checkmark integrals can also be carried out analytically. When variable thickness is included the arrays must be expanded to include all possibilities. In this case the \checkmark and β integrals may not be independent.

Also set up initially are arrays FD and FE which give the coefficients obtained as the result of differentiation with reference to \propto and β

 $FD(IJ) = Coefficient\left[\frac{d^{J}\left(\frac{d}{AL}\right)^{I}}{d}\right] = J(I-I) - \cdots (I-(J-I))/AL^{J}$

 $FE(IJ) = \left(\operatorname{orficient} \left[\frac{d^{\mathcal{J}}(\frac{B}{BL})^{T}}{d^{\mathcal{J}}} \right] = I(I-1) - \cdots - \left(I - (J-1) \right) / BL^{\mathcal{J}}$

56.

3.4.9.

3.4.10.

From this initial information the contribution from every energy term can be evaluated as a simple product for each term in the power series expansion of the middle surface displacements.

For example refer back to equation 3.4.2, and consider the constant thickness case, so that the X and \overline{X} terms are not included. Evaluation of the contribution to the energy expression of the term arising from the general cross term

 $\left[\left(\frac{\alpha}{AL}\right)^{I}\left(\frac{B}{BL}\right)^{J}\right]\left[\left(\frac{\alpha}{AL}\right)^{H}\left(\frac{B}{BL}\right)^{H}\right]$

can be written down for each term in the energy expression as

CE(IC)* COEF(JS, JC, I + K)* SC(J + H)*FD(I, ID)* FD(K, KD)* FE(J, JD)* FE(H, HD)

and its position in the appropriate partition of the energy array is defined by I, J, K, H, ID, JD, KD and HD.

The program SESET organises the setting up of the energy matrices. Having formed the initial matrices SC, COEF, FD and FE, it then sets up each of the partitions of the energy matrices. The IS loop controls the strain energy and kinetic energy calculations. For each in turn it calls the appropriate COEFFN routines, reads AN and IN from disc and evaluates each of the six partitions. The evaluations for AA, BB and CC are carried out in EVENAB, and those for AB, AC and BC in ODDAB. The flow diagram for EVENAB is given in fig 5; ODDAB is very similar. The routines given here are for shells symmetric about $\alpha' = 0$, the array AS and AT storing the symmetric and asymmetric parts of each partition. Without symmetry the routines become much simpler.

The routines EVENAB and ODDAB set up loops, one for the control information stored in AN for the particular partition under consideration, the others for four integers I, J, K, H, which define a cross term in the energy expression of the form 3.4.11. In these latter loops the coefficient of the

57.

3.4.11.

3.4.12.

term is evaluated and stored in the appropriate position in either the AS or AT matrix.

The resulting AS and AT matrices are stored in predetermined position on disc. In EVENAB, scale factors are also stored, equal to the diagonal terms in the strain energy. These are used later, in the boundary condition reduction (section 3.5).

Thus, at the end of SESET all the energy matrices have been set up and. stored on disc. The program for the constant thickness thick twisted plate theory is presented in Appendix 2, and the energy control information, required for this shell, in Appendix 3. The programs for the thin twisted plate and the thin cylindrical shell require a few changes in the dimensions of arrays, and in the case of the cylinder a different array COEF. The main differences are that the routines COEFFN must be set up and the control information IN and AN stored on disc for the particular theory.

For the thick cylinder theory, other changes have to be made to take account of the arbitrary function $T(\ll)$ introduced in section 2.6.







3.5. Boundary Conditions

This part of the program divides naturally into three parts :-

- (i) Forming boundary condition matrices.
- (ii) Choosing variables to reduce out.

(iii) Forming reduction matrix, and operating on energy matrices. These three parts are referred to as BCMAT, BCVAR and BCRED respectively. For problems where this boundary condition reduction does not have to be applied, a substitute routine BCSUB has been written to merge all the partitions of the energy matrices, in preparation for the solution phase. This routine is straightforward and is not discussed further here; the program is included in the overall program, given in Appendix 2. The reduction parts of the program are now discussed in more detail. The program is given in Appendix 4 for application to the thin cylinder problem. (i) BCMAT

The flow diagram for this part of the program is given in fig 6. This is for symmetry problems. Without symmetry, the program becomes much simpler. The whole of this section of the program is written for symmetry problems, so that comparisons can be made with results obtained without the boundary condition reductions. Most of this program is common for all shells. It has been written so that small blocks have to be added for particular boundary conditions to be applied. These blocks are included inside two loops for I and J which refer to terms $\left(\frac{\propto}{AL}\right)^{T} \left(\frac{\beta}{\beta L}\right)^{T}$ in each displacement function. Referring to the program in Appendix 4 the lines between each updating of the counter CC are for one boundary condition. These are displacement or stress-free boundary conditions. The stress-free conditions are applied to the major terms in the resultant forces and couples, on the edges, defined by equations 2.3.14. As an example consider the condition

 $N^{12} = 0$ on \propto = AL and β = BL for the thin cylindrical shell.

3.5.1.



These imply that

$$\frac{\partial A}{\partial x} + \frac{\partial B}{\partial \beta} = 0 \quad on \quad x \cdot AL, \quad \beta \cdot BL \qquad 3.5.2.$$

Substituting the displacement function 3.2.5. into these equations gives

 $\sum_{T=0}^{M-1} \frac{I}{AL} A_{I}(T_{+1}) + \frac{(J_{+1})}{BL} B_{IJ} = 0 \qquad J_{-1}, N_{-1}$ 3.5.3.

on $\alpha = AL$

and $\sum_{i=1}^{N-1} \frac{(I+I)}{AL} A_{IJ} + \frac{J}{BL} B_{(I+I)J} = 0$ I = I, M-13.5.4.

 $\operatorname{cn} \beta = \operatorname{BL}$

where in both cases the suffices of A and B cannot exceed M-1 or M-1 respectively. The parts of the computer program relating to these conditions are shown in Appendix 4 between lines 42 to 47 and lines 56 to 60 respectively. All other boundary conditions are set up in a similar manner. For the cylindrical shell the equations are straightforward, but for the twisted plate difficulties occur for conditions on lines of constant β , because of the trignometric terms involved. In this case, instead of separating out orthogonal terms, the equations should be set up at M equally spaced points along the edge.

It should be noted that in applying equations 3.5.3 and 3.5.4, one equation overdetermines the system, as the condition is applied twice at one point $\alpha' = AL$, $\beta' = BL$. This is referred to as a corner condition. Care must be taken to eliminate such equations as they give rise to a singular matrix G_2 .

The scaling applied to the equations at this stage is necessary because of the problem of errors later in the program. For this purpose, scale factors were evaluated in SESET, in terms of the diagonal terms in the strain energy matrix. Thus,

$$SS(I) = 1 / \sqrt{K(I, I)}$$
 3.5.5.

This scaling is applied to each term in the energy matrices so that

$$K(I, J) = K(I, J)^* SS(I)^* SS(J)$$
 3.5.6.

which makes each diagonal term unity, and all terms in the energy matrices of the same order of magnitude. The scaling is applied to the boundary condition matrix so that

$$G(I, J) = G(I, J) * SS(J)$$
 3.5.7.

The boundary condition matrix is then scaled again so that the major term in each row is unity. If this scaling is not applied, the order of terms in each equation can vary greatly, leading to an ill-conditioned reduction matrix.

(b) BCVAR

All the matrices referred to in this section are defined in section 3.2, equations 3.2.7 to 3.2.12. The problem is to choose reduction variables so that the resulting matrix G_{2} is non-singular, and the resulting energy matrices, are well conditioned. The boundary condition matrix, G, is, in general, sparse, and has a definite pattern to it. These properties mean that the choice of reduction variables is better made by manual rather than sutomatic methods. For this purpose, at the end of BCMAT, all the information required later in the program is stored on magnetic tape, and the boundary condition matrices printed out. The reduction variables are then chosen and input to the next section of the program BCRED.

The basic criteria applied in the choice of the reduction variables is to make the major terms in each row and column of C_2 as large as possible, whilst ensuring that the matrix is not obviously singular. The terms are made large so that the coefficients of the matrix $G_2^{-1}G_1$ are as small as possible. The matrix Q, which is used to reduce the energy matrices, is made up of a unit matrix and the matrix $G_2^{-1}G_1$


3.5.8.

Therefore, the reduced energy matrices \overline{M} , \overline{K} are made up of original matrices for the unreduced variables, plus terms from the rest of the energy matrix times $G_2^{-1}G_1$ terms. Thus, if the terms of $G_2^{-1}G_1$ are kept as small as possible, the effect on the conditioning of the resultant energy matrices is minimised.

For displacement boundary conditions this section of the program can be easily automated; but, for the stress free conditions the automatic methods tried, either gave rise to a singular G_2 , extreme ill-conditioning of the resultant energy matrices, or became time consuming.

(c) BCRED

The only problem in this section of the program is with core store on the computer, as the arrays involved can be very large. This means the use of a complex block structure. The flow diagram is given in fig 7, and the program in Appendix 4.

First the program reorders the boundary condition matrix G into G_1 and G_2 , this being dependent on the reduction variables chosen in BCVAR. From these, the reduction matrix Q is formed. Culy the partition $G_2^{-1}G_1$ of this is stored as the other partition is the unit matrix (see 3.5.8.)

The energy equations must now be pre- and post-multiplied by Q and its transpose. Here the problem arises with core store requirements, because due to the size of equations to be considered it becomes impossible to store the reduction matrix, and the whole energy matrix, at one time. Therefore, only three partitions of the energy matrix are in core at one time. These ere brought down from disc to form complete columns. For exemple in 3.4.1. partitions AA, BA and CA are formed in one large partition. The columns of this matrix are then reordered vertically to correspond with the Q matrix, and the premultiplication is carried out. The resultant matrix is stored on disc, and the operation repeated for all the other columns. These matrices are then brought down to form complete rows. The rows are reordered horizontally, and the post-multiplication performed. The resultant matrix is the energy matrix required by the eigenvalue solution procedure. The process is repeated for all the energy matrices.





3.6. Solution of the Eigenvalue Problem

The flow diagram for this section of the program is given in fig 8 and the program in Appendix 2. This section can be divided into four distinct sections.

(a) Operation on matrices \overline{M} , \overline{K} of equation 3.2.15 to bring the problem into the form of the standard eigenvalue equation 3.2.16.

(b) The eigenvalue solution routine.

(c) Operation on eigenvectors to convert them into coefficients of the original displacement functions.

(d) Evaluation of frequencies and mode shapes.

In the program the inverse eigenvalue problem is solved. This is because the smallest frequencies are the most important. By solving the inverse eigenvalue problem, these correspond to the largest eigenvalues, which are the ones that are most accurately determined. Unless all rigid body motions are reduced out, this technique cannot be used as \overline{X} would be singular, \overline{M} on the other hand is positive definite. For a singular matrix \overline{X} the operations 3.6.1 or 3.6.2 are not possible. Thus if a completely free vibration problem is to be considered, the matrices \overline{M} and \overline{X} must be interchanged in the program, and the true eigenvalue problem solved.

The first operation (a) is dependent on the eigenvalue solution routine to be used in (b) i.e. Householder or the Hessenberg Q-R (HQR). In the latter case, the library routine MATDIV is used which forms

 $Z = \overline{K}^{-1} \overline{M}$

3.6.1.





In Householder's method the lower triangular matrix L must be formed so that

 $\overline{K} = LL^{T}$ 3.6.2.

The matrix L is formed first, then the matrix Z is formed by operating on the mass matrix

$$z = L^{-1} \overline{M} L^{-T}$$
 3.6.3.

Special routines, INVLOW, MATML' and TRAML have been written to do this as opposed to using the standard library routines, because with L being a lower triangle matrix and Z and \overline{M} being symmetric this makes the program more efficient.

Part (b) the eigenvalue solution is carried out using standard library routines HOUSEH for the Householder solution and NSEIGB for the HQR method.

For the Householder solution, the eigenvectors obtained have to be operated on to reconvert them to the eigenvectors of the original problem. For this,

$$q = L^{T}y \qquad 3.6.4.$$

where y are the eigenvectors obtained in HOUSEH.

The above parts of the program have been written as two routines SOLUTION, one for each technique. The appropriate one is substituted as required.

If reduction has been used, then the eigenvectors must now be operated on again to obtain the original coefficients of the displacement functions. For this, the operation

$$x = Qq$$
 3.6.5.

is carried out. The resulting matrix x is reordered to return it to its original form. The eigenvectors are then rescaled to take account of the scaling introduced into the energy expressions. For the reduction operations, two routines TRANSF are provided; one to perform the operations outlined above, the other a dummy to be used when no reduction has been applied. The operations on the eigenvectors are only performed for those mode shapes required.

The frequencies are found from the eigenvalues as

 $f = \frac{1}{2\pi}\sqrt{\frac{1}{\lambda}}$ 3.6.6.

The mode shapes for the shell are output in the form of three matrices, one for each of the middle surface displacements A, B and C. Each element of the matrix refers to the displacements at a mesh point in the shell, and this matrix then gives a complete picture of the mode involved. It should be noted that the displacements obtained directly from the eigenvectors are tensor displacements, and these have to be converted to physical displacements by the use of equation 2.2.8.

3.7. Discussion

The program outlined in this chapter has been used to obtain the results presented in Chapter 4. All these results have been obtained for symmetric shells, without the use of the stress-free boundary conditions. It is shown (section 4.2) that good convergence is obtained for the lower frequencies of such shells. The accuracies of the shell theories are discussed in Chapter 4.

The one major difficulty in the numerical part of this project has been with the reduction technique used on the stress-free boundary conditions. The technique works for displacement boundary conditions, as has been shown by Webster (29). The program was tested on Webster's problem of a cylindrical panel clamped on all four sides, and gave identical results.

Testing of the stress-free boundary conditions was carried out on the thin cylindrical shell clamped at one end. Symmetry about the middle axis $\alpha = 0$ was used, so that direct comparison could be made with well converged results obtained without the stress free conditions.

With Householder's eigenvalue solution, break down of the program occurs when the build up of errors makes it impossible to form the lower triangular matrix L from the strain energy matrix \overline{K} . This is caused by the illconditioning of the matrix \overline{K} . This ill-conditioning can occur without the use of a boundary condition reduction, because of ill-conditioning of high

order polynomial expressions used in the displacement function. However, with the use of boundary condition reduction it occurs for lower order polynomials.

The form of the matrices M and K means that they are both symmetric and positive definite. For matrices with this property the eigenvalue problem 3.2.15. gives real positive eigenvalues and real eigenvectors. As extra terms are included in the displacement functions, the resulting low frequencies of vibration converge, but at a certain stage error terms give rise to negative eigenvalues. Additional terms then cause the break down in the formation of the lower triangular matrix L. However, until this break down occurs the major real eigenvalues are still well converged. To investigate the effects of the additional error terms on the major real eigenvalues, the HQR eigenvalue solution was used. This gave results similar to those of the Householder method, up to the break down point. At this stage the HQR method gave complex eigenvalues and eigenvectors, but still well converged real eigenvalues. The addition of further terms gave an increase in number and magnitude of the complex eigenvalues. Eventually these were as large as the major real eigenvalues. However, the real eigenvalues were still well converged. The problem now became one of time, as with this number of terms included, very few runs could be made. In fact, insufficient computer time was available to carry on with this work.

From the results obtained, however, it seems that the HQR method coupled with the stree-free boundary conditions is capable of solving general shell problems. The number of terms that had been used were sufficient to give good convergence for the problem without symmetry, provided the convergence criteria is the same as for the problem with symmetry. But, errors involved in this technique require further study.

The ill-conditioning of the energy matrices \overline{M} and \overline{K} is caused by the use of high order polynomials. It could have been overcome to some extent by using better conditioned series. The simple power series expansions were used here because of the ease of applying the boundary conditions.

The other reduction technique, mentioned earlier, but not applied, is that using the finite element technique, referred to as mass condensation or Guyan reduction, Zeinkiewitz (27). In this, degrees of freedom for which the inertia effects are negligible are reduced out. In the problems considered here this could be applied for a flap type mode to the inplane displacements A and B. However, as the purpose of the shell theories presented here is to investigate small order effects, it is dangerous to reduce out of the solution any minor terms. For this reason the technique was not used.

CHAPTER 4

RESULTS AND CONCLUSIONS

4.1. Introduction

In this chapter are presented the results of the application of the numerical techniques, outlined in Chapter 3, to the thick shell theories proposed in Chapter 2. The results are compared with those for the relevant thin shell theories, and with the experimental and theoretical results obtained by other researchers.

The thick shell theories considered are:

- 1) For the twisted plate the theory presented in section 2.8, which includes transverse shear stresses of the same order as the inplane stresses.
- 2) For the cylindrical shell the theory presented in section 2.6, which includes both transverse shear and normal stresses to the same order as the in-plane stresses.

All the shells considered are of constant thickness, and all have the following material constants

 $E = 3 \times 10^7 \text{ p.s.i.}$ $\eta = 0.3$ $\wp = 0.284 \text{ lbs/cu. in.}$

the values for steel. In each case, the shell is clamped along one edge, the other three being free. Also, the shells have an axis of symmetry, along the middle axis perpendicular to the clamped edge. This facilitates the use of the sumerical techniques cutlined in Chapter 3, without use being made of the stress-free boundary conditions. Thus, the numerical method is an application of the Rayleigh-Ritz technique.

In section 4.2, the convergence of the frequencies and mode shapes for varying number of terms in the power series expansions, of the middle surface displacements, is considered. This gives a measure of the accuracy of the numerical methods. The accuracy of the theories as compared with experimental and theoretical results obtained by other researchers is analysed in section 4.3. The effects of the thin and thick shell theories on the natural frequencies and mode shapes is then shown in section 4.4 for the twisted plate, and section 4.5 for the cylindrical shell. In both cases the effects of the two theories for various shapes and thicknesses are considered, and in the case of the twisted plate the effects of different angles of twist.

Conclusions on the use of the thick shell theories, and the finite element solutions to the problem are then drawn in section 4.6.

4.2. Convergence

The numerical techniques used here in the solution of the free vibration problem are approximate methods, the accuracy of which depend on how well the zeries expansions for the middle surface displacements approximate to the true solution. For convenience, in setting up energy expressions and in applying boundary conditions a double power series in each of the middle surface displacements has been used. The test of the numerical accuracy of the computing methods employed, is how well the frequencies and mode shapes converge, as the number of terms in the power series is increased.

The limitations on the number of terms that can be included in the series depend on two factors: the eigenvalue solution routines, and the conditioning of the system of equations obtained. In the results considered here, the limit has been set by the eigenvalue routines which are only effective for systems of equations of order 75.

In practice, this necessitates limiting the terms in the power series to those of order $\ll^{5}\beta^{7}$ or $\ll^{7}\beta^{5}$, where \ll and β are the middle surface coordinates of the shell. For the convergence tests, series including terms up to $\ll^{5}\beta^{7}$, $\ll^{5}\beta^{6}$, $\ll^{5}\beta^{7}$ and $\ll^{7}\beta^{5}$ were used.

Convergence tests were carried out for both the cylindrical shell and the twisted plate, for thick and thin shell theories, and for different shapes. In all cases, the results obtained were similar, so here two cases are presented.

- 1) The square cylindrical shell, referred to later as the Lindberg-Olsen fan blade, using Plügge's thin shell theory.
- 2) The 5:1 Length/Width ratio twisted plate with angle of twist of 30° using the thick shell theory.

For the cylindrical shall case, the blade considered is shown in fig 9. It is a square cylindrical panel 12" x 12" with radius of curvature 24" and thickness 0.12". The frequencies of the blade for varying number of terms in the power series are shown in table 2. In fig 10 the mode shapes are plotted, in each case for the lowest and highest frequency presented in table 2; the modes for the other values are intermediate to these two cases. The full line shows the lower frequency value, the better converged, . and the dotted line the higher value. Where no dotted line is shown the two lines coincide. The tables and figures show good convergence for the lower frequencies, to within 1%. The fact that the frequencies have converged to the correct values isshown later in section 4.3.

For the twisted plate, the blade considered has dimensions $24^{\circ} \ge 4.8^{\circ}$, en angle of twist of 30° and a thickness of 0.24° . The frequencies and mode shapes for this blade are given in table 3 and fig 11 respectively, in the same way as for the cylindrical shell. The results show the same degree of accuracy as for the cylinder.

Based on the results here, all results presented later in this chapter use the series with terms up to the order $\propto \beta^2$ except for shells with a length/width ratio of 1:1 where the $\propto \beta^2$ series has been used. This does not, for every mode, give the most closely converged result, but the accuracy is sufficient to show the effects of the various theories, and of varying thickness and angles of twist. If more accurate results are required, it is nocessary to run the computer program with the $\propto \beta^2$ and the $\alpha^2 \beta^5$ series, choosing the lowest value of frequency for each mode.

4.3. Comparison of Theoretical with Experimental Results

To investigate the validity of the theories used, comparisons are made with the practical and theoretical results obtained by other researchers.

The cylindrical shell results are compared with those of Lindberg end Olson (32,33) for a cylindrical fan blade. They present experimental and theoretical results. The shell is that considered in the convergence tests of section 4.2, and it is shown in fig 9. The blade was welded to a thick steel block, and the vibrational modes excited using a sinusoidal magnetic force.



This work was carried out twice by Lindberg and Olsen, and Was, in each case, compared with a different finite element theory. First, a non-conforming cylindrical shell element was used, and second, a doubly curved triangular shell element. A finite element analysis of the same blade has been carried out by Zeinkiewicz et al (17) using a doubly curved thick shell element.

Table 4 shows the comparison between the results obtained by Lindberg end Olsen, Zeinkiewicz, and those obtained using the power series technique applied to both thin and thick shell theories presented in Chapter 2. Also, the mode shapes obtained using the thin shell theory are compared with the theoretical and practical results of Lindberg and Olsen in fig 12. The thick shell theory mode shapes bore no relation to the experimental results.

The results show that the thin shell theory is comparable with any of the finite elements used, and that they have converged to within a reasonable accuracy of the frequencies of the experimental model, which are of course, subject to errors.

It is noticeable that both thick shell theories give frequencies significantly higher than the experimental and thin shell theory results. This effect will be discussed later in sections 4.4 and 4.5.

The twisted plate theories have been compared with results for twisted beems presented by Carnegie (34) and Dawson (35). A great deal of work has been published for twisted turbine and compressor blades, but no other results were found for constant thickness twisted plates clamped at one and.

The blade considered by Carnegie was 6" long, 1" wide and 0.0635" thick, the angles of twist were varied in steps of 15° from 0° to 90°. Cernegie compares his experimental results with theoretical values for the fundamental bending-bending frequency and the first five torsional frequencies, these being obtained by application of the Rayleigh method to twisted beams. For twisted beams, the two independent flexural modes of a straight beam become coupled. These are referred to, by Carnegie, as bending-bending modes.

Dawson's beam was 12" long, 1" wide and 0.25" thick. He gives results for angles of twist of 30°, 60° and 90°. He only presents results for bandingbending frequencies. His theoretical results were obtained using the Rayleigh-Ritz technique, expanding the displacements as power series in the length variabl

The comparison of Carnegie's results with both thick and thin shall theories is shown for frequencies in table 5, and for modo shapes in fig 13. These comparisons show that the thin shell theory gives the more accurate results for the bending-bending modes, whilst the thick shell theory is better for the torsional ones. In each case, the frequencies are predicted to within 10%, except in the thin shell theory for angles of twist of 90°. The mode shapes compare very well, the effect of twist on these being negligible. The break down of the thin shell theory for 90° angles of twist is shown later, in section 4.4, to occur for other shapes as well as the "beams" considered here. Discussion of this effect is left until then.

For the Dawson blade, the frequency comparisons are given in table 6, and the mode shapes in fig 14. Once again the thin shell theory gives more accurate results for the bending-bending modes, the frequencies being predicted to within 55 for all except the 60° fundamental mode, and all the 90° twist cases. For the torsional modes the thick theory gives the lower frequencies. But here, there are no results from Dawson to make comparisons. The comparisons in the tables are with Dawson's theoretical results. He only presents experimental results for mode shapes, these being indistinguishable from the mode shapes presented in fig 14.

4.4. Twisted Plate Results

Two further twisted plate shapes are considered here, these have length/width (L/N) ratios of 5:1 and 1:1. The effects of varying the emount of twist, and the thickness of the shells, for both thin and thick shell theories are given. The dimensions of the two shells are 24" x 4.8" and 12^{n} x 12" respectively.

The effects of angle of twist on the frequencies of the two shells is shown in tables 7 and 8. The results given in these tables for the flat plate case are obtained from Plunkett (36), who presents tables of frequency parameters for clamped cantilever plates with various L/W ratios. These tables of Flunkett are based on Rayleigh-Ritz results obtained by Earton (37) and Young and Felgar (38), and on experimental work by Plunkett himself and Crinstend (59). The flat plate values are only given for the flexural and torsional modes, as only these are considered by Flunkett.

The mode shapes for the 5:1 L/W ratio shell are shown in fig 15, these mode shapes are for both shell theories, there being no noticeable difference between them for this particular shell. Also, the angle of twist has a negligible effect on these modes, the 30° case being plotted. For the 1:1 L/W ratio shell the mode shapes are shown in figs 16 and 17 for the thin and thick shell theories respectively. In this case the effects of angle of twist are shown. These figures highlight the differences between the two shell theories, especially for the higher angles of twist.

The effect of thickness on the frequencies of the two shells, for the 30° angle of twist case, is shown in tables 9 and 10. For the 1:1 L/W ratio shell the mode shapes are given for the two theories in figs 18 and 19. For the 5:1 L/W ratio, shell thickness had very little effect on the modes shapes.

The thin shell theory applied to the two shells considered here, shows the same effect as with the Carnegie and Dawson blades; of very high frequencies for the 90° twist case. This effect, relative to the lower angles of twist, is not shown by the thick shell theory, even when it is applied to "thin shells" *. Considering the comparison with Carnegie's results, for the symmetric modes, it can be seen that the thick shell theory is predicting the lower frequencies are much higher than Carnegie's experimental results. The thick theory is giving the correct trend as the angle of twist is increased, but incorrect values.

Cohen (40) in his static study of the twisted plates, showed that O(t) terms in the in-plane stresses and strains (neglected by Love (1)) are important to the analysis of this shell. He calls these "strain-curvature" terms. These terms are included in the thin shell theory used here, but the theory has proved inadequate to deal with shells with high angles of twist. *<u>Note</u>: "Thin shells" are those for which the ratio of wavelength/thickness

is greater than 10.

It appears that for "thin shells", with high angles of twist, a theory is required which includes transverse shear stresses of O(t) compared to the in - plane stresses, rather than the O(1) ones included in the thick shell theory.

The major point of interest here is the difference between the thin and thick shell theories. The general "rule of thumb" approach to applying thin shell theory, is that results should be accurate for wavelength/thickness (λ/t) ratios greater than 10. This shell theory then breaks down, giving too high frequencies. This is because it overestimates the stiffness of the shell, by failing to take account of the transverse shear effects. Thick shell theories, on the other hand, include these transverse shear terms, and so for λ/t ratios less then 10 tend to give lower frequencies than this shell theory. However, when thick shell theory is applied to "this shells", is with λ/t ratios greater than 10, the inclusion of the transverse shear terms stiffens the shell and gives high frequencies. This result is shown by Zeinkiewicz's application of thick shell theory to the Lindberg-Olsen fan blade, table 4.

In the results of table 10 for the 1:1 L/W ratio shell the thick shell theory is predicting lower frequencies than the thin theory for λ/t ratio less than 10. For the 5:1 L/W ratio shell, and the Carnegie and Dawson blades, all symmetric modes have λ/t ratios greater than 10, and all the thin shell results, apart from those for 90° of twist, give lower frequencies. For the asymmetric modes, however, the thick shell theory is predicting lower frequencies for λ/t ratics of up to 80:1. This is because these blades can be considered as beams rather than shells, and for torsional vibrations of beams both shear terms in the plane of the cross-section are important. The thin shell theory assumes that one of these, the transverse shear term, is zero, and thus stiffens the blade too much.

Thus, the thick shell theory in predicting lower frequencies than the thin theory for all cases where λ /t is less than 10. These are the results that should be obtained from such a theory. The accuracy of the results obtained requires further confirmation from experiments carried out on shells equivalent to the 1:1 L/W ratio shell considered here.

4.5. Cylindrical Shall Results

For the cylindrical shell, as with the twisted plate, two shapes are considered with 5:1 and 1:1 L/W ratios. The thin and thick shell theories are applied for shells of various thicknesses. The dimensions of the blades are 24" x 4.8" and 12" x 12", both with a radius of curvature of 24". The second one is equivalent to the Lindberg-Olsen fan blade considered earlier.

Tables 11 and 12 show the effects of varying the thickness on the frequencies predicted by the two shell theories. The mode shapes are given in fig 20 for the 5:1 L/W ratio shell, end in fig 21 for the 1:1 L/W ratio shell. For the 5:1 L/W shell the thickness had no noticeable effect on the mode shapes for the thin shell theory, but they do vary slightly for the thick theory. Fig 20, therefore, shows one representative thin mode shape, and the two extremely distorted mode shapes for the thinner shells, so results are given in this case for only the thickest shell considered, 1.2" thick. Fig 21, gives the comparisons between the two theories for this blade. The effect of thickness on the mode shapes for the thin shell theory is shown in fig 22.

For the cylindrical shell, the thick shell theory applied contains transworse normal stress, as well as the transverse shear stresses. It can be seen in tables 11 and 12 that the thick shell theory is predicting higher frequencies than the thin theory, even for λ/t ratios much smaller than 10:1, well after the thin shell theory should have broken down. To confirm this, experimental results are required, but even so, the thick shell theory is obviously not going to predict frequencies more effectively than the far simpler thin theory.

The effect of thick shell theory predicting high frequencies for thin shells is shown in table 4 for the Lindberg-Olsen fan blade. The finite element thick theory, which does not include transverse normal stress, predicts high frequencies, although not to the extent that the thick shell theory presented here does. The reasons for this have been discussed in section 4.4. The inclusion of transverse normal stress has increased the stiffness still further, thus giving higher frequencies.

It would be of interest to see what results the thick theory presented in section 2.6, where transverse shear stress in the curved coordinate direction only is included, would give. However, this was discovered too late for the computer runs to be made.

4.6. Conclusions

The results presented for the twisted plate in section 4.4 show that the thick shell theory prodicts lower frequencies than the thin shell theory for λ/t ratios less than 10, this being the result that is expected from a thick shell theory. The accuracy of the results need to be evaluated by experimental means, however, before this theory can be applied with confidence. If experimental confirmation is obtained then the general thick shell theory presented in section 2.5, for shell with double curvature, including transverse shear stresses of the same order as the in-plane stresses, can be used as a half-way stage between thin-shell and full three-dimensional enalysis. This theory could then be applied to more complicated shell structures, by the construction of a finite element based on it.

Over the last three years curved thick shell elements have been developed besed on the usual thick shell theory discussed in Chapter 1, with five functions defining the displacement field, and constant transverse shear stresses across the element. This kind of element has been applied with reasonable success by Zeinkiewicz to Compressor end Turbine blades. Similar thin shell elements have also been developed. It would be interesting to see what regults these give when applied to the twisted plate with a 90° twist angle.

The break down of the thin shell theory for twisted plates with high angles of twist is a topic that requires further study. This appears to necessitate a thin shell theory which includes transverse shear stresses of O(t) compared to the in plane stresses.

For the thick shall theories, presented it can be seen that the energy expressions, and hence the solutions, become extremely complicated, when compared to the far simpler thin shell theories. Therefore, its use can only be justified when all other techniques have broken down. Finite element techniques based on traditional thin and thick shell theories would appear to be the best approaches to solving shell vibration problems. However the break down of thin shell theories for the twisted plate with high angle of twist shows the need for other approaches to be investigated.

The results for the cylindrical shell show that a thick shell theory, including transverse normal stresses in this way is not worthwhile. A threedimensional theory would be required before the theory could give accurate results.

•

TABLE 2. THIN CYLINDER CONVERGENCE TESTS

	Freq	uoncy c/s	1	
Mode	5 x 6	5 x 7	5 x 8	7 x 6
1	85.9	85.9	85.9	85.8
2	139	139	139	138
3	248	248	248	247
4	345	345	344	342
5	393	392	392	387
6	576	576	576	• 529
7	738	735	734	734
8	750	741	738	736
9.	790	781	780	780
10	848	845	844	807
11	•	•		1064
12	1242	1233	1226	1240
13	1291	1268	1238	1254
14	1409	1318	1275	1299
15	1402	1318	1297	1389
16	1764	1762	1761	1763

Dimensions 12" x 12" x 0.12". 24" Radius

TABLE 3. THICK TWISTED FLATE CONVERGENCE TESTS

				equency c/s		i Minadon di Anglorina di Sala anti Sala Sala di Anglo di	
Mode	of B 5	5 x 6	2 g 	5 x 7	5 x 8	7 x 6	
					**************************************	n A. (1997) - Maria Carlos (1997) - Maria 1	
a 1 .		15.7		15.7	15.6	15.7	
2		85		85	85	85	
: 3		154	A. S. S.	154	154	154	
4		283		248	247	283	
5	an a	395		386	379	395	
6		468		467	467	467	
7		579		569	510	578	
8		811		798	796	803	
. 9		1320		1192	1170	1316	
10	e to second	-		1781	1701	an a	
5 - K - 4		1			1		

Dimensions: 24" x 4.8" x 0.24", 30° Twist

• •

TABLE 4. LINDREEG - OLSEN CYLINDER COMPARISON

			Frequer	ncy c/s	1		
Mode	Experin Resul	rental lts	Lindberg Olse Element	en Finito ts	Thin Cylinder	Zeinkiewicz Thick Shell	Thick Cylinder
•	1971	1967	Triangular Curved(1971)	Triangular Cylindor Curved(1971) (1967)		Inecry	THEORY
1	85.6	86.6	86.6	93.5	85.8	113	183.5
2	134.5	135.5	139.2	147.6	138	147	413.2
3	259	258.9	251.3	255.1	247	296	543
4	351	350.6	348.6	393.1	342	442,0	758
5	395	395.2	393.4	423.5	387	475	1200
6	531	531.1	533.4	534.3	529		
7	751	751.2	746.4	781.5	736		
ð	743	743.2	752.1	792.2	734		
9	790.	792.1	790.1	863.2	780		
10	809	809.2	813.8	862.4	807		a a ser e
11	997	996.8	1009	1002	1064		
12	1216	1215	1232	1175	1233		
13	1252		1246		1299		an ta sa ta sa
14	1241	الحالي موجوعين مورد الم	1266		1254		an an ann an Anna an Anna Anna
15	1281		1286		1318		,
16	1310	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1303		1318	 Bernstein auf der Bernstein der B Bernstein der Bernstein d	t i para i i
17	1706		1652		1682		
18	1625		1653		1762	•	
19	1643	n an an an tha	1678		1811		an Anna Anna Anna Anna Anna Anna
20	1841		1838		1881		
21	1830		1885		2128		 The second second
22	1867		1926	n de la serie ∎ de la serie d	2006	• 	n an an an Arthre
23	1890		1941		2746		
24	2257		2224		2151		
. 25	2207		2274	•	2378	and the second sec	
4 4							

TABLE 5. CARIENCIE TWISTED BEAM COMPARISON

		Frequency c/s							
Bending Mode	Angle (Decrees)	Carneg	ie	Thin	Thick				
		Experimental	Theoretical	Plate	Tristed Plato				
1 1 1 1 1 1 1	30 60 90	57 59 61	62 63 65	64 77 174	71 132 231				
2	30	320		349	380				
2	60	265		284	334				
2	90	210		637	338				
3	30	1000		952	1116				
3	60	900		940	1056				
3	90	800		1100	1020				
4	30	1000		1124	1344				
4	60	1250		1324	1663				
4	90	1500		2900	2044				
5	30	1950		2396	2524				
5	60	1980		2640	2760				
5	90	2000		5516	3064				
Torsional Modə									
an a	30	730	725	896	770				
	60	800	790	948	864				
	90	900	910	1556	992				
2	30	2200	2150	2712	2340				
2	60	2500	2450	2860	2612				
2	90	2800	2650	4696	3000				
3	30	3800	3750	4596	4000				
3	60	4200	4100	4824	4403				
3	90	4700	4600	7920	5004				
4	30	5500	5500	6668	5924				
4	60	5900	5950	7012	6500				
4	90	6600	6600	11652	7308				

TABLE 6. DAWSON TWISTED BEAM COMPARISON

	Angle of	Freq	uency (C/S)	1996 - 1996 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 199 - 1997 - 199
Kodo Symaetric	.Twist (Dogrees)	Denson	Thin Twisted Plato	Thick Twisted Plate
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	30 60 90	28.5 28.8 29.3	29 •3 35 •5 79 •9	31.8 56.9 98.7
2	30	107	106	134
2	60	92.8	94.3	117
2	90	80.8	211	116
3	30	192	191	216
3	60	221	218	272
3	90	255	473	283
4	30	476	478	507
4	60	430	435	483
4	90	385	995	451
5	30	749	725	933
5	60	821	798	1021
5	90	849	1821	1079
Asympetric 1 1 1	30 60 90		775 775 1060	602 603 604
2	30		2104	2175
2	60		2104	2151
2	90		3041	2140
3	30		2329	1808
3	60		2330	1815
3	90		3752	1820
4	30	-	3895	3029
4	60		3895	3039
4	90		5553	3047
5	30		5487	4284
5	60		5494	4309
5	90		8007	4332

TABLE 7. EFFECTS OF TWIST ON TWISTED PLATE FREQUENCIES

5:1 I/W Ratio Shell

Dimensions 24" x 4.8" x 0.48"

			F	raquency	7 C/S				₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩	
	Plunkett Flat: Plate	Thin :	Thin Twisted Plate Theory				Thick Twisted Plate Theory			
Angle of Twist (Degrees) Mode	0 	15	30	60	90	15	30	60	90 90	
<u>Symmetric</u> 1 2 3 4 5 6	28.0 171 481 934	27.2 167 276 483 1014 1490	28.2 151 303 474 1018 1515	34.1 122 360 463 1040 1596	76.5 273 785 1031 2240 3188	29.2 178 355 511 1072 1756	31.0 165 382 508 1073 1758	57.9 143 432 539 1098 1779	101.3 143 443 631 1142 1817	
Asymmetric 1 2 3	260 795 1370	325 986 1687 2105	328 996 1700 2105	339 1028 1749 2101	545 1658 2845 3144	272 832 1441 2632	277 848 1463 2176	300 914 1557 2140	330 1010 1698 2125	
5 6	2011	2481 3637	2498 3662	2562 3760	1,265 6272	2169 3211	2195 3252	2315 3433	2492 3704	

TABLE 8. EFFECTS OF THIST ON THISTED PLATE FREQUENCIES

1:1 L/W Ratio Shell

Dimensions 12" x 12" x 0.48"

				Fre	equency	c/s	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		
	Plunkott Flat Plato	Thin T	wisted	Plate :	Theory	Thick	Twisted	. Plate	Theory
Angle of Twist (Degrees) Mode	0	15	30	60	90	15	30	60	90
Symmetric 1 2 3 4 5 6	113 695 891 1982	110 677 936 1859 1959 2098	113 619 952 1792 1997 2270	126 485 1009 1586 2133 2658	256 1005 1476 2958 3335 4272	121 715 959 2019 1863 2234	128 674 985 1921 1992 2393	223 571 1092 1756 2239 2776	347 552 1176 1573 2362 3031
Asymmetric 1 2 3 4 5 6	277 1011 2497 4458	397 1155 2175 2414 3354 4217	561 1297 2182 2447 3385 4166	857 1589 2220 2515 3550 3969	1257 2078 2794 3854 4832 5527	413 1119 2220 2487 3236 4539	640 1315 2240 2528 3301 4335	1008 1700 2296 2649 3558 4071	1200 1899 2345 2692 3718 3854

TABLE 9. EFFECT OF THICKNESS ON THISTED PLATE FREQUENCIES

5:1 L/W Ratio Shell

Dimensions 24" x 4.8" 30° Twist

	Frequency (C/S)							
and a second sec	Thin Ta	isted Plate	Theory	Thick Twisted Plate Theory				
Thickness/ Length Hodo	0.005	0.01	0.02	0.005	0.01	0.02		
Symmetric						- 		
1 2 3	6.69 40.3 119	14 .1 77.8 229	28.2 151 474	7.83 42.7 126	15.6 84.6 21.8	31 .1 165 508		
4 5 6	285 328 535	541 304 925	1018 303 1799	315 388 538	569 386 971	1073 382 1758		
Asympetric		الا موجد في الم	n an Araba Araba Marina an Araba					
1 2 3	98.2 296 498	171 520 884	328 996 1700	97.6 293 487	154 467 798	. 277 848 1463		
4 5 6	722 1066 2105	1295 1904 2105	2498 3662 2105	709 1072 2528	1192 1781 2176	2195 3252 2176		

TABLE 10. EFFECT OF THICKNESS ON TWISTED PLATE FREQUENCIES

1:1 L/J Retio Shell

Dimensions 12" x 12".30° Twist

	Frequency c/s								
	Thin Twisted Plato				Thic	k Twist	ed Plate		
Thickness/ Length Hode	0.01	0.02	0.04	0.1	0.01	0.02	0.04	0.1	
<u>Symmetric</u> 1 2 3 4 5 6	26.9 161 407 478 920 1085	53.6 321 564 944 1255 1932	113 619 952 1792 1997 3519	281 1424 2038 4468 4736 5509	32.9 175 476 530 1069 1153	64.8 346 623 1028 1335 2005	128 674 986 1921 1992 3381	311 1512 2038 4096 4746 5484	
<u>Asymmetric</u> 1 2 3 4 5 6	409 639 789 941 1291 1361	480 883 1189 1415 1890 2330	561 1297 2182 2447 3385 4397	836 2706 5115 5658 7883 10220	477 693 906 1043 1408 1463	571 943 1270 1499 1946 2431	640 1315 2240 2528 3301 4335	864 2473 4953 5491 6040 9350	

TABLE 11. EFFECT OF THICKNESS ON CLYINDRICAL SHELL FREQUENCIES

5:1 L/W Ratio Shell

Dimensions 24" x 4.8" Radius of Curvature 24"

an an an Anna a	Frequency c/s									
	Thin	Cylinder The	ory	Thick Cylinder Theory						
Inichness/ Longth Node	0.005	0.01	0.02	0.005	0.01	0.02				
<u>Symmetric</u> 1 2 3 4 5 6	9.82 60.5 168 333 792 2106	15.4 96.0 269 550 1108 2105	28.4 177 497 1031 1882 2106	16.9 107 371 1084 1942 2129	25.6 167 578 1362 3269 2129	44.2 292 921 2010 5387 2129				
<u>Asymmetric</u> 1 2 3 4 5 6	68.9 211 264 365 564 970	137 . 3 264 420 727 1102 1428	261 276 838 1425 2172 1448	83.2 264 280 579 1209 1495	161 280 505 1014 1495 1819	279 308 957 1493 1779 3082				

Table 12. Effect of Thickness on Cylindrical Shell Frequencies

1:1 L/W Ratio Shell

Dimensions 12" x 12", Radius of Curvature 24"

	Frequency C/3								
	Thin	Thin Cylinder Theory				Thick Cylinder Theory			
Thickness/ Length Mode	0.01	0.02	0.04	0.1	0.01	0.02	0.04	0.1	
Symmetric									
1	139	162	192	319	413	462	535	836	
2	248	441	824	1773	543	821	1497	2608	
3	392	632	948	2170	1735	1971	3011	4351	
4	735	1005	1808	4301	4123	42,04	6197	8050	
5	781	1279	2159	4877	7896	8415	8594	8748	
6	1233	1762	3229	7563	8164	8777	9918	11632	
16	4224	4226	4225	4226	4286	4284	4284	4284	
Asymmetric					P				
1	85.9	147	280	685	183	278	501	908	
2	345	557	1027	2469	758	1106	1423	3318	
3	576	1090	2063	5066	1199	2009	3107	6684	
4	741	1252	2377	5577	2521	3234	5434	9236	
5	845	1579	2989	7312	5350	6308	8507	11180	
6	1263	2302	4421	10350	11180	12970	14010	15950	
8	1765	1764	1762	1763	1825	1823	1824	1825	
15	4747	4743	4743	4742	4878	4878	4879	4880	



L/w = 1 H/L = 0.01 LOWEST FREQUENCY (BEST CONVERGED VALUE) FIG. IO. CYLINDRICAL SHELL CONVERGENCE TESTS, THIN SHELL THEORY.

4

, 1.6

99

updet ets etslague 100

die



L/W = 5 H/L = 0.01 30° TWIST

FIG.II.TWISTED PLATE CONVERGENCE TESTS, THICK SHELL THEORY.



•

н н. Т.

-
SYMMETRIC ASYMMETRIC 12 2 3 . ÷ 4 PERPENDICULAR FLAP (U2) 5

1

a series and

102

FIG.13. CARNEGIE BEAM MODES.

N. X.



FIG.14. DAWSON BEAM MODES

n de la companya de la comp



FIG. 15. MODES FOR 5:1 L/W RATIO TWISTED PLATE.



FIG.16. EFFECT OF TWIST ON MODES OF I:1 L/W RATIO TWISTED PLATE, THIN SHELL THEORY.

Sec. 1



H/L = 0.02

FIG.17. EFFECT OF TWIST ON MODES OF 1:1 L/W RATIO TWISTED PLATE, THICK SHELL THEORY

and the second secon



FIG.18. EFFECT OF THICKNESS ON MODES OF 1:1 L/W RATIO TWISTED PLATE, THIN SHELL THEORY.

> eta la este destructiones de la companya de la comp Este companya de la co



FIG 19. EFFECT OF THICKNESS ON MODES OF 1:1 L/W RATIO TWISTED PLATE , THICK SHELL THEORY.

ά.

109

الأبهاد المهالة المتحد والتناجين والم

1. 1. 1.36



ار ایران از ۲۰ واقعان کام میمان ایران در از ایران میتر ایروه میتر از میترد ا


FIG.21. COMPARISON BETWEEN THIN AND THICK CYLINDRICAL SHELL THEORIES, MODES FOR I:I L/W RATIO SHELL.

> e por la presidente de la constante de la const Constante de la constante de la

Sec. 2



FIG.22. EFFECT OF THICKNESS ON MODES FOR 1:1 L/W RATIO CYLINDRICAL SHELL THIN SHELL THEORY. ing na an A

al ang barang na kanalan na kanalan na kanalan na kanalan kanalan kanalan kanalan kanalan kanalan kanalan kana Kanalan kanalan

na an ann an taoine ann an 18 a Taoine a' fhairte an 18 ann an 18 ann an taoine ann an 18 ann

the share as a straight for the straight of the

Ser Stand Carry

BIBLICGRAPHY:

- 1. LOVE A E H "A Treatise on the Mathematical Theory of Elesticity" Dover Publications, New York, Fourth Edition 1944.
- 2. NASH W A "Bibliography on Shell and Shell-Like Structures" Vol 1, David W Taylor Model Basin, Report 863, Nov 1954 Vol 2, University of Florida Report, June 1957.
- FLUCCE W. "Stresses in Shells"
 Springer-Verlag Berlin/Heidelberg, 1960
- 4. NOVOZHILOV V. "The Theory of Thin Shells"P Noordhoff Ltd, 1959
- 5. GOLDENVEISER A.L. "Theory of Thin Elastic Shells" Pergamon Press, 1961
- 6. NAGHDI P M "Foundations of Elastic Shell Theory" Progress in Solid Mechanics Vol 4 North Holland Publishing Co., Amsterdam, 1963
- 7. KOITER W T "Foundations and Basic Equations of Shell Theory. A Survey of Recent Progress"

Proc of 2nd Symposium on The Theory of Thin Shells Copenhagen 1967 Springer-Verlag Berlin/Heidelberg, 1969.

- 8. KALNINS A "Dynamic Problems of Elastic Shalls" App Mech Rev Vol 18, p 867, 1965
- 9. GRCS C G and FCRSEERG K "Vibrations of Thin Shells: A Partially Annotated Bibliogrephy"

Lockheed Missiles and Space Co., Report SB-63-43, Apr 1963

- 10. HU W C L "A Survey of the Literature on the Vibrations of Thin Shells" Southwest Research Institute, Report No 1, June 1964.
- REISSNER E "Stress-Strein Relations in the Theory of Thin Elastic Shells" J Naths Phys, Vol 31, p 109, 1952
- 12. NAGHDI P M "On the Theory of Thin Elestic Shells" Quart. Appl. Nath., Vol 14, p 369, 1957
- MINDLIN R D "Influence of Rotatory Inertia and Shear on Flexural Motions of Isotropic Elastic Plates"

J. App. Mech, Vol 73, p 31, 1951

- HERRMANN G and MIRSKY I. "Three-Dimensional and Shell Theory Analysis of Axially Symmetric Motion of Cylinders"
 J. Appl. Mech. Vol 14, p 563, 1956
- 15. TSUI E Y W "Natural Vibrations of Cylindrical Panals" J Eng Mach, Proc ASCE, Vol 94, EM6, p 1425, 1968
- 16. ZEINKIEWICZ O C, ANMED S and IRONS B M "Analysis of Thick and Thin Shall Structures by Curved Finite Elements" Int. J. Nun. Meth. Engug., Vol 2, p 419, 1970
- 17. ZEINKIEWICZ O C, AHLED S and ANDERSON R G "Vibration of Thick Curved Shells, with particular reference to Turbine Blades" J Strain Analysis, Vol 5, p 200, 1970
- 18. MARTINEZ-MARQUEZ A M "General Theory for Thick Shell Analysis" J Eng Mech, Proc ASCE, Vol 92, EN6, p 185, 1966
- 19. AEE H "A Linear Theory of Thick Shells" Tech Reports Tohoku Univ., Vol 33, p 13, 1968.
- 20. JOHNSON M W and REISSNER E "On the Foundations of the Theory of Thin Elastic Shells"

J. Math. Phys., Vol 37, p 371, 1958

21. REISS E L "A Theory of the Small Rotationally Symmetric Deformations of Cylindrical Shells"

Commun. Pure. Appl. Math., Vol 13, p 531, 1960

- 22. GREEN A E "On the Linear Theory of Thin Elestic Shells" Proc Roy Soc, Vol A266, p 143, 1962
- 23. GREEN A E "Boundary Layer Equations in the Linear Theory of Thin Elastic Shells"

Proc. Roy. Soc., Vol A269, p 481, 1962

- 24. GREEN A E and NAGHDI P M "Some Remarks on the Linear Theory of Shells" Quart J. Mech. Appl. Math., Vol 18, p 257, 1965
- 25. MIKHLIN S G "Variational Methods in Mathematical Physics" Pergamen Fress. 1964
- 26. WASHIZU K "Variational Methods in Elasticity and Plasticity" Fergamon Press, 1968

- 27. ZEINKIEWICZ O C "The Finite Element Method" McGraw-Hill, New York/London, 2nd Ed, 1970
- 28. GREEN A E and ZERNA W "Theoretical Elasticity" Clarendon Press, Oxford, 1954
- 29. WEBSIER J J "Free Vibrations of Rectangular Curved Panels" Int. J. Mech. Sc., Vol 10, p 571, 1968
- 30. WILKINSON J H "The Algebraic Eigenvalue Problem" Clarendon Press, Oxford, 1965
- 31. FRANCES J G F "The Q-R Transformation" Comp. J., Vol 4, p 265, p 332, 1961/2
- 32. LINDBERG G M and OLSEN M D "A Finite Cylindrical Shell Element end the Vibrations of a Curved Fan Blede" NRCC No 10198, Report LR-497, 1968
- 33. LINDEERG G M and OLSEN M D "Dynamic Analysis of Shallow Shalls with a Doubly-Curved Triangular Finite Element" J. Sound. Vib., Vol 19, p 299, 1971
- 34. CARNEGIE W "Vibrations of Pro-Twisted Cantilover Blading" Proc. Inst. Mach. Engn., Vol 173, p 343, 1959
- 35. DAWSON B "Coupled Bending-Bending Vibrations of Pre-Twisted Cantilever Blading Treated by the Rayleigh-Ritz Energy Method" J. Mach. Eng. Sc., Vol 10, p 381, 1968
- 36. PLUNKETT R "Natural Frequencies of Uniform and Non-Uniform Rectangular Cantilever Plates"

J. Mech. Eng. Sci, Vol 5, p 146, 1963

- 37. BARTON M V "Vibrations of Rectangular and Skew Cantilever Plates"J. Appl. Mech., Vol 18, p 129, 1951
- 38. YOUNG D and FELGAR R P "Tables of Characteristic Functions Representing Normal Modes of Vibration of a Beam" ERB 4913, Univ of Texas, 1949
- 39. GRINSTEAD B "Nodal Fattern Analysis" Proc. Inst. Mech. Engn. Vol 166, p 309, 1952

40. COMEN J W "The Inadequacy of the Classical Stress-Strain Relations for the Right Helicoidal Shell"

Proc. of Symposium on The Theory of Thin Elastic Shells Delft 1959 North Holland Fublishing Co., Amsterdam, 1960.

ACYNCULEDGEMENTS

I would like to acknowledge the contributions of the following to this thesis.

Prof A J M Spencer for the use of the Theoretical Mechanics Departmental facilities.

Dr W A Green for his supervision and edvice throughout the work.

The Cripps Computing Centre staff for their help with the computing side of the project.

The Science Research Council for their financial assistance.

Miss Jean McAlonan and Mrs Sandra Snelling for typing the thesis, and Miss Carol Gould for her help with the illustrations.

My Father for his encouragement, and patience in writing out the equations.

And Ginny, for all her help.

APPENDIX 1. BASIS OF THE VARIATIONAL METHOD

Consider the vibrations of en elastic body subject to the conditions $u_4 = 0$ on S_u for all t A.1 $t_{ij}n_{j} = 0$ on S_{m} for all t

Assume there exist solutions

so that if t_{ij}^{α} is the stress arising from U_i^{α}

 $\frac{\partial t_{ij}}{\partial X_i} = -\beta \mathcal{J}_{\alpha}^2 \mathcal{U}_i^{\alpha}$ A.3.

Multiplying this equation by \mathcal{U}_{i}^{β} and the similar equation for \mathcal{I}_{ij}^{β} and \mathcal{U}_{i}^{β} by \mathcal{U}_{i}^{α} and subtracting gives

 $\mathcal{U}_{i}^{\beta} \frac{\partial t_{ij}}{\partial x_{i}} - \mathcal{U}_{i}^{\gamma} \frac{\partial t_{ij}}{\partial x_{j}} = \beta \left(\mathcal{I}_{\beta}^{2} - \mathcal{I}_{\alpha}^{2} \right) \mathcal{U}_{i}^{\gamma} \mathcal{U}_{i}^{\beta} \mathbf{A} \cdot \mathbf{A} \cdot$

Integrating over the body gives

 $\int \frac{\partial}{\partial x_j} \left[U_i^{\beta} t_{ij}^{\gamma} - U_i^{\gamma} t_{ij} \right] dV$ $-\int \left[t_{ij} \frac{\partial U_i^{B}}{\partial x_i} - t_{ij} \frac{\partial U_i^{A}}{\partial x_j} \right] dV$ $= \beta \left(\mathcal{J}_{\beta}^{2} - \mathcal{J}_{\alpha}^{2} \right) \left(\mathcal{U}_{i}^{*} \mathcal{U}_{i}^{B} dV \right)$

117.

A.5.

Using the divergence theorem and the boundary condition makes the first. integral on the L.H.S. of A.5 vanish, and the second also vanishes since

 $t_{ij} \stackrel{\alpha}{\rightarrow} \frac{\partial U_i}{\partial X_j} = t_{ij} e_{ij} = t_{ij} e_{ij}$ $\Pi_{s} \neq \Pi_{s}$

Hence since

 $\int U_i^{\alpha} U_i^{\beta} dV = 0$ A.7.

Assume that the solutions \mathcal{U}_i^{α} form a complete set, and consider the Hamiltonian, H, associated with an arbitrary displacement function

 $W_i = W_i e^{i\omega t}$

A.8.

A.9.

A.10

A.6.

satisfying

Wi=0 on Su

 $H \cdot V - T = \frac{i}{2} \int t_{ij} e_{ij}^* dV - \frac{i}{2} \int \omega_i \omega_i dV$

where f_{ij}^{*} and f_{ij}^{*} are the stresses and strain arising from W_i Writing these as f_{ij}^{*} and f_{ij}^{*} and f_{ij}^{*} and dropping the time dependent term

 $\hat{H} = \pm \left[\hat{t}_{ij} \hat{e}_{ij} dV + \pm \omega^2 \right] \delta W_i W_i dV$

A.11.

A.12.

A.13

and

 $\hat{H} = \frac{1}{2} \sum_{a'i}^{\infty} \sum_{b'i}^{\infty} \int a'a' t_{ij} e_{ij} dV$ $+\frac{1}{2}\omega^{2} \lesssim \int \lambda a_{\alpha}^{2} U_{i}^{\alpha} U_{i}^{\alpha} dV$

 $\hat{e}_{ij} = \hat{z} \quad a^* e_{ij} \quad \hat{t}_{ij} = \hat{z} \quad a^* t_{ij}$

Also since

 $\int U_i^* U_i^* = 0$ $\propto \neq \beta$ A.14. $-\frac{1}{\rho 5l_{*}^{2}} \int_{U_{*}} \frac{\partial t_{ij}}{\partial x_{i}} U_{i}^{\beta} = 0$ $or = \frac{1}{\sqrt{5l_{x}^{2}}} \int \frac{\partial}{\partial x_{j}} \left(t_{ij}^{x} U_{i}^{\beta} \right) dV + \frac{1}{\sqrt{5l_{x}^{2}}} \left(t_{ij}^{\alpha} e_{ij}^{\beta} dV = 0 \right)$

Using the divergence theorem and the boundary condition, the first integral vanishes giving that

 $\int t_{ij}^{\alpha'} e_{ij}^{\beta} dV = 0 \qquad \alpha \neq \beta$ A.15.

Hence

 $\hat{H} = \frac{1}{2} \sum_{n=1}^{\infty} a_n^2 \int \left(t_{ij} \epsilon_{ij} + \beta w u_i^* u_i^* \right) dV$

 $\int \frac{\partial t_{ij}}{\partial x_i} U_i^* dV = \beta 5 \int_{a}^{2} \int U_i^* U_i^* dV$

 $\int \frac{\partial}{\partial x_j} \left(t_{ij}^{\alpha} U_i^{\alpha} \right) dV - \int t_{ij}^{\alpha} \ell_{ij}^{\alpha} dV$ $= \beta \Pi_{\alpha}^{2} \Big(U_{i}^{\alpha} U_{i}^{\alpha} dV$

Divergence theorem and boundary conditions, imply the first integral vanishes, therefore

 $\int t_{ij} e_{ij} dV = -\rho 5 \sqrt{2} \int U_i U_i dV$ A.19.

A.17.

A.18.

120.

Thus

 $\hat{H} = \frac{1}{2} \sum_{i=1}^{\infty} a_{x}^{2} \rho(\omega^{2} - 5l_{x}^{2}) \left(\frac{u_{i}^{\alpha}}{u_{i}^{\alpha}} \frac{dv}{dv} \right)$

A.20.

and

 $\frac{\partial H}{\partial a_{n}} = a_{n} \beta \left(\omega^{2} - 5 \zeta^{2} \right) \left(u_{i}^{*} u_{i}^{*} dv = 0 \right)$ A.21. if a = 0 or $w^2 = 5T_{\omega}^2$

Thus the stationary values of the Hamiltonian, \hat{H} , give the frequencies of free vibration $\hat{W}^2 = \int_{-\infty}^{2} \alpha (x, y) dx$. This is the basis of the method.

APPENDIX 2 THICK TWISTED PLATE COMPUTER PROGRAM

MAIN

			PUBL	.10	AL,	BL,	AI,E	BI,E	TA, E1	AZTH	T,HT2	JARC.	.K.S M	• N • N 4	.NR.	NC.NN		•	
			al(8)	10		.18.	10.1	2.1.1		AD.P.	01183		E.) . C.O.	FF/R.					
			PUDG		000	5/1				1 A B F L	01127	· UNV		61121	1416	11)	· · · · · ·		
			PUBL	16		F \ 1	1101		i is nr				÷.,				•	1 101	1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -
			PVBL	16	221	200	1131	S(20)	11153	SDP(1),SBD	P(1)							
			PUBL	.1C	JFł	3).	JEDE	?(1) <i>i</i>	, R M , F	RX,FA	(2,30) . GA	(2,30) J F A D	P(2)	GADE	(2)	· .	
			PUBL	.1C	MAt	211	MADI	°(1)/	• I X (3	3,31,	IXDP(2)	-•						
		1. J. J.	PURI	1 C	RN.	NMA	(3)	NMR	133.1	MOP	1.1.NR	DPAT	v						
								141101		- WEL	1.1.1.10	0-11	,			· · · ·	· · ·		
			REAL		,														
			INTE	GE	K KN					·	•								a sector
			TNTE	GEF	₹ RM	PRX.	, F A i	GA											• * *
			TNTE	EGEF	R S									•	· ·			•	2.52
						- <u>1</u>				- •	••••							- -	
		RE													•				
		PIN	FGE	1	Jak	141		·	· · · · ·							•			
		STAP	د ب آب	INI	INT (M)] .	PIP 4	NH E	3, 66,	THEN	STOP	1							
		SUP	vi(:0+(CODP	115) I S (JPDVI	1 (C N #	CNDP.	151)1						•		
		SUP) v i e s	55.5	SSDP	,20	0);;	SUPDA	/icsE	B, CBDI	p,200) :	• • • •		· · · · ·	1997 - 19		· · · ·	
		CUDI	Ňi N.	1E	FDP	.35	2 SUE	20421	FA.F	ANP.	2.301								
		tual					201	1 21101	i i i i i	44.44	09.21	i						.	
		3051				, . ,	307.	JUFI		ANGAN	UFFEI	- 1 a							•
		SUP	γz()	(X)	XDP	131	333							•					
		SU	PDV	{ N/	1 A # N	MDP.	,3)	SUP	24161	MB N	80°,3); 👘						• • •	
		SUP) w 3 ((OEP	.0.	4 + = -	8,5,	0.4	•M)2.								• • • • •		r e î se e
		1 N 1 1	TEN	11	•••		-••	•••			•	e i e e i i	1						
			1 m = 11	- u a	. N M	•	• • •	• • •				4.1.1	a.						
						:			·· .										
		NAD	• N A • •			•					· · · · · ·								
		NN	A+NE	3 • N (- 1													• • •	
		RN=!	1N3	· .				•						• • • •	•••••••		••	• • • • • • • • • • • • • • • •	······································
		NMA	() =)	4A3 !	3 A M I	2)=	NBII	MAC:	3)=N(1	aine inte	••••							former to be
		NMB	11380	3 N N	4B (2	$\mathbf{b} = \mathbf{N}$	A J NI	1B(3)		Na.		· · · · · ·	1995 - 1997		· • · · ·			···· .~	ير مد ہے ا
		TNP				LAF		17.0								·			
							100												
				K.,															
		SPA	531		EA1	1-1		- 23.	1 N	1.02	.).01	. (451)1	~L *(*	1451	')'BL/	(145		
			TAP (11	45")	PHT		457) * K S I	') +) ;					•			an a	
		LINU	Es (2)) 1 1	PRIN	TCM	,2))	SPA	CES(;	3):IP	RINT(N, 5).	SPAC	EStay	1		وجود جروا دارد و		
		EPRI	INTE	31,4	4115	PAC	ES(:	371 1									+ 1 - F		
		EPR	INTER		4115	PAC	E\$(:	3) 1 F	PRIN	ffal .	ANISP	Acis	/ 3	PRINT					
		SPAC	FS(111	EPRI	NTE	ΉΤ.	11151	PACES		CPRIN	TILE		1.040					
		. : 27		0.		- B +	B				5.00	1183		1 NE 31	4 7 1				
			1.00					1.10-											
		HIBU	1+2#I	11.00	9118	C3 4	14 = 0	191.91	;				• · · ·						بد محمد المرد
		BL"	11				-	_			• • • • • •	· · · · · · ·							
		Als;	1/AL1	; ^ L :	=1;B	1=1,	1868	₿Ľ∎¦	11										
		HT 2 1		17 2 8	ETA 2	ET.	A+E1	14.	·					·					
		100		E	- 4 2							·							
	·		1.0																
		UNC	11=1/	310	NCZ	2		JUN	[]]=	/351		•					an sa		مرورت منهار
		COC	1=1-	• E T /	100	(Z)	=1/0	20(1)	3100	(3)=E	TAZ+C	0(2)	;CO(4)=1=1	2+87/	1	• • • •	· · · ·	
		CO(\$	\$y≠C()(2))*C0	(2)	100	6)=[ETAZI	C0(5	100110	7)=C	0(4)+	COLAS	11				•••••••
		001	33×E1	r A # (50(6	120	0(9)	= 00	(6)#(:0141	:001	03-E	TANCO	(7)+(015				
		coi	1140	:013	3340	018	1100	112			10011	2. 1	E =	A					
					1	. 21	. 1 2 9	· · · · ·	, - , - , 		1.21-	2146				5 T A # (0(2)1		
		1210				1 - 1				64] X (11618	31 [X	(1,3)	=471)	((Z)3	1)=51	•	مهري تدري در	arannas arain
		FU	(12)	12.	000	* * U	K. 11	1+14	STEP	°' 1' U	NTIL	31 PO	'IX(J	>I)=	[X{],	J31			
		CLE/	RJ														· · · · · ·	• • ··· •	er in min
	1	M=24	MJ 😳		: • ·		•••	· · · .			· ·								
			1.2.16	12) = M +	JET	3)=2			•			•	·····	· · · · ·		1.1.1. 7 .5.5	<u>a sining</u>	de la la
		97 X 4			ei ø	A. J.		- 19 4											
		UHA	NCZ	ð 🕴		• •												7 : # : : : 1 7	क्रांच प्रसंध्य
-		ENC	12			•					•						8 - A - A - A - A - A - A - A - A - A -	• • •	
-										· · · -				••••	****				

SESET

		PUBLIC AL, BL, AI, BI, ETA, ETA, HT, HTZ, ACC, KS, M, N, NA, NB, NC, NN
		PUBLIC JAA, IBAJCARA, IMAJNANARACO(15)ACN(5)ACOEF(5+14+21)
	2.	
	3	PUBLIC CODP(I)) UNDP(I), AH
		PUBLIC SS(200), SB(200), SSDP(1), SBDP(1)
	_	DUBLIC 15/33- JEDP(1)-RM-RV, FA(2,30)-GA(2-30)-FADP(2)-GADP(2)
	5	PUBLIC OF CALL AND ALL AND ALL OF A COMPANY AND A
· · · · · · · · · · · · · · · · · · ·	5 -	PUBLIC MA(2), MADP(1), IX(3,3), IXDP(2)
		PUBLIC PN.NMA(3), NMB(3), NMDP(1), NBDP(1),
	r	
	8	REAL KS
	0	INTEGER RN
	-	THITCON DIA DY FAAGA
- 10	D	INTEGER RETRACTOR
1	1 .	INTEGER R
	- -	A DEGINA
- 1 -	2	
- TANKI 1	3	REALPAX
1	4	JREAL SCHISTINTEGERSIJJKSHJZYYJIPJICI
· · · · · · · · · ·	•	
1	5	
in all get the A	Cotta - Lar	' <u>ARRAY'A.B(1'.'R,1'.'R),</u> CB(1'.'114);
- 승규는 것 같아요~		1APDAY1 SO(11, 12#11), ED(01, 12-1,01, 13), EE(01, 1N;01, 13);
17		
. 16	3	"INTEGENARIAN" 11, 12, 13 (1, 4/)
)	'FROCEDURE' EVENAB (A1, A2, HI);
		STYRING DOADS AT A2.
) n	TATAVALAN AT
21		* INTEGER* HI ;
		BCIN
23	,	· LICTEAFIAT KLJ
	1 <u>1</u> 1	PRGCEDUREPRESET(PPUJJPINIEGERPPUJ
•		
- 25	5	PECIN'
26	an in the second	• 1F+0+5+P-P*/*2*G1*0+3*THEN**G010*STM3
	فسأ ويصبقن	D-D-P//2:0=0///2:A(P/Q)=A(P/Q)+CW;/GCTO/REST
21		ANNA, AP-PA (47-110-02/47-11B(P, 0)-B(PA0)ACW1
28	la la serere	24W++++
	a di seri sel	REST · / END'
		FOPKINISTEP/1/UNTIL/3/DO//BEGIN/
30	فيشر جيورا	
31		NULL(A)K)K)
- e en erster sites	i se e en s	NULL(B/R/R))
	- in the second	among the relevant a summary of the at the
33)	$FOR' I = II(KI) + 1 \cdot STEP \cdot 1 \cdot UATLE \cdot II (KI + 1) \cdot DO'$
	140° N	*BEGIN*
و ماستىت سىرى 🖌		TD = A1 (VI 1), TD = A1(VI 2), VD = A1(VI 3), VD = A1(VI A),
37	,	$TD = A \left[(M_{2})^{2} O \right] = A \left[(M_{2})^{2} O \right] = A \left[(M_{2})^{2} O \right] = A \left[(M_{2})^{2} O \right]$
36	•	JS = A1(IJ,5); JC = A1(KJ,6); IC = A1(KJ,7);
	· · · · · · · · · · · · · · · · · · ·	ITEL AT 1001 3 POHENI IT - 6 PRISE! II - 1.
3		
30	3	FOR $I = 0$ STEP I UNTIL $M = I = 10$ DO
	• •	FOR' $J = \emptyset$ STEF' 1 UNTIL' $N = JD = JJ$ DO' BEGIN
·	• • • • • • • • • • • • • •	$\mathbf{T}_{\mathbf{D}} = \mathbf{T} + \mathbf{I}_{\mathbf{D}} + \mathbf{K}_{\mathbf{D}}$
•		15 = 1 + D + D
4	1	' IF 0-5* IE - IE'/'2 'GT' 0.3 'THEN' KC = 1 'ELSE' KC = 0;
internation and the Address of Address of the Addre	2	TOR! K = KC $SESP!$ 2 'HETEL! N = 1 = KD $IDO!$
4.	3	FOR $H = 0$ (SIMP, 1 OWIT, $N = 2D = 22$ (D) HOIL
	4	$\mathbf{Z} = \mathbf{I} + \mathbf{K}_{\mathbf{i}} \mathbf{Y} = \mathbf{J} + \mathbf{H} + 1_{\mathbf{i}}$
a la la compañía de la 🔒	g (1.1. 3)	₽=(.]+,ID)+H+I+ID+1=JF(K1);Q=(H+HD)+H+K+KD+1=JF(K1);
· •		
- 4	6 21 2 2 2	TEAL TINES OF THE TARGET OF THE T
alandar dir da 🛓	7	CW=CE(IC)+CGEF(JS/JC/Z)+SC(Y)+FD(I/ID)+FD(K/KD)+FE(J/J)+FE(H/UD))
		A1
- 注意的 175 44	9	·
an an an an an an 🖉 🔺	9	* 1Exxxx01.031HFNacotVs1x=x1x=010a1h2xENDs1
	·	RESET(P,Q):
5	1	FINETOTIENDTA e a construction de la constru
Statemente recente de la	2	FRID : FIND :
20 - 20 - 20 - 20 - 20 - 20 - 20 - 20 -		$\frac{1}{1000} + \frac{1}{1000} + \frac{1}{10000} + \frac{1}{100000} + \frac{1}{1000000} + \frac{1}{10000000000000000000000000000000000$
5	3	$IOU, W = IS(VI) \triangleq IOUTI, I OUTIP, IS(VI+1), IO,$
	€ 111	'BEGIN'
وني مانا المتحدة أعتبيت	* ••* ••••	$TD = A2 (KI 1) \cdot ID = A2(KI 2)$
7		$J D = A C \left(A D + j \right) J D = A C \left(A D + C \right) J$
50	5	JS = A2(KJ,3); JC = A2(KJ,4); IC = A2(KJ,5);
57		TEN KT TEA'S THENT II - O TELSET II - 1.
· · ·		
5	•	FOR $\mathbf{I} = \mathbf{p}$ STAP. I ONTLL, $\mathbf{m} = \mathbf{I} = \mathbf{ID}$ DO
59	9	FOR' J = Ø 'STEP' 1 'UNTIL' N - JD - JC 'DO' 'BEGIN'
ananarian ann a' a' a'	A 1	ITPL 0 5# T _ T 1/1 2 1001 0.3 IMPORT VO _ 4 IFTSEL VO _ 4-
0	Y	$\frac{1}{12} + \frac{1}{12} $
6	1 .	FOR' K = KC 'STEP' 2 'UNTIL' I 'DO'
		Amount in the house of the the the the second man
	2	$\mathbf{W} \mathbf{D} \mathbf{K}^{T} \mathbf{H} = \mathbf{O} \mathbf{T} \mathbf{S}^{T} \mathbf{S}^{T} \mathbf{M}^{T} \mathbf{T} \mathbf{T} \mathbf{T} \mathbf{M} \mathbf{M}^{T} \mathbf{L} \mathbf{T} \mathbf{N} \mathbf{T} \mathbf{M} \mathbf{M} \mathbf{T} \mathbf{T} \mathbf{M} \mathbf{M} \mathbf{T} \mathbf{T} \mathbf{M} \mathbf{M} \mathbf{T} \mathbf{M} \mathbf{M} \mathbf{T} \mathbf{M} \mathbf{M} \mathbf{T} \mathbf{M} \mathbf{M} \mathbf{M} \mathbf{M} \mathbf{M} \mathbf{M} \mathbf{M} M$
	2	FOR $H = g$ (ST2) 1 (MITL) $N = JD = JJ$ (DO) (BEGIN)
	2 3	Z = I + K; Y = J + H + 1;
6 	2 3	Z = I + K; Y = J + H + 1; $KD = ID; HD = JD;$
6		$\begin{array}{l} FOR^* H = \emptyset & (ST D)^* T & (DATTL)^* N = JD = JJ & (DO)^* (BEGIN)^* \\ Z = I + K_1 & Y = J + H + 1_1 \\ KD = ID_1 & (D) = M_1 T A TD + I = JF(KT) I B = (H + uD) + M + K + KD + I = JF(KT) I \\ P = (JA + UD) = M_1 T A TD + I = JF(KT) I B = (H + uD) + M + K + KD + I = JF(KT) I \\ \end{array}$
6		<pre>'FOR' H = Ø 'STAP' T 'UNTIL' N = JD = JJ 'DO' 'BEGIN' Z = I + K; Y= J + H + 1; KD=1D;HD=JD: P=(J+JD)*M+I+ID+1=JF(KI)2G=(H+HD)*M+K+KD+1=JF(KI);</pre>
6	2 3 4 5 5	<pre>"FOR' H = Ø 'STAP' T 'UNTIL' N = JD = JJ 'DO' 'BEGIN' Z = I + K; Y= J + H + 1; KD=ID;HD=JD: P=(J+JD)*M+I+ID+1=JF(KI);Q=(H+HD)*M+K+KD+1=JF(KI); *IF*P*LT*1*OR*Q*LT*1*THEN**GOTO*FINEE;</pre>
6		<pre>"FOR" H = 0 'ST D' T 'UNTIL' N = JD = JJ 'DO' 'BEGIN' Z = I + K; Y= J + H + 1; KD=ID;HD=JD; P=(J+JD)*M+I+ID+1=JF(KI)2G=(H+HD)*M+K+KD+1=JF(KI)2 *IF*P*LT*1*OR*G*LT*1*THEN**GOTO*FINEE; Cw=2*CE(IC)*COEF(JS*JC*Z)*SC(Y)*FD(I*ID)*FD(K*KD)*FE(J*JD)*FE(U*HD)2</pre>
		<pre>'FOR' H = Ø 'STAP' T 'UNTIL' N = JD = JJ 'DO' 'BEGIN' Z = I + K; Y= J + H + 1; KD=ID;HD=JD; P=(J+JD)*H+I+ID+1=JF(KI)IG=(H+HD)*H+K+KD+1=JF(KI)I 'IF'P*LT'I*OR*G*LT'I*THEN**GOTO*FINEE; CW=2*CE(IC)*COEF(JS*JC*Z)*SC(Y)*FD(I*ID)*FD(K*KD)*FE(J*JD)*FE(H*HD)I</pre>
		<pre>"FOR" H = \$ 'ST_2' T 'UNTIL' N = JD = JJ 'DO' 'BEGIN' 2 = I + K; Y= J + H + 1; KD=ID;HD=JD: *IF*P*LT*1*OR*Q*LT*1*THEN**GOTO*FINEE; CW=2*CE(IC;*COEF(JS*JC*Z)*SC(Y)*FD(I*ID)*FD(K*KD)*FE(J*JD)*FE(H*HD)\$ RESET(P*Q);</pre>
6		<pre>"FOR" H = Ø 'ST_P' T 'UNTIL' N = JD = JJ 'DO' 'BEGIN' Z = I + K; Y= J + H + 1; KD=ID;HD=JD; P=(J+JD)*M+I+ID+I=JF(KI)}G=(H+HD)*M+K+KD+I=JF(KI)} *IF'P*LT*I*OR*G*LT*I*THEN**GOTO*FINEE; Cw=2*CE(IC;*COEF(JS*JC*Z)*SC(Y)*FD(I*ID)*FD(K*KD)*FE(J*JD)*FE(H*HD); RESET(P*Q); FINEE****END*;</pre>
	2	<pre>'FOR' H = Ø 'STEP' T 'UNTIL' N = JD = JJ 'DO' 'BEGIN' Z = I + K; Y= J + H + 1; KD=ID;HD=JD; P=(J+JD)*H+I+ID+1=JF(KI);G=(H+HD)*H+K+KD+1=JF(KI); 'IF'P+LT'1+OR*G'LT'1*THEN**GOTO*FINEE; Cw=2*CE(IC;*COEF(JS*JC*Z)*SC(Y)*FD(I*ID)*FD(K*KD)*FE(J*JD)*FE(H*HD); RESET(P*G); FINEE****END*;</pre>
6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	2	<pre>'FOR' H = Ø 'ST_P' T 'UNTIL' N = JD = JJ 'DO' 'BEGIN' Z = I + K; Y = J + H + 1; KD=ID;HD=JD; P=(J+JD)*H+I+ID+1=JF(KI);G=(H+HD)*H+K+KD+1=JF(KI); /IF'P*LT']*OR'G'LT'I'THEN''GOTO'FINEE; CW=2*CE(IC)*COEF(JS*JC*Z)*SC(Y)*FD(I*ID)*FD(K*KD)*FE(J*JD)*FE(H*HD); RESET(P*Q); FINEE'*'*END';</pre>
6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6		<pre>"FOR" H = Ø 'ST 2P' T 'UNTIL' N = JD = JJ 'DO' 'BEGIN' Z = I + K; Y = J + H + 1; KD=ID;HD=JD; P=(J+JD)*M+I+ID+I=JF(KI);Q=(H+HD)*M+K+KD+I=JF(KI); 'IF'P*LT'I*OR*Q*LT'I*THEN'*GOTO'FINEE; Cw=2*CE(IC;*COEF(JS*JC*Z)*SC(Y)*FD(I*ID)*FD(K*KD)*FE(J*JD)*FE(H*HD); RESET(P*Q); FINEE****END'; 'END'; 'END';</pre>

. . . .

	n en
	이 같은 것은 것은 것은 것은 것이 있는 것이 같은 것이 같은 것은 것은 것은 것이 같이 많이 많이 없다.
דל	*IF'HI'EQ'O'THEN''FOR'I=1'STEP'1'UNTIL'NMA(KI)'DQ''BEGIN'
72	*IF*KI*NE*1*THEN**BEGIN*
73	SS(I+NMB(K[))=1/SURI(ABS(B(I)])))SB(I+NMB(K[))=1/SORT(ABS(A(I,I))))
- 74 - Subrianna 7875 -	PIEPKIPEOPIPTHENPPBEGINP
76	SS(1+NHB(K1))=1/SQRT(ABS(A(1,1)))SB(1+NMB(K1))=1/SQRT(ABS(B(1,1)))
77	'END';'END';
78	* IF * KI * NE * 1 * THEN * WATD(B, 1 + 6 * HI + KI = 1) * ELSE * WATD(B, 1 + 1 + 6 * HI + KI);
 79:	TE / IF/KI/NE/1/THEN/WATD(A,1,11+6+HI+K1)/ELSE/WATD(A,1,6+HI+KI-1):
80 8	'END';'END';
81	'PROCEDURE' ODDAB (A3,HI);
82 0	'INTEGENARRAY' A3; 'INTEGER' HI;
	FOR'KI=1'STEP' 1'UNTIL'3'DO''BFGIN'
	THE NULL(A,R,R);
	NULL(B,R,R);
2000 87 8 7 8 7 8	"FOR KJ = I3 (KI) +1 'STEP' 1 'UNTIL' I3 (KI+1) 'DO'
88-07 ·	
89	$ID = A_3(K_0, i)_1 JD = A_3(K_0, 2)_1 KD = A_3(K_0, 3)_1 HD = A_3(K_0, 4)_1.$
5 7V	$I = A_1(m_1, j_1)$ ($J = A_2(m_1, j_1)$) $I = A_2(m_1, j_1)$ $I = A_$
92	'FOR' I = Ø 'STEP' 1 'UNTIL' K - 1 - ID 'DQ'
93 5	'FOR' $J = \phi$ 'STEP' 1 'UNTIL' $J = 1 - JD$ 'DO' 'BEGIN'
9407	IE = I + ID + ID;
95	$1F 0.5* IE - 1E'/' 2'GT' 0.3 'THEN' KC = 1 'ELSE' KC = \emptyset;$
YO	FOR $H = KG$ (SEEP, 2 (MITL) $M = 1 - KD$ (DO)
	$7 = 1 + K_2 + Y = J + H + 1_2$
	P=21Q=311 IF+KI+NE+3+ THEN+P=11+ IF+K1+E0+1+T +EN+0=21
100	P=(J+JD)+H+I+ID+1-JF(P);0=(H+HD)+M+K+KD+1-JF(0)1
101	*IF*P*LT*1*OR*Q*LT*1*THEN**GOTG*FINIS
102	$C_{i} = CE(IC) * COEF(JS, JC, Z) * SC(Y) * $
107	FD(I,ID) * FD(K,KD) * FE(J,JD) * FD(H,HD);
	- 1 1
106	P=P//21Q=C//2+11A(P,G)=A(P,Q)+CH1+GOTO+FINI
a 107 a 1	\$1'.'P=P'//2+1;0=0'/'2;B(P,Q)=B(P,Q)+CW;'GQTQ'FINI;
108	'END';
109	JIF/KI/EQ/3/THEN//BEUIN/
	$P=P^{1/2}(a_0^{1/2}(A(P_{2}Q) = A(P_{2}Q) + CN)) + COTOPETNUS$
112	S2***P=P*//2+1;Q=Q*/*2+1;B(P,Q)=B(P,Q)+CW;*END*1
<u> </u>	FINI', ''END'J
<u> </u>	'END'; 'END';
والمراجع المستوجعة والمتعاد	M = M' / '2;
110	INTEG;
118	M=2*N3
119	JASJDSJUSM。 オロヘログリカリノテログリグリがT11ノフェルキリグロットSc/TNSP1++*アメリト
120	Provide StepP is the step is the state of the state of the step is
121	+FOR'J=0'STEP'1'UNTIL'N'DO'FE(J)0)*15
122	+FOR I=1 STEP 1'UNTIL' 3'DC'' BEGIN'
123	*FOR J=0'STEP'1'UNTIL'H=1'DO''BEGIN
125	W=1; FOR K=1'STEP' ' UNTIL' ' DO W=W+(J+K); FD(J,]) = W+A1++1;
126	ATOPA TOUR STEPAINTILANADOLABECTNA
127	H=11PFORPK=1PSTEPP1PUNTILPIPDAPW=W+CJ+K31FFC1+13=W+R1++13 +
128	PEND'S
129	END'S COMPANY OF A CONTRACT OF
- 130	IM=.14.1N=.10.1
132	MAXARR(250C);
133	COEFFI(CE);
المتعمد المعاجب علية	

. .

.

1	134	I1(1) = I2(1) = I3(1) = 0;
	135	'BEGIN'
	136	'INTEGERARRAY' A1 (1'.'53,1'.'7), A2 (1'.'47,1'.'5);
	137	MAXARR (800);
	138	RAFD $(A1,4,\phi)$; RAFD $(A2,4,1)$;
	139	MAXARR (2500);
	140	I1(2) = 9; I1(3) = 46; I1(4) = 53;
	141 -	12(2) = 13; 12(3) = 40; 12(4) = 47;
	142	EVELAB (A1, A2, ϕ);
	143	*END*;
		COEFF2 (CE);
	145	BEGIN!
	146	LINTEGERARRAY' A3(1'.' 109, 1'.'7);
	147	MAXARR (800);
	148	RAFD $(A3,4,3);$
	149	MAXARR (2500);
	150	I3(2) = 56; I3(3) = 76; I3(4) = 109;
	151	ODDAB $(A3, \phi)$;
1.4 A.A.	152	END, ?
	153	COEFF3 (CE);
		BECIN!
		'INTEGERARRAY' A1(1'.'7, 1'.'7), A2(1'.'16, 1'.'5);
الم المراجعة	150	MAXARR (800);
	157	RAFD (A1,4,4); RAFD (A2,4,5);
	150	MAXARR (2500);
	159	I1(2) = 1; I1(3) = 7; I1(4) = 7;
	100	I2(2) = 4; I2(3) = 12; I2(4) = 16;
•	101	EVENAB (A1,A2,1);
	106	"END";
	103	COEFF4 (CE);
	145	BEGIN'
	107	'INTEGERARRAY' A3(1'.'19, 1'.'7);
	147	LIAXARR (BOO);
	14.8	RAFD (A3,4,6);
<u>.</u>	140	EAXARR (2500);
-	107	13(2) = 10; $13(3) = 13$; $13(4) = 19$;
	1 7 0	UDAB (AS,T);
	+ / + 172 [°] - 2013 - 2014	* END* 5
	173	
• • • • • • • • • • • • • • • • • • • •	174	UDALN (S/)
	e r 🥆 🖉 🖓 🖓 🖓	· CALU ·

INTEG

	PUBLIC AL, BL, AI, BI, ETA, ETA2, HT, HT2, ACC, KS, M, N, NA, NB, NC, NN PUBLIC JA, B, JC, R, JN, JN, NAB, CO(15), CN(5), COEF(5, 14, 21) PUBLIC CODP(1), CNDP(1), AA PUBLIC SS(200), SB(200), SSDP(1), SBDP(1) PUBLIC JF(3), JFDP(1), RM, RX, FA(2, 30), GA(2, 30), FADP(2), GADP(2)	
	PUBLIC MA(2), MADP(1), 1X(3,3), 1XDP(2) PUBLIC RN, NMA(3), NMB(3), NMDP(1), NBDP(1)	
	REAL KS Integer RN Integer RM-PX-FA-GA	
	INTEGER R PROCEDURE'INTEG	•
	PBEGINP PREALPACS	
	PINTEGERPIPJIKJ PROCEDUREPSIMP(J)JPINTEGERPJJ	
	BEGIN *REALPROCEDURE*EVAL(XX)}*VALUE*XX}*REAL*XX} *BEGIN*	. •
	<pre>*REAL*X1,XJ,XK3 *IF*1*EQ*0*THEN*X1=1*ELSE*X1=SIN(XX/A1)**13 *IF*1*EQ*0*THEN*X1=1*ELSE*X1=SIN(XX/A1)**13</pre>	
	<pre>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>></pre>	
	+ 1F*K+EQ+O*THEN+XK=1+ELSE*XK=XX++K; =EVAL=x1+XJ+XK;	•
•	PENDIS PREAL PROCEDURE SIMPIN(FNX.AA.BB.FE.) 1	•
	YVALUE' & A, BB, EES 'REAL' AA, BB, EES	
	PREAL PROCEDUREP FNX3 Comment (X) ** Pvalue X** Preal X3	
	BEGINS REAL OLD, NEW, ENDS, EVENS, ODDS, HS	
	* {EAL*X, A, B, E} AxAA1B=BB;E#EE;	
	HE B-AJ	1996) 1
	/BEGIN/	
	AGOTOP ENDI	
	ENDS FNX(A); ENDS=ENDS+FNX(B);	•
	0DDS=(++B)/2+ 0DDS= FNX(0DDS);	
	EVENS ² OJ	
	AGAIN ⁹ .	
	EVENST EVENS+ODDSJ	
	ODDS ^I O;	
	ODDS= ODDS+FNX(X);	·
	NEW= (ENDS+2*EVENS+4*ODDS)*H+(1/6);	
	END * • *	
	SIMPIN=NEWJ *END*J	
	AC=0.5#ACC+AL+EVAL(AL); COEF(1.1.K)=SIMPIN(FVAL.0.0.0.AL.AC);	
	PEND ^a j	
	M=2*N; *FnR*1±n/\$TFP*1*UNTIL*4*D0**FnR*J±=8*STFP*1*UNT11*5*D0#	
	FOR K=0* STEP* 1* UNTIL* 2*M* DO* COEF(1, J, K)=0;	
•	<pre>#FOR*K=0*STEP*1*UNTIL*2+M*DO**BEGIN* 1=03</pre>	
	FOR JEN7'STEP'1'UNTIL'S'DO'SIMP(J);	
	1-2, "FOR" J==7,-5,-3,-2,-1,0,1,3"DO! SIMP(J)]	
	[=4]SIMP(-5)} [=1]	
	FOR J == 6" STEP" 1" UNTIL" 4" DO'SIMP (J) 3	
	1-J. "FOR" J=-4,-2,-1" DO" SIMP(J)]	
· .·	■ ↓ PND*J Astronomy and the second state of the second state o	
	'END'	

	COEFFI
er a suger en en e	$\frac{PUDCIC}{PUDCIC} = \frac{PUDCIC}{PUDCIC} = PU$
3	PUBLIC CODP(1), CNDP(1), AA
	PUBLIC SS(200)/SB(200)/SSDP(1)/SBDP(1)
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	PUBLIC HA(2), MADP(1), IX(3,3), IXDP(2)
7	PUBLIC RN, NMA; 3), NMB(3), NMDP(1), NBDP(1)
8	REALS KS TO FRANK THE PROPERTY OF THE PROPERTY AND THE REAL THE PROPERTY AND THE PROPERTY A
	INTEGER RH, RX. FA. GA
11	INTEGER R
12	PROCEDURE/COEFF1(CE)J/ARRAY/CEJ
14	CE(1)=CO(1)=CO(3)}
15	CE(2)=CN(1)+(5+CO(1)=6+CC(4)=7+CO(3)+6+CO(9)+2+CO(5)+CO(4)+CO(7)+
16	2+CC(8)}+HT2;CE(3)=C+8+(ETA=CO(8)=CO(10))+HT2;CE(4)=+CE(3);
18	CE(7)=0+0+(2+CO(9)+KS+HT2)CE(8)=34+CO(3)+CO(11)+HT2)
19	CE(9)=0+8+(2+ETA-2+CO(3)-ETA+CO(2)+CO(4))+HT23CE(10)=2+CE(8)3
20	CE(11)=8,5+CN(3)+CO(9)+HTZ/CE(12)=+2*CE(11)/CE(13)=+CE(12)/
21	CE(1+)=0+5+CO(4)+CE(15)=CH(1)+(5+5+CO(1)+(++CA)+HT2) CF(16)=0+8+ETA+CO(2)+CO(4)+HT2+CF(17)=CE(6)+CE(18)=CE(7)+(++CA)+HT2+CE(7)+(++CA)+HT2+CE(7)+(++CA)+HT2+CE(7)+(++CA)+HT2+CE(7)+(++CA)+HT2+CE(7)+(++CA)+HT2+CE(7)+(++CA)+HT2+CE(7)+(++CA)+HT2+CE(7)+(++CA)+HT2+CE(7)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(++CA)+(+(++CA)+(++CA
23	CE(19)=CE(11);CE(20)=CE(12);CE(21)=CE(8);CE(22)=CE(11);
24	CE(23)=5+CO(1)+4+ETA-9+CO(3); Constant of the state of th
25	$CE(24)=2_{4}4_{0}(9)+3_{3}$ $CE(25)=CN(1)+(-2*CO(12)+25*CO(1)-24*CO(13)+6*ETA+16*CO(4)-5)*CO(3)+6*ETA+16*CO(4)-5)*CO(3)+6*ETA+16*CO(4)-5)*CO(3)+6*ETA+16*CO(4)-5)*CO(3)+6*ETA+16*CO(4)-5)*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*ETA+16*CO(3)+6*ETA+16*CO(4)+5)*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO(3)+6*CO($
20	4+ETA+CO(2)+CO(12)+6+CO(6)+CO(12)+2+CO(4)+CO(13)+CO(13)+CO(5)+
28	24+CO(6)+CO(13)+18+CO(8))+HT2;
29	CE(20)=0+8+(9+ETA=9+CO(8)=3+ETA+CO(4)+CO(13)+CO(5))+HY21
31	CE(29)=0+8+(15+ETA+6+CO(3)=9+CO(8)=3+ETA+CO(4)+CO(13)+CO(5))+H+2:
32	CE(30)=153+CN(3)+CO(9)+HT2;
·· 33	CE(31)=2+ETA+4+CO(1)=6+CO(3);CE(32)=10+8+CO(9)+Ks;
35	CE(33)=CE(36)+CN(1)+(2*ETA=12*CO(4)=42*CO(3)+2*CO(5)*CO(12)+20*CO(1)+
36	12+CU(8)+4+CO(5)+CO(7)+CU(13)+8+CO(13)+18+CO(9)+C+CO(6)+CO(13))+HT2+
37	2.4+CO(3)+HT2;
30	CE(34)=0,8+(3+ETA+7+CU(8))+HT2; FF(36)=CE(36)+0+8+(-12+ETA+15+CO(8)-3+ETA+CO(7)+CO(5))+HT2+CF(-0)+
40	(-36+CN(2)+CO(9)+KS+HT2);CE(37)=CE(30);
41	CE(38)=3,6+CC(9)+KS;CE(39)=-12+CN(2)+CO(9)+KS+HT2;
42	<pre>CE(40)*0086(/*E)A*C0(3)=6*C0(8)*2*CTA*C0(4)*C0(13)*C0(5))*HT2; rr(4)*102*CN(3)*C0(9)*HT21</pre>
44	CE(42)=0+8+(4+ETA-CO(3)=ETA+CO(4)+CO(13)+CO(5)=3+CO(8))+HT23
45	CE(43)=51+CN(3)+CO(9)+HT2J
46	CE(44) = 0 + 0 + 0 + 0 + 0 + 0 + 0 + 0 + 0 + 0
48	CE(50)=0+8+(6+ETA-6+CO(3)=3+ETA+CO(4)+CO(2))+HT2;
49	CE(51) = 306 + CO(11) + HT21
20	CE(49)= 10+CH(2)=CO(1)+5+5+CO(4)+4+ETA)+HT2+CE(51)1
52	CE(50)=CE(50)+CE(51))
53	CE(51)=4+CE(51);CE(52)=CE(38);
54	CE(53) = 12 + 60 + CN(3) + CO(11) + H(2) $CE(53) = +12 + CN(2) + CO(9) + KS^{+}H(2 + O + A + (8 + ETA + 8 + CO(3) - 4 + ETA + CO(4) + CO(3) + 4 + CO(3) + CO(3$
56	HT2+0,5*CE(54);
57	CE(56)=408+CN(3)+CO(11)+HT21+date of the second descent descent descent descent descent descent descent descent
50 FERTING 80	CE(55)=U+8+(2+E(A+2+CU(3)+CU(4)+ETA+CU(2))+HT2+U+5+CE(56)]
ου το	CE(58)=0+8+(6+ETA=3+CO(3)=3+CO(8)=3+ETA+CO(7)+CO(5))+HT21
61	CE(58)=CE(58)+CE(57);
62	CE(57)#CE(57); CF(60)#20CN(1)#CE(57);CE(61)#0.54CE(60)*CF(60)*CF(6)*****
	CE(57)=0,5+CE(57)3
arten annañ an 65	CE(63)=CE(38);
66	CE(64)=CE(60)=JZ+LN(Z)+CO(9)+KS+HTZ+0+8+(ETA+CO(3)=2+CO(8)= 2+FTA+CO(7)+CO(5))+HT2+
67 &8	CE(65)=CE(60)JCE(66)=CE(61)+0.8+(ETA=CO(8)=ETA=CO(7)+CO(5))+HTat
69	CE(67)=CE(61);CE(68)=CE(61)+0.8+(2+ETA-CO(3)=CO(8)=ETA+CO(7)+CO(5))+
70	HT2;CE(69)=CE(61);CE(70)=CO(1)=CO(3);CE(71)=CE(24);
71	CE(74)=0+8+(LU(3)=LU(0)++12+18+CN(2)+CO(9)+KS+H12}
73	CE(75)=0.5+CE(30)
74	CE(70)*CE(70)+0.5*CE(30);
- 75	CE(73)#UR(1)+(5+CU(1)+6+CU(4)+7+CO(3)+2+CO(8)+2+CO(7)+CO(4)+CO(5)+
70	CE(74)=CE(74)=4+8+HT2+(ETA+CO(8))+2+CE(75)1
78	CE(76)=0+6+CO(9)+KSJCE(77)=-2+CN(2)+CO(9)+KS+HT21
79	CE(75)=34+CN(3)+CO(9)+HT2;CE(79)=CE(80)=CE(78);
्र <del>प्राह्म सम्</del> 80	LE101 = U+25+ LE101 + LE102 = U+3+ CE(70) + CE(83) + CE(76) } CF(84) + CE(77) + CE(85) + 544+ CN(3) + Co(1) + 0+7 + CE(85) + CE(8
82	CE(87)=CE(81);CE(88)=34+CN(3)+CO(11)+HT2;CE(89)=2+CO(4);
83	CE(90)=CN(1)+HT2+(18+ETA-14+CO(1)+18+CO(4))
84	CE(91)=CN(1)+HT2+(6+CO(1)-2+ETA=2+CO(4)); CE(92)=CC(91)+CE(93)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE(91)+CE
66	CE(95)==2+CE(94);CE(96)=CN(1)+CO(1)+HT21
87	CE(97)=CN(1)+HT2+(2+CO(1)-4+ETA);CE(98)=2+CN(1)+ETA+HT2;
88	CE(100)=CN(1)+HT2+(5+CO(1)=4+ETA))CE(101)=CN(1)+HT2+(2+ETA=4+CO(1))
57 67	

· .			128.
•		COEFF2	
_ 1	1	PUBLIC AL, BL, AI, BI, ETA, ETA2, HT, HT2, ACC, KS, M, N, NA, NB, NC, NN	•
- 1.	2 3	FUBLIC JA, JB, JC, R, JM, JN, NAB, CO(15), CN(5), COEF(5, 14, 21) PUBLIC CCDP(1), CNDP(1), AA	
	4 8	PUBLIC SS(200)+SB(200), SSDP(1), SBDP(1)	a ata
	6	PUBLIC MA(2), MADP(1), IX(3, 3), IXDP(2)	•
	* 7 1 8 - 1	PUBLIC RN, NMA(3), NMB(3), NMDP(1), NBDP(1) RFAL Ke	·
	9	INTEGER RN	
•	10 11 ·	INTEGER RHJRX/FA/GA	
	12	PRUCEDURE COEFF2(COD); ARRAY COD;	
	14 *	CoD(1)=2+Cc(1)+4+ETA=6+Cc(3);	
-	15 · · · ·	COD(2)=CN(1)+HT2+(10+CO(1)=2+CC(12)=6+CO(4)=8+CO(13)+16+ETA=38+CO(3) 4+ETA+CO(4)+CO(2)+18+CO(9)+2+CO(6)+CO(12)+4+CO(7)+CO(13)+CO(5)	)+
	17	8+Cv(6)+Cv(13)+12+Co(8)); Cv(6)+Cv(13)+12+Co(8));	
	19	COD(5)=2+ETA-2+CO(3); COD(5)=2+ETA-2+CO(3);	
	20	COD(4)=0+8+HT2+(3+ETA=3+CO(8)=3+ETA+CO(7)+CO(5)); COD(6)=CN(1)+HT2+(6+ETA=8+CO(4)+4+FTA+CO(4)+CO(7)-10+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)+CO(7)	
م در ده ه	22	12+C0(9)+4+C0(7)+C0(4)+C0(5))+0+8+HT2+(6+ETA-3+C0(7)+ETA+C0(5)+	
	23 24	6+C0(8)); CUD(7)=2+4+HT2+(ETA=CO(8));	
	25	COD(8)=0.8+HT2+(6+ETA-6+CO(3)-3+ETA+CO(7)+CO(5)); COD(8)=0.8+HT2+(6+ETA-6+CO(3)-3+ETA+CO(7)+CO(5));	· • •
	27	COD(10)=0+8+HT2+(2+ETA-CO(3)=C0(8)=ETA+CO(7)+CO(5));	а.
	28	CuD(11)=U+B+HT2+(ETA=CO(8)=ETA+CO(7)+CO(5)); CuD(12)=3+6+KS+CO(9);CuD(13)=4+CN(3)+Cu(9)+KemHt2+	
	30	COD(15)=403+CN(3)+CO(11)+HT2;	
•	31 32 ·	CUD(14)=C+5*COD(15)+C+8+HT2*(2+ETA+2*CO(3)+ETA+CO(4)*CO(2)); COD(16)=CU(1)*COD(12);COD(17)=CN(1)*COD(13);	
	33	COD(18)=272+CN(3)+CO(11)+HT2;CCD(19)=0+25+COD(18);	
	34	COD(20)=0+6+HT2+(5+ETA+2+CO(3)+3+CO(3)+ETA+CO(4)+CO(13)+CO(5)); COD(21)=51+CN(3)+CO(9)+UT2:COD(22)+COD(3)+C)	
	36	COD(23) *3+8+HT2+(2+ETA-CO(11)-ETA+CO(7)+COD(21);	
•	37 35	CoD(24)=CGD(21);COD(25)=2*CN(1)*COD(21);COD(26)=0,5*COD(25); CuD(27)=COD(26);COD(28)=CO(4);	• •
· · ·	39	COD(30)=1+6+HT2+(6+ETA-6+CO(3)=3+ETA+CO(4)+CO(2));	
· · · ·	40	COD(31)=3+2+HT2+(2+ETA+2+CO(3)+ETA+CO(4)+CO(2));	
	42	COD(32)=0.25+COD(31);COD(33)=COD(12);COD(34)=COD(13); COD(35)=0.8+HT2+(=5+FTA+2+CO(2)+2+CO(3)+=COD(13);	
. • · · · ·	44	CoD(36)=-51+CN(3)+CO(9)+HT2;COD(37)=COD(36);COD(36)=COD(12);	} <b>}</b> ∓ .
ere in ge generation e	45	CoD(39)=COD(13);COD(40)=0+8+HT2+(-2+ETA+CO(11)+ETA+CO(7)+CO(5))+ CoD(36);	
	47	CoD(39)=CoD(39)+CCD(40);	÷.
	48 49	COD(41)=CVD(36);CCD(42)=CN(1)+COD(12);COD(43)=CN(1)+COD(13); COD(44)=Z+CN(1)+COD(36);COD(45)=0+5+COD(44);COD(46)=COD(45);	•
•	50	Cop(48)=40c+CN(3)+CO(11)+HT2; Cop(47)T0, CoHT2+(2+FT4=2+CO(3), F=4+CO(4), CO(4))	•
	52	COD(49)=272+CN(3)+CO(11)+HT2;CGD(50)=0+25+COD(49);	•
•	53	$C_{0D}(5_1)=0.6+HT2+(4+ETA-CC(3)-ETA+CC(4)+CO(13)+CO(5)-3+CO(8));$	
	55	COD(54)=0.8+HT2+(ETA-CD(8)-ETA+CG(7)+CG(5))+COD(52);	
	56 57	COD(55)=COD(52);COD(56)==CUD(44);COD(57)=0.5+CQD(56); CUD(53)=CUD(57);	
	58	COD(59)=CN(1)+HT2+(=6+CO(1)+2+ETA+4+CO(3)+4+CO(7)+CO(2));	
• • *	59 60	CUD(60)=-CCD(59); CUD(61)=0+8+HT2+(6+ETA-2+CQ(3)=2+ETA+CO(4)+Cc(2));	
•	61	COD(62)=0-8+HT2+(ETA+CO(3));COD(63)==COD(62);COD(64)=COD(62);	
	63	COD(65)==0,0+H12+C(4)+CO(4)+CO(2);COD(66)==COD(65);CGD(67)=2+CO(4); COD(65)=CN(1)+H12+(4+CO(1)-4+E1A-CO(4));	-
	64	COD(70)=2*CH(1)*HT2*CO(4)+COD(69)\$COD(71)=-CN(1)*HT2*CC(4);	
•	66	COD(74)=COD(88),CCD(73)=CO(85); COD(74)=C++T2+(6+ETA=2+CO(3)=2+ETA+CO(2)+CO(4));	•
	67	COD(75)=0.8+HT2+(ETA+CO(3));COD(76)=+COD(75);COD(77)=COD(75);	•
	69	COD(8)=CN(1)+HT2+(+18+CQ(1)+6+ETA+12+CQ(3)+4+CQ(4)+CQ(13)+CQ(7));	
	70	COD(61)=3+COD(65);COD(82)=+COD(80);COD(83)=+COD(81); COD(34)=2+(O(4);COD(84)=12+COD(80);COD(83)=+COD(81);	
	72	CuD(35)=G.5+CQD(86);COD(88)=4.8+HT2+(ETA+CO(3));	•
	73 - 74	CoD(90)=-CoD(88);COD(87)=+CN(1)+HT2+CO(4)+C+S+COD(88); CoD(89)=CN(1)+HT2+(8+FTA+CO(4)-8+CO(1))+O+S+COD(80);	
	75	CoD(91)=CN(1)+HT2+(4+CD(1)-CO(4)-4+ETA)+C.5+COD(88);	
	70 · · · 77	COD(72)=COD(88);COD(73)=COD(81);COD(74)=COD(83); COD(75)=CUD(81)+2+CN(1)+HT2+(ETA+2+CO(3)+2+CC(7)+CC(7)-3=CO(1)+++	
· · · · ·	78	CoD(96)=CoD(81);CoD(97)==COD(95);COD(98)=COD(83);COD(99)=2*COD(65); COD(100)==COD(82);COD(97)==COD(95);COD(98)=COD(83);COD(99)=2*COD(65);	
1	80	COD(1u3)=6.4+HT2+(3+ETA=CO(4)+CO(14)=CO(3));	n en en
· · ·	81 82	COD(104)=3.2+HT2+(ETA+CO(3));COD(105)=-COD(104);COD(106)=COD(104); COD(107)=COD(65);COD(108)=COD(40);COD(106)=COD(104);	
	83	CoD(110)=0.25+COD(104);COD(111)==COD(110);COD(112)=COD(110);	
÷	85 - 5	LOD(13)=COD(13)+COD(14); ************************************	

·

.

•.

COEFF3

•		PUBLIC ALOBLOAIOBISETASETAZOHTOHTZOACCOKSOMONONAONBONCONN
	2	PUBLIC JA, JB, JC, R, JM, JN, NAB, CO(15), CN(5), COEF(5, 14, 21)
	3	PUBLIC CODP(1),CNDP(1), AA
	4	PUBLIC SS(200),SB(200),SSDP(1),SBDP(1), SHOP(1)
	. 5	PUBLIC JF(3), JFDP(1), RM, RX, FA(2,30), GA(2,30), FADP(2), GADP(2)
	6	PUBLIC MA(2), MADP(1), IX(3,3), IXDP(2)
	7	PUBLIC RN, NMA(3), NMB(3), NMDP(1), NBDP(1)
	8	A REAL KS
	9	INTEGER RN
	10	INTEGER RM, RX, FA, GA
	11	INTEGER R
	12	PROCEDURE COEFF3(CE); ARRAY CE;
	13	/BEGIN/
	14	CE(1)=1+0;CE(2)=-0+8*HT2*CG(14);CE(3)*CH(1)*HT2*CO(6);
	15	CE(4)=17+CN(3)+HT2+CO(6);CE(5)=CE(4);CE(6)=1+153+CN(3)+HT2+CO(4);
	16	$CE(7) = 3 + H_72 + CO(6) ICE(8) = -2 + 4 + H_72 + CO(14) I$
x; ·	17.	CE(9)=2+HT2+CO(6)=2+4+HT2+CO(14):CE(10)=306+CN(3)+HT2+CO(6):
	18	CF(11)=CE(2);CF(12)=102+CN(3)+HT2+CO(6);
	19	CE(13)=0.5*CE(10);CE(14)=CE(12);
	20	CE(15)=CE(3);CE(16)=CE(13);CE(17)=CE(12);CE(18)=CE(4);CE(19)=CE(4);
e în g	21	CF(20)=1+(;CE(21)==CN(1)+HT2;CE(22)=CE(23)==CE(21)1
مودين للمحار بم الم	22	/FND/



	1	PUBLIC AL, BL, AI, BI, ATA, LTA, LTA, HT, HTZ, ACC, KS, M, N, NA, NB, NC, NN are not in diseased different statements of the second statements of
	2	PUBLIC JA, JB, JC, R, JM, JN, NAB, CO.(15), CN(5), COEF(5, 14, 21)
문화 감독은	3	PUBLIC CODP(1), CNDP(1), AA
ى دەمەر يەت يېكىكىتى تېتىر 19		PUBLIC \$\$(200), \$B(200), \$SDP(1), \$BDP(1)
	5	PUBLIC JF(3), JFDP(1), RM, RX, FA(2, 30), GA(2, 30), FADP(2), GADP(2)
	6	PUBLIC MA(2), MADP(1), IX(3,3), IXOP(2)
an a	7	PUBLIC RN, NMA(3), NMB(3), NMDP(1), NBDP(1)
STIL THE ALL AND AND	8	REAL KS
	9	INTEGER RN STATES STATES IN THE STATES AND A STAT
متا المسيد والمستقار وال	10	INTEGER RH, RX, FA, GA
	11	INTEGER R
a a serie de la	12	PRUCEDURE/COEFF4(COD)
ومربعين سريد سر	13	ARRAY'COD:
	14	'BEGIN'
مسر ، جار المراجع . ا	15	COD(1)=-?,4*HT2*CO(14);COD(2)=CN(1)*COD(1);COD(3)=2*HT2*CO(6);
مساعيدتيا متشايد جي	16	COD(4)=CN(1)+COD(3);COD(5)=102+CN(3)+HT2+CO(6);
	17	CoD(6)=CN(1)+CoD(5);COD(7)=COD(2);COD(8)=CoD(5);COD(9)=COD(5);
الم المعالية المعالية الم	18	C(D(10)=C(D(6);COD(11)=-C(D(2);COD(12)=COD(11);COD(13)=-2+CN(1)+HT2;
	19	COD(14)=CCD(13)1CCD(15)==COD(1)1CCD(16)=COD(15)1CCD(17)=COD(15)1
فاحطا للمساء فيناتي		
· .•	20	COD(12)=COD(14)=COD(11); end of each state of the contract manual state of the stat
	21	· *END?

BCSUB

PUBLIC AL, BL, AI, BI, ETA, ETA2, HT, HT	ZJACCJKSJM,N,NA,NB,NC,NN	•
PUBLIC JA, JB, JC, R, JM, JN, NAB, CO(15	), CN(5), COEF(5,14,21)	
PUBLIC CODP(1) CNDP(1), AA		• • • •
PUBLIC SS(200), SB(200), SSDP(1), SB	DP(1)	•
PUBLIC JF(3), JFDP(1), RM, RX, FA(2,3	0), GA(2, 30), FADP(2), GADP	(2)
PUBLIC MA(2) + MADP(1) + IX(3+3) + IXDP	(2)	
PUBLIC RN, NMA(3), NMB(3), NMDP(1), N	BDP(1)	
RFAL KS		
INTEGER RN	•	·
INTEGER RMARXAFAAGA	•	
INTEGER R		-
*BEGINF * INTEGER * I. J.K. H. HII	•	
/INTEGER/KY:		
#ARRAY#A(1***NN+1***NN)+B(1***R+1***R	11	* .
PROCEDURE FULL(EFAR) I APRAY FEITINTE	GER/R1	
AREGINE INTEGERTING		•
FORTALISTEPTIONTILIRAL DOPTEORIAL	STEPFIFUNTU PRIDOF	
EF(J.1)xEF(1.1)}	old to delte webus	
JENDI:		
MAXARR(2500)3		
*FOR*KY=0,2*DO**BEGIN*	<ul> <li>Me e e e e e e e e e e</li> </ul>	97.99 AN
NMB(1)=0;NMB(2)=NAJNMB(3)=NA+NB;	•	
FOR HIFO'STEP' 1' UNTIL' 1'DO''BEGIN	<ul> <li>A second sec second second sec</li></ul>	* •
+FOR 1=1+ST(P+1+UNTIL+3+D0+		
+FOR+J=I+STEP+1+UNTIL+3+DO++BEGIN+		•
IF'IFEDIJ'THENFK=I-1FELSEFK=I+JJ		•
RAFD(8,1,6*KY+6+HI+K);	•	•
FOR K=1 STEP 1 UNTIL NMA(1) DOP FOR	H=1"STEP"1"UNTIL "NHACA)	001
A(K+NMB(1)+H+NMB(J))=B(K,H);		
PEND 3	•	
FULL(A, NN)3	•	
MAXARR(5000);		
WATD(A,2,KY+H1+2)1		•
MAXARR(2500)3		ę .
/END/:	•	`
PEND ICHAIN(5)	•	· .
PEND/		· c
	•	•

		SOLVE	•••
	1220	PUBLIC AL, BL, AI, BI, ETA, ETA2, HT, HTZ, ACC, KS, M, N, NA, NB, NC, NN	
	2	PUBLIC JA, JB, JC, R, JM, JN, NAB, CO(15), CO(5), COEF(5, 14, 21)	
ی جست میں میں میں ایک اور	4 - 10010 4	PUBLIC SS(200), SB(200), SSDp(1), SBDP(1)	🕐 and a think is a sum
م. م. م. م. م. م. م	5	PUBLIC JF(3), JFDP(1), RM, RX, FA(2, 30), GA(2, 30), FADP(2), GADP(2)	in the second second
	0 7	PUBLIC MA(2), MADP(1), IX(3,3), IXDP(2) PUBLIC RN, NMA(3), NMB(3), NMDP(1), NBDP(1)	
, in the second s	8	REAL KS	
ر. 1910 - میکند میکند میکند.	9 n	INTEGER RN - Constant of the second	ار ایرانی محکوم کی ایرانی سال بی افرانی محکوم کی ایرانی ایرانی
1	1	THE INTEGER R AND A THE AN	
1	2	ABEGINA	
المستعدية المساد	4	INTEGER' NS,KY,RD;	•
	5	'ARRAY' EV(1'.'RN), A(1'.'RN,1'.'EN);	•
	0. 7	NS = N TNREAL (E.RO):	
1	8	ININT (BD);	
	9 0	PI = 3.141592654;	
2	1	SC = $1.0/(2*PI*AA)*$ SCRT (E*G/ (RO* (1-2*ETA)* (1+ETA)));	
2	2	$^{+}FOR^{+}KY = 0,2$ $^{+}DO^{+}$ $^{+}BEGIN^{+}$	
. <u></u>	4	ABEGINA PREAL SUWIP INTEGER IN JAKAWANXANYI	•
2	5	#ARRAY#D(1".*RD,1".*NN)}	
2	6	PROCEDURE'EVALU(A)X); ARRAY'A; INTEGER'X;	
2	5	PREAL*AX, BX	per de la construcción de la constru
21	9	INTEGERIKI	
3(		JEORALANX; BX=J+BL/NY;	
3	2	SUM=03	
3	]	<pre>&gt; IF 0 R + 1 + STEP + 1 * UNTIL + NS* C &gt; SUM = SUM + A (1, H) + BX + + H;</pre>	وروار المتعلقة المراجع
- Des rainebre 3	<b>5</b> 19 .54	+FOR+K=2+STEP+1+UNTIL+H+DG++FOR+H=1*STEP+1+UNTIL+NS+DO+	•
3(	6	SUM=SUM+A(K+H)+AX++(2+K+2)+BX++H;	
3	/ 8	+ IF + KY + EQ + 2 + AND + X + NE + 1 + OR + KY + EQ + 0 + AND + X + FQ + 1 + THEN + + BEGIN +	n an
3	9	>FOR * K= 1 * STEP * 1 * UNTIL * M* DO * * FOR * H= 1 * STEP * 1 * UNTIL * NS * DO *	الدينية المعالم المراجع المراجع المراجع المراجع والمراجع الم
and the second s	0	SUMISUM+A(K)H)+AA++(2+K+(/+DA++H)	1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -
- 1911 - <b>4</b>	2	A1(1+1, J+1)=SUM3	a a secondaria de la composición de la Composición de la composición de la comp
4	3	1 INEC(4): EMATPR(A1) NX+1, NY+1, 4)'	بيدي ومسيسي يأدم
- 1943 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 1944 - 194	5	JEND/J CONSTRUCTION OF A CONSTRUCTURA A CONSTRUCTURA A CONSTRUCT	
4	5	PROCEDURE'SCALE(U); ARRAT'D; BEGIN; PFAL'SUM; AB;	ويتعارفهم والمتعارفة
41	7 B	FOR I= 1 - STEP - 1 - UNTIL - RD - DO - BEGIN	
49	9	SUN=0:	······································
50	0	AREARS(D(I));	
- Standard St	2	IF'AB'GT'SUM'THEN'SUMEABS	teren en en
5	3	JEND/J climiti / climit	
	5 - 1 - 1 - 1	FOR J=1+STEP+1+UNTIL+NN+DO+D(1+J)=D(1+J)+SUM5	an managan di Karangan di K
5	5	IENDIJEND'S CONTRACTOR OF A CONTRA	
20	/	THAND (A, D, LU, HN, LI);	
51	?	ININT(NX,NY);	and the second
60	) 	LINES(1);	
61	2	SPACES(1);TEXT(*(*FR:0*(*S*)*AND*(*S*)*MDDES+)*)\$	а 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 — 1910 —
6	3	BEGINS ARRAYSASBSC(10, MS1' OFNS)	
6	5	FOR/KEISIEPII'UNIIL'RD'DU''BEGIN' FOR/IEISTEP/I'UNTIL'N'DO''FOR/IEI'STEP/I/UNTIL'NS/DO/FRFGIN/	المهم جمر المدام أدار محاكمة
6		IF'J'GE'NS'THEN''GOTO'U3	
61		A(1)J)=D(K/(J-1)+M+I); B(1, ()-D(V,NNB(2)+()=1)+M+T)+	and the second secon
69	,		a hi i shi a ka
	P E E E	C(I)J)=D(K)NMB(3)+(J=2)+M+I)I	
7		FORPIELPSTEPP1PUNTILPMPDOPACT.NELEBCTANELECE	- in state in state of the second
			and alt direct

•	
73-	+FOR+1=1+STEP+1/UNTIL+M*DO*C(1,1)=01
- FILE 74	LINES(1);
75	TEXT(*(*FREG*(**)*)NUMBER*)*)]SPACES(1)]IPRINT(K,2)]
70	SPACES(*); EFRINIVEV(N); *); so the second sec
7.8	FVALU(B/2)3 1918-00 STED041+30045310405-00 STED0405-00 STED0405-00 STED0405-00 STED0405-00 STED0405-00 STED0405
79	The EVALU(C,3)3 Contact (States) - A second s
80	LINES(6) J. C.
81	ENDITARY STATES AND
82	JENDJJPLND"J Acnodi Acnodi 1998 - Takan Angeler and an and an and an and an and an an an and an an and an and an and an and
841	CHAIN(1); The WE defines a free for the second s
851	STATISTICS FROM THE STATE OF A STA
1.2	에는 것은 것은 것은 것은 것은 것은 사람이 있는 것은 것은 것은 것은 것은 것은 것을 가지요. 같은 것은
	에는 것은 것 같아요. 이 것은 사람을 해야 한다. 이 가지는 것을 수 있는 것은 가지는 것이 가 이 가지 않는 것은
\$ ?	
• * 1 V	
· · · ·	
19	· 我想到了我,我们要要你们的意思。""你们,你们们的你们,你们们的你们,你们们的你们,你们们不知道你们,你们们不知道你们,你们不知道你们,你们不知道你们,你们不
2	"我们还是不是我最有意义,你把我们的问道,你是你是你说道:"你们的你们的?""你们还不知道你说,你们还是你们,你不知道,我们不知道,你们不知道你?""你们不知道。 "我都没有我们,你我们们就是我我们就是我们的?""你说你,你们就是我们的你?""你们你你们,你们们还是你们们的你们,你们们不是你们的?""你们们你们们不知道。""
	,我们就是你们的问题,我们就是你的,我们就是你的你的。""你们,你们就是你们的你们,我们们们的你们,你们不是你们的你们不是你的?""你不是你的?""你们,你们不是 "你们我们们,我们就是你们,我们就是你们的你们,你们们就是你们的你?""你们,你们们们们们,你们们们们们们不是你们的?""你们们,你们们们们们们们们们们,你们们们
	ENVLOW MANAGEMENT MARKET STREET, SUCCESSION AND AND AND AND AND AND AND AND AND AN
·	If the two interview and the second state of the second state of the second se Second second sec
15	PROCEDURE INVLOW(A,L,N); ARRAY A,L; INTEGERIN; Construction of the second statement of the second se
2	BEGIN''PCAL'SUMI'INTEGER'I, J, KINULL(A, N, N)
3	FOR' I = I'STEP' I' UNTIL'N' DU'A([, I) = I'L'(I, I);
6	A(I)J)*-SUM*A(I)J} cost for fairful as an end of the first statement of the second statement of
T	n na PENDY (see a second and investigation and a second respectively) and a second second second second second
<del></del>	'END'
3.2	
1. A. 1. A.	
	かって、「夏秋橋原来省森舎市場」を行きた時にでは、水水・ボックボールとして、「夏水・水」にして、「夏水・水」になっていた。「夏水・水」になって、「夏水・水」として、「「「 「夏水・水」として、「夏水・水」として、「夏水・水」として、「夏水・水」として、「夏水・水」として、「夏水・水」として、「夏水・水」として、「夏水・水」として、「夏水・水」
1	"你们们,我看到这些意义像是不是是你说这个,我们们们的你,他们们的你,这个不知道你,你们这个人,你们不是你的?""你们,你们们们们,你们不是你的?""你们,你不是 我们们们,我有什么?"
	MAINL
1	PROCEDURE, MATHLIL, A, N); ARRAY, L, A; INTEGER, N;
2 1	BEGIN''INTEGER'IJJKJARRAY'EE(1'. N); see a see
3	FOR JEISTEP JOHILLNYDOU'BEGIN
	FF(J)=C2FFOR [#] K=1 ² STEP ² 1 ² UNT1L ² 1 ² D0 ² FE(J)=FF(J)=A(1, V)=(/V)
6	'FOR' J=1'STEP'1'UNTIL'N'DO'A(1, J)=EE(J): ENDI
7	PEND, Contraction of the second se
· · · · · · · · · · · · · · · · · · ·	n an the applied with the second state of the second state of the second state of the second state was made an The second state of the second s
i.e	
	IRAML
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	PROCEDURE TRAML (L.A.N.) : ARRAYIL.A: INTEGERIN:
2	BEGIN'' INTEGER' I, J, KJ ARRAY'EE(1' + N) J
3	FOR'I=1'STEP'1'UNTIL'N'DO'BEGIN'
- P. 1. T. 1. T. <b>4</b> j. 10	'FUR'JEI'STEP'I'UNTIL'N'DO''BEGIN'
<b>5</b>	EE(J)=U)=FOR(K=1-S)EF(]=UNT1L=U=DU=EE(J)=EE(J)=E(J)=E(J)=E(J)=E(J)=E(J)=E(
· · · · · · · · · · · · · · · · · · ·	FUN UNALSIEFTI UNITALUTALITATEN VII ANDRES VII ANDRES ANDRES ANDRES ANDRES AND
e e e e e e e e e e e e e e e e e e e	
	n an

## HOUSEHOLDER SOLUTION

	PUBLIC AL, BL, AI, BI, ETA, ETAZ, HT, HT2, ACC, KS, M, N, NA, NB, NC, NN
	2 PUBLIC JA, JB, JC, R, JH, JN, NAB, CO(15), CO(5), COEF(5, 14, 21)
a an <del>increas</del> aire, a	3 TO THE OWNER OF THE CODP(1), CNDP(1),
	4 PUBLIC SS(200), SB(200), SSDP(1), SBDP(1)
	5 PUBLIC JF(3), JFDP(1), RM, RX, FA(2,30), GA(2,30), FADP(2), GADP(2)
1	6 PUBLIC MA(2), MADP(1), IX(3,3), IXDP(2)
دربوه احمده مادستورینی	PUBLIC RN, NMA(3), NMB(3), NMDP(1), NBDP(1)
	NETERA
	Y TATAN INTERER KUTAN AND AND AND AND AND AND AND AND AND A
i transforma 🚺	O INIEVER RIJKAJFAJGA
1	INTEGER & C. C. M. Strategy and Strategy a
- Emilia (* 11	'FROCEDURE' SOLUTION (A, EV, SC, NS, LY.);
11	'ARRAY' A, EV; 'REAL' SC; 'INTEGER' 18, KY;
	ter and the Bocin's statement of the sta
	INTEGER' N,I,J,K;
e prag Ser con 🚦	'ARRAY' L(1'.'RN, 1'.'RN);
e en	$\mathbf{N} = \mathbf{N} = \mathbf{R} \mathbf{N}$
- E	
19	
20	D FOR 1=1 SILP 1 UNILL N. DU JS(1) = SGRT(ABS(1/A(1,1)))
2	FOR 1=1 STEP 1 UNTIL'N' DO' FOR J=1 STEP 1 UNTIL'N' DO'
21	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
2:	3 L(1+1)=SQRT(A(1+1))) FOR+1=2*STEP+1*UNTIL*N*DO+L(1+1)=A(1+1)+1+1+1+1
	FOR 1=2'STEP 1'UNTIL'N'DO'FFOR J=1'STEP 1'UNTIL'N'DO'FFOR J=1'STEP 1'UNTIL'N'DO'FFOR N'
	A TEA TANES OF THEN'S GOTOPOUT STEPRINE IS STEPPINE UNTING THE PORT
	1 (1 + 1) = 1 (1 + 1) + 1 (1 + K) + 1 (1 + K) = 1 (1 + 1) = 50 PT (1 + 1) = 1 (1 + 1) + 1
20	
2 (	
	3 001 1 1 0 K K=1 3 C F 1 0 K IL 1 - 1 - 0 - L ( J ) I ) = L ( J ) I ) + L ( J ) K ) \$
29	
30	D EXIT / · / END · .
31	INVLOV(A,L,N); INVLOV(A,L,N); IN INTERNAL AND INTERNAL AND
32	RAFD(L,2,KY+3);
	FOR 1=1+STEP+1*UNTIL+N*DO**FOR j=1*STEP+1*UNTIL+N*DO*
	L(I)J=L(I)J+JS(I)+JS(J); C) C C C C C C C C C C C C C C C C C
	TRAML(L/A/N);WATD(A/2/2);MATML(L,A/N);
r <u>atu</u> r sturt <b>9</b> 2	FOR/1=1+STEP/1/UNTIL/N=1/DO/FOOP/J=I+1/ CTFP/1/UNTIL/NARCA
30	A(J, T)=A(T, J)=0.5+(A(T, J)+A(J, T))I
<u> </u>	*END:
-FD-1 39	PUNITUS
40	ENDINE BEOINT INTEOLET INDEAN SHI ARRATEE, ENTINES
(週刊) 41	n na stra <b>k = 1 /</b>
42	HOUSEH(AJEE)NJKJ
. 43	1 IF'K'EQ'9'''THEN'TEXT(/(/''('N')'JACOBI/)))
44	<pre>*FOR*K=1*STEP*1*UNTIL*N*DO*EE(K)=1/EE(K);</pre>
	LINES(5); EVECPR(EE, N, 4); LINES(5);
	+FOR/I=1+STEP+1/UNTIL+N'DO'
	<pre>&gt; 1F'EE(1)'GT'O'THEN'EV(1)=SC*SQRT(EE(1));</pre>
iaira∺ s. *•f	A REGINAL REGINAL AND A REGINAL AND
	DCAL / SIM : / ARRAY/B(1/ JN. 1/ JN. )
50	RAF V(D) LT LA FTCD + 1 JUNTIL +NANDAAD CALLA FOODA + AAA THAT HAND CALLA FOODA + AAA
51	FURIAISSEPTIONTLENS DEGINSFURJEISSEPTIONTILSNADDA
52	EE(J) #A(I,J);
	+FOR+K=1+STEP+1*UNTIL+N+DO++BEGIN+
54	SUM=0;*FOR*H=K*STEP*1*UNT1L*N*D0*SUM=SUM+B(H,K)*EE(H)1
55	FUCK)=SUMI -
E A CONTRACTOR OF ST	The second s
	JEORAJ-14 STEPATAUNTILANADOPACT LIVEENALIVEENAL
······································	The second s
ax.ect : 20	
<u>2</u> 59	CARD TO WE IT IT'S TEP I' UNITERD'DO'FFORJEI'STEP'I'UNTIL'RN'DO
60	
60 61	
60 61 62	$A(I \downarrow J) \neq JS(J) \neq A(I \downarrow J)$ $PND'$ $PND'$

### HQR SOLUTION

- <u>The States</u>	1 PUBLIC AL, BL, AI, BI, ETA, ETA2, HT, HT2, ACC, KS, H, N, NA, NB, NC, NN
	2 PUBLIC JA, JB, JC, R, JM, JN, NAB, CO(15), CO(5), COEF(5, 14, 21)
American A	3 PUBLIC CODP(1), CNDP(1), AA
	4 PUBLIC SS(200), SB(200), SSDP(1), SBDP(1)
	5 PUBLIC JF(3), JFDP(1), RM, RX, FA(2, 30), GA(2, 30), FADP(2), GADP(2)
an a	6 PUBLIC MA(2), MADP(1), IX(3,3), IXDP(2)
fall a ser e	7 PUBLIC RN, NMA(3), NMB(3), NMDP(1), NBDP(1)
	Branchart REAL KS
	9 INTEGER RN
3 M 1	O INTEGER RN,RX,FA,GA
1	1 INTEGER R
1	2 PROCEDURE' SOLUTION (A, EV, SC, INS, KY);
1	3 ARRAI A, SV; 'REAL' SC; 'INTEGER' NS, FI;
1	Bissin
1	5 $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1)$ $(1$
· I	6 ARRAT SA(1., 200), D(1., AR, 1., AR), ER, EI(1., AR);
inter destant 1	7 PROCEDURE VECCUT (SATSSTATS ARRAY SATSST INTEGER X)
- 1	$\frac{8}{3} = \frac{1}{3} = \frac{1}$
<b>I</b>	
	$0 \qquad \text{In } = \text{ In } (300 \text{ A mean } (30000 \text{ (SA SS M)}) \text{ (FSS) } \text{ (SA SS M)}).$
Z	hebi
	2 PAED (A+2, 2+2+HY)1RAED/B.2/3+2+HY11
	MATDIV(B.A.RN, RN)
<b>-</b>	NSEIGB(B,A,2,2,R) ER, E1)3
	EVECPR(ER, RN, 4) JEVECPR(EI, RN, 4) J
	EMATPR (BARNARNA 4) JEMATPR (AARNARNA 4) 1
	HATCOP(A/B,RN/RN)}
<u> </u>	· FOR'I=I'STEP'I'UNTIL'RN'DO'
3	1 / IF/ER(I)/GT/0/THEN/
3	Ev(1)=SC/SQRT(ER(1));
- 72 - 7 - 5 <b>3</b>	

# TRANSE No Reduction

	PROCE	וטעב	RE!	TRANSF	(A	L,D,RD,NI	i,Ki	);	
	A RAT	(' )	A,D;	'INTEX	<u>i i i</u>	RD,NN	,ŁY j	f	
£1. i	BEGI	11	INT.	EGER :	E,J	;			
	'FOR'	I	- 1	'STEP'	1	'UNTIL'	RD	'DO'	
	"FOR"	J	= 1	'STEP'	1	'UNTIL'	$\mathbf{N}\mathbf{N}$	1D01	
	D(I,J)	) =	A()	[,J);					
	*E.D*		•						

<pre>PUBLIC AL, BL, AI, BI, ETA, ETA, FTA2, AT, HT2, ACC, KS, M, NA, NB, NC, NN PUBLIC JA, JB, JC, R, JK, JK, MAB, CO(15), CO(F(5,14,21) PUBLIC SS(200), SS(200), SSDP(1), SBDP(1) PUBLIC SS(200), SSDP(1), SR, KK, FA(2,30), GA(2,30), FADP(2), GADP(2) PUBLIC MA(2), MADP(1), IX(3,3), IXOP(2) PUBLIC MA(2), MADP(1), IX(3,3), IXOP(2) PUBLIC MA(2), MADP(1), IX(3,3), IXOP(2) PUBLIC MA(2), MADP(1), IX(3,3), IXOP(2) PUBLIC MA(2), MADP(1), NB(3), NMDP(1), NBDP(1) REAL KS INTEGER RN PTROCEUNCH' THATASY (A, D, HD, NN, KY); ARRAY' BJDTM' ITTEDER' HD, IX, KY; BEDTM' ITTEDER' HA, IX, FA, GA INTEGER RN (I', 'RR, I', 'RR, I', 'RRAL'STMA; ARRAY' BJTM' ITTEDER' HA, JK, KY; BEDTM' ITTEDER' HA, IX, FY; BEDTM' ITTEDER' HA, IX, FY; 'REAL'STMA; ARRAY' B(1', 'RR, I', 'RN); FOR' 1='STEP' I', UNTIL', RA'DO'E(I, J) = D(I, RN+J); FOR' 1='STEP'I', UNTIL', RA'DO'E(I, J) = F(I, J) = G(I); FOR' 1='STEP'I', UNTIL', RA'DO'D(I, J) = F(I, J) = G(I); FOR' J=C', I'STEP'I', UNTIL', RA'DO'D(I, J) = F(I, J) = G(I); FOR' J=C', I'STEP'I', UNTIL', RA'DO'D(I, J) = F(I, J) = G(I); FOR' J=C', I'STEP'I', UNTIL', RA'DO'D(I, J) = F(I, J) = G(I); FOR' J=C', I'STEP'I', UNTIL', RA'DO'D(I, J) = F(I, J) = G(I); FOR' J=C', I'STEP'I', UNTIL', RD'DO''SEGIN' MAXAR(2500); RAFO(B, Z, H X); FAC' 1='STEP'I', UNTIL', RD'DO''SEGIN' MAXAR(2500); RAFO(B, Z, H X); FOR' I = STEP'I', UNTIL', RD'DO''SEGIN' SUM=0; FOR J=; STEP'I', UNTIL', RD'DO''SEGIN' FERD'; FOR'J=I'SSEP'I', UNTIL', RD'DO''SEGIN' FOR'J=I'SSEP'I', UNTIL', RD'DO''SEGIN' FOR'J=I'SSEP'I', UNTIL', RD'DO''SEGIN' FOR'J=I'SSEP'I', UNTIL', RD'DO''SEGIN' FOR'J=I'SSEP'I'SSEP'I', UNTIL', RD'</pre>		TRANSF	With	Reduction			
<pre>PUBLIC JAJBJJCR, MAJA, MA, NAB, CO(15), CO(5), COEF(5, 14, 21) PUBLIC SOFCOP(1), KNAB, CO(15), COEF(5, 14, 21) PUBLIC SS(200), SS(200), SS(201), SSOP(1), SSOP(1) PUBLIC HA(2), MAOP(1), NAB, FA(2, 30), FADP(2), GADP(2) PUBLIC HA(2), MAOP(1), NADP(1), NBOP(1) REAL KS INTEGER RN, RA, FA, GA POCEDURE: SHIFTBK(D); AND, KY; PROCEDURE: SHIFTBK(D); AND, KY; PROCEDURE: SHIFTBK(D); AND, KY; PROCEDURE: SHIFTBK(D); AND, FA, FA, GA POCEDURE: SHIFTBK(D); AND, FA, FA, GA POCEDURE: SHIFTBK(D); AND, FA, FA, GA POCEDURE: SHIFTBK(D); AND, FA, FA, GA, FA, FA, FA, GA, FA, FA, GA, FA, FA, GA, FA, FA, FA, GA, FA, FA, FA, GA, FA, FA, FA, GA, FA, FA, FA, FA, FA, FA, FA, FA, FA, F</pre>	्र हत्ताय∙श्हित्त ∎य्यतः	PUBLIC AL,B	L, A1, B1, E	TA, ETAZ, HT, HTZ	ACC, KS, M. N, NA, NE	NC. NN	
<pre>3 PUBLIC CODP(1),CNDp(1) 9 PUBLIC S1(200),SSD(2(1),SBDP(1),SBDP(1) 9 PUBLIC JF(3),JFPP(1),RH,RX,FA(2,30),GA(2,30),FADP(2),GADP(2) 9 PUBLIC RN,NMA(3),NMB(3),NMPP(1),NBDP(1) 8 REAL KS 9 INTEGER RN 10 INTEGER RN 10 INTEGER RN 11 INTEGER RN 12 'PROCEDURL' THANSP (A,D,HE,MN,KY); 13 'ARDAY' A,D, 'ITTEPER' HD,NX,KY; 13 'ARDAY' A,D, 'ITTEPER' HD,NX,KY; 14 'BDDIN' 'ITTEPER' HD,NX,KY; 15 'PROCEDURL' THANSP (A,D,HE,MN,KY); 16 'PROCEDURL' THANSP (A,D,HE,MN,KY); 17 'ARDAY' A,D, 'ITTEPER' HD,NX,KY; 18 'PROCEDURL' THANSP (A,D,HE,MN,KY); 19 'PROCEDURL' SHIFTAK(D): AARRAY*D; 19 'PROR'JE'STEP'!'UNTIL'RN'DO'ABEGIN* 20 'PGR'JE'STEP'!'UNTIL'RN'DO'F(I,J)=D(I,RN+J); 21 'PGR'JE'STEP'!'UNTIL'RN'DO'F(I,J)=D(I,J); 22 CF=CG=CH=L]=CK=D; 23 'FGR'KA'STEP'!'UNTIL'RN'DO'F(I,J)=D(I,J)=F(I,J)=CG); 24 'FGR'JI'STEP'!'UNTIL'RN'DO'F(I,J)=F(I,J)=CG); 25 'FGR'JI'STEP'!'UNTIL'RN'DO'P(J,J)=F(I,J)=CG); 26 'FGR'JI'STEP'!'UNTIL'RD'DO'*FGR'J=!*STEP'!*UNTIL*RN'DO'D(I,J)=A(I,J)] 27 'FGR'I'STEP'!'UNTIL'RD'DO'*FGR'J=!*STEP'!*UNTIL*RN'DO'D(I,J)=A(I,J)] 28 'FGR'JI'STEP'!'UNTIL'RD'DO'*FGR'J=!*STEP'!*UNTIL*RN'DO'D(I,J)=A(I,J)] 29 'FGR'JI'STEP'!'UNTIL'RD'DO'*FGR'J=!*STEP'!*UNTIL*RN'DO'D(I,J)=A(I,J)] 30 MAXAR(2500); 31 AAFD(B,Z+H X]; 33 'FGR'I'STEP'!'UNTIL'RD'DO'*FGR'J=!*STEP'!*UNTIL*RN'DO'D(I,J)=A(I,J)] 34 'FGR'I'STEP'!'UNTIL'RD'DO'*FGR'J=!*STEP'!*UNTIL*RN'DO'D(I,J)=A(I,J)] 35 DIKAJ!*STEP'!'UNTIL*RD'DO'*FGR'J=!*STEP'!*UNTIL*RN'DO'D(I,J)=A(I,J)] 36 DIKAJ!*SUH; 37 'FGR'J=STEP'!'UNTIL*RD'DO'*FGR'J=!*STEP'!*UNTIL*RN'DO' 36 DIKAJ!*STEP'!*UNTIL*RD'DO'*FGR']=!*STEP'!*UNTIL*NN*DO' 36 DIKAJ!*STEP'!*UNTIL*RD'DO'*FGR']=!*STEP'!*UNTIL*NN*DO' 37 'FGR'J=!*STEP'!*UNTIL*RD'DO'*FGR']=!*STEP'!*UNTIL*NN*DO' 36 DIKAJ!*STEP'!*UNTIL*RD'DO'*FGR']=!*STEP'!*UNTIL*NN*DO' 37 'FGR'J=!*STEP'!*UNTIL*RD'DO'*FGR']=!*STEP'!*UNTIL*NN*DO' 36 DIKAJ!*STEP'!*UNTIL*RD'DO'*FGR']=!*STEP'!*UNTIL*NN*DO' 37 'FGR'J=!*STEP'!*UNTIL*RD'DO'*FGR']=!*STEP'!*UNTIL*NN</pre>	2	PUBLIC JA,J	B, JC, R, JN	1, JN, NAB, CO(15),	CN(5), COEF(5,14,	21)	
<pre>     PubLic SS(200), SB(200), SSDP(1), SDP(1), SDP(1)     PubLic MA(2), MADP(1), IX(3,3), IXDP(2)     PubLic MA(2), MADP(1), IX(3,3), IXDP(2)     PubLic RA(RX, FAAGA     INTEGER RN, MA(3), NMBC), NMDP(1), NBDP(1)     REAL KS     INTEGER RN, TANISP (A, D, MD, NN, KY);     INTEGER RN, TANISP (A, D, MD, NN, KY);     IATATA' A, D, 'INTEGER' HD, NN, KY);     'AREAY' A, D, 'INTEGER' HD, NN, KY;     'AREAY' C, IA NR, I', 'NN, ', 'REAL'SUM;     'AREAY' C, IA NR, I', 'NN, ', 'REAL'SUM;     'AREAY' C, IA 'RM, I', 'NN, ', 'REAL'SUM;     'AREAY' C, IA 'RM, I', 'RN, 'I', 'REAL'SUM;     'AREAY' C, IA 'RM, I', 'RN, 'I', 'REAL'SUM;     'AFGR'JE', 'STEP' I'UNTIL' RN'DO' C, (L, J) 'A', 'RN, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' F (I, J) 'RO (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' F (I, J) 'RO (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' F (I, J) 'RO (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' F (I, J) 'RO (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' F (I, J) 'RO (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' F (I, J) 'RO (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' F (I, J) 'RO (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' F (I, J) 'RO (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' P (I, J) 'STEP' I'UNTIL'RN'DO' D (I, J) 'A' (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' P (I, J) 'STEP' I'UNTIL'RN'DO' D (I, J); 'A' (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' P (I, J) 'STEP' I'UNTIL'RN'DO' D (I, J); 'A' (I, J);     'FOR'JE', 'STEP' I'UNTIL'RN'DO' SUM*SUM+B (I -RN, J) *A' (K, J); 'END';     'FOR'JE', 'STEP' I'UNTIL'RN'DO' SUM*SUM+B (I -RN, J) *A' (K, J); 'END';     D (K, I) 'SUM;     'STEP' I'UNTIL'RD'DO' F (FOR'] *I'STEP' I'UNTIL'RN'DO'     D (K, I) 'SUM;     'ACR'I'STEP' I'UNTIL'RD'DO' F (FOR'] *I'STEP' I'UNTIL'RN'DO'     D (J, I) *A' (I);     'ATD';     '</pre>	3	PUBLIC CODP	(1) / CNDP(	12			
<pre>5</pre>		PUBLIC SS(2)	00),88(20	)0),SSDp(1),SBDP	(1)	a na na na a gagan ang én ni nganarang mgan ninganan A	
<pre>6</pre>	5	PUBLIC JF(3	),JFDP(1)	,RM,RX,FA(2,30)	,GA(2,30),FADP(2	(),GADP(2)	
<pre>7 PUBLIC RN, NHA(3), NHB(3), NHDP(1), NBDP(1) 8 REAL KS 9 INTEGER RN 10 INTEGER RN 11 INTEGER RN 12 'PROCEDURL' THAILSP( A, D, RD, NN, KY); 13 'BADIN' 'INTEGER' RN, TL, J, E; 'RBAL'SUM; 14 'BDDIN' 'INTEGER' RN, TL, J, E; 'RBAL'SUM; 15 'ARRAY'E(1, 'RR, 'I', 'RR); 16 'PROCEDURE'S NIFTBK(D); ARRAY'D; 17 'ABEGIN' INTEGER' I, J, K, H, CF, CG, CH, CJ, CK; 17 'ARRAY'E(1, 'RR, 'I', 'RR); 18 'PROCEDURE'S NIFTBK(D); ARRAY'D; 19 'FOR'J='STEP'!'UNTIL'RN'DO'E(I, J)=D(I, RN+J); 20 'FOR'J='STEP'!'UNTIL'RN'DO'E(I, J)=D(I, RN+J); 21 'FOR'J='STEP'!'UNTIL'RN'DO'E(I, J)=D(I, RN+J); 22 'FOR'J='STEP'!'UNTIL'RN'DO'E(I, J)=D(I, RN+J); 23 'FOR'J='STEP'!'UNTIL'RN'DO'E(I, J)=D(I, RN+J); 24 'FOR'J='STEP'!'UNTIL'RN'DO'E(I, J)=D(I, NL'); 25 'FOR'J=I'STEP'!'UNTIL'RN'DO'E(I, J)=CI, J-CG); 26 'FOR'J=C:I'STEP'!'UNTIL'RN'DO'D(I, J)=F(I, J-CG); 27 'FOR'J=C:I'STEP'!'UNTIL'RD'DO''FOR'J=I'STEP'!'UNTIL'RN'DO'D(I, J)=A(I, J); 28 'FOR'J=C:I'STEP'!'UNTIL'RD'DO''FOR'J=I'STEP'!'UNTIL'RN'DO'D(I, J)=A(I, J); 29 'FONJ; 20 'FOR'J=C:I'STEP'!'UNTIL'RD'DO''FOR'J=I'STEP'!'UNTIL'RN'DO'D(I, J)=A(I, J); 21 'FOR'J=C:I'STEP'!'UNTIL'RD'DO''FOR'J=I'STEP'!'UNTIL'RN'DO'D(I, J)=A(I, J); 23 'FOR'I=I'STEP'!'UNTIL'RD'DO''FOR'J=I'STEP'!'UNTIL'RN'DO'D(I, J)=A(I, J); 34 'FOR'I=I'STEP'!'UNTIL'RD'DO''FOR'J=I'STEP'!'UNTIL'RN'DO'D(I, J)=A(I, J); 35 SUM O; 'FORJ=I'STEP'!'UNTIL'RN'DO'SUM=SUM=SUM+B(I=RN, J)=A(K, J);'END'; 36 'FON'] 37 'END'; 36 'FON'] 37 'FON']='STEP'!'UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO' 36 'FON']='STEP'!'UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO' 37 'END'; 36 'FON']='STEP'!'UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO' 36 'FON']='STEP'!'UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO' 37 'FON']='STEP'!'UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO' 36 'FON']='STEP'!'UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO' 37 'END'; 36 'STURD;'STEP'!'UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO' 37 'END'; 38 'STURD;'STEP'!'UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO' 39 'FON']='STEP'!'UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO' 36 'FON']='STEP']='UNTIL'RD'DO''FOR']=I'STEP'!'UNTIL'NN'DO'' 37 'FON']='STEP'!'U</pre>		PUBLIC HA(2	) MADP(1)	),IX(3,3),IXDP(2	•	•	
<pre>6</pre>	7 1	PUBLIC RNIN	MA(3)/NME	3(3),NMDP(1),NBC	P(1)		•
<pre>9 INTEGER RN 10 INTEGER RN 11 INTEGER RN 12 'PROCEDURA' TRAINSP ( A, D, ND, NN, KY); 13 'ARRAY' A, D, 'INTEGER' ND, NN, KY; 14 'BEDIN' 'INTEGER' ND, NN, KY; 15 'ARRAY' B, D, 'INTEGER' ND, NN, KY; 16 'PROCEDURE' SHIFTBK(D); ARRAY'D; 17 'ARRAY' B, NTEGER' I, J, K, H, CF, CG, CH, CJ, CK; 18 'ARRAY'E (1', 'RN, I', 'RN); 19 'FOR' ISTEP'; UNTIL'RN'DO'F, C, I, J, K', 'RN, J); 19 'FOR' JSTEP'; UNTIL'RN'DO'F (I, J) SD(I, RN+J); 20 'FOR' JSTEP'; UNTIL'RN'DO'F (I, J) SD(I, RN+J); 21 'FOR' JSTEP'; UNTIL'RN'DO'F (I, J) SD(I, RN+J); 22 'FOR' JSTEP'; UNTIL'RN'DO'F (I, J) SD(I, RN+J); 23 'FOR' STEP'; UNTIL'RN'DO'F (I, J) SEGIN' 24 CHSCF+IAKINS; STEP'; UNTIL'RN'DO'F (I, J) SEGIN' 25 'FOR' JSTEP'; UNTIL'RN'DO'F (I, J) SEGIN' 26 'FOR' JSTEP'; UNTIL'RN'DO'F (I, J) SEGIN' 27 CFSCK:CGCJ; 28 'FOR' JSCF+I'STEP'; UNTIL'RD'DO'FOR*J=I'STEP'; UNTIL'RN'DO'D(I, J):A(I, J); 29 'HAXARR(CSOO); 29 'FOR'; STEP'; UNTIL'RD'DO'FOR*J=I'STEP'; UNTIL'RN'DO'D(I, J):A(I, J); 30 'MAXARR(CSOO); 31 'FOR' ISTEP'; UNTIL'RD'DO'FBCGIN' 35 OFOR'; STEP'; UNTIL'RD'DO'FBCGIN' 36 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 37 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 38 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 39 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 30 'MAXARR(CSOO); 31 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 32 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 33 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 34 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 35 OFOR'; STEP'; UNTIL'RD'DO'FBCGIN' 36 OFOR'; STEP'; UNTIL'RD'DO'FBCGIN' 37 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 36 OFOR'; STEP'; UNTIL'RD'DO'FBCGIN' 37 'FOR'; STEP'; UNTIL'RD'DO'FBCGIN' 38 'FOR'; STEP'; STEP'; UNTIL'RD'DO'FBCGIN' 39 'FOR'; STEP'; STEP'; UNTIL'RD'DO'FBCGIN' 30 'FOR'; STEP'; STEP'; UNTIL'RD'DO'FBCGIN' 31 'FOR'; STEP'; STEP'; UNTIL'RD'DO'FBCGIN' 32 'FOR'; STEP'; ST</pre>	8	REAL KS		• • •		•	
<pre>10 11 14 14 14 14 14 15 15 16 17 17 18 18 18 19 19 10 10 10 10 10 10 10 10 10 10</pre>	9	INTEGER RN	•				
<pre>11 INTEGER R 12 'PROCEDUR.' THANSP ( A, D, HD, NN, KY); 13 'ARRAY' A, D; 'INTEGER' HD, NN, KY; 14 'BOTIN' 'INTEGER' HD, NN, KY; 15 'ARRAY'E (1', 'RN, 1', 'RN); 16 'PROCEDURE'S HIFTBK (D); ARRAY'D; 17 'BEGIN''INTEGER'IJJK, 'RCA'CG'CH, CJ, CK; 18 'ARRAY'E (1; A'RD')' 'KRN, F(1', *RN); 19 'FOR']=1'STEP'1'UNTIL'RD'DO''BEGIN' 19 'FOR']=1'STEP'1'UNTIL'RN'DO'E (I, J)=D(I, RN+J); 20 'FOR'J=1'STEP'1'UNTIL'RN'DO'F (I, J)=D(I, RN+J); 21 CF=CG=CH=CJ=CK=CJ; 23 'FOR'K=1'STEP'1'UNTIL'RN'DO'F(I)J)=F(I)J=C(I); 24 'FOR'J=CK=CJ; 25 'FOR'J=CH=CJ=CK=CJ; 26 'FOR'J=CG=CJ; 27 'FOR'J=CH=CJ=CK=CJ; 26 'FOR'J=CH=CJ=CK=CJ; 27 'FOR'J=CH=CJ=CK=CJ; 28 'FOR'J=CH=CJ=CK=CJ; 29 'FOR'J=CH=CJ=CK=CJ; 20 'FOR'J=CH=CJ=CK=CJ; 21 'FOR'J=CH=CJ=CK=CJ; 23 'FOR'J=CH=CJ=CK=CJ; 24 'FOR'J=CH=CJ=CK=CJ; 25 'FOR'J=CH=CJ=CK=CJ; 26 'FOR'J=CH=CJ=CK=CJ; 27 'FOR'J=CH=CJ=CK=CJ; 28 'FOR'J=CH=CJ=CK=CJ; 29 'FOR'J=CH=CJ=CK=CJ; 20 'FOR'J=CH=CJ=CK=CJ; 21 'FOR'J=CH=CJ=CK=CJ; 22 'FOR'J=CH=CJ=CK=CJ; 23 'FOR'J=CH=CJ=CK=CJ; 24 'FOR'J=CH=CJ=CK=CJ; 25 'FOR'J=CH=CJ=CK=CJ; 26 'FOR'J=CH=CJ=CK=CJ; 27 'FOR'J=CH=CJ=CK=CJ; 28 'FOR'J=CH=CJ=CK=CJ; 29 'FOR'J=CH=CJ=CK=CJ; 20 'FOR'J=CH=CJ=CK=CJ; 21 'FOR'J=CH=CJ=CK=CJ; 22 'FOR'J=CH=CJ=CK=CJ; 23 'FOR'J=CH=CJ=CK=CJ; 24 'FOR'J=CH=CJ=CK=CJ; 25 'FOR'J=CH=CJ=CK=CJ; 26 'FOR'J=CH=CJ=CK=CJ; 27 'FOR'J=CH=CJ=CK=CJ; 28 'FOR'J=CH=CJ=CK=CJ; 29 'FOR'J=CH=CJ=CK=CJ; 20 'FOR'J=CH=CJ=CK=CJ; 21 'FOR'J=CH=CJ=CH=CG]; 22 'FOR'J=CH=CJ=CH=CG]; 23 'FOR'J=CH=CJ=CH=CH=CH=CH=CH=CH=CH=CH=CH=CH=CH=CH=CH=</pre>	10	INTEGER RM,	RX;FA;GA		· · · · · · · · · · · · · · · · · · ·		
<pre>PROCEDURL' THAILSP ( A, D, HU, NN, KY); 'ARRAY' A, D; 'IITEGER' HX, T, J, E; 'REAL'SUN; 'ARRAY' B(1', 'RK, 1', 'RN); 'ARRAY' B(1', 'RK, 1', 'RN); 'FOR' != 1'STEP' 1'UNT1L' RN'DO' BEGIN' 'FOR' J= 1'STEP' 1'UNT1L' RN'DO' F(1, J) *D('I, N+J); 'FOR' J= 1'STEP' 1'UNT1L' RN'DO' F(1, J) *D('I, J); 'FOR' J= 1'STEP' 1'UNT1L' RN'DO' F(1, J) *D('I, J); 'FOR' J= 1'STEP' 1'UNT1L' CK'DO' D('I, J) *E(1, J-CG); 'FOR' J= 1'STEP' 1'UNT1L' CK'DO' D(I, J) *E(1, J-CH+CG); 'FOR' J= 1'STEP' 1'UNT1L' RD' DO' FOR*J=1'STEP' 1'UNT1L' RN'DO' D(I, J) *A((I, J)); 'FOR' I * STEP' 1'UNT1L' RD' DO' * BEGIN' 'FOR' I * STEP' 1'UNT1L' RN'DO' SUM*SUM*B(I=RN, J) *A(K, J) * END' ; D(K, I) * SUM; 'FOR' J = 1'STEP' 1'UNT1L' RD' DO' * FOR' I = 1'STEP' 1'UNT1L' NN'DO' D(K, I) = SUM; 'FOR' J = 1'STEP' 1'UNT1L' RD' DO' * FOR' I = 1'STEP' 1'UNT1L' NN'DO' D(K, I) = SUM; 'FOR' J = 1'STEP' 1'UNT1L' RD' DO' * FOR' I = 1'STEP' 1'UNT1L' NN'DO' D(J, I) = D(J, I) = SA(I) ; 'SID'' 'SID''</pre>		INTEGER R					
<pre>*ARRAY A,D; 'IITEDER' HD,RX,FY; 'BDCIN' INTEDER' HX,I,J,E; 'RBAL'SUN; 'ARRAY' B(1'*RX,1'*IN); 'PROCEDURE'SHIFTBK(D);*ARRAY'D; 'PROCEDURE'SHIFTBK(D);*ARRAY'D; 'PROCEDURE'SHIFTBK(D);*ARRAY'D; 'PROCEDURE'SHIFTBK(D);*ARRAY'D; 'PROCEDURE'SHIFTBK(D);*ARRAY'D; 'PROCEDURE'SHIFTBK(D);*ARRAY'D; 'PROCEDURE'SHIFTBK(D);*ARRAY'D; 'PROCEDURE'SHIFTBK(D);*ARRAY'D; 'PROCEDURE'SHIFTBK(D);*ARRAY'D; 'PROCEDURE'STEP';'UNTIL'RD'DO'SEG[N* 'PROCEDURE'STEP';'UNTIL'RM'DO'F(I;J);*D(I;J); 'PROCECHEUJECK:D; 'PROCECHEUJECK:D; 'PROCEFA(HX*I;X);'DI'CH'DO'BEG[N* 'PROCEFA(HX*I;X);'DI'CH'DO'D(I;J);*E(I;J-CG); 'PROCECH:'STEP';'UNTIL'RD'DO'*FOR'J=1'STEP';'UNTIL'RN'DO'D(I;J);*A(I;J);' 'PROCECHE';'STEP';'UNTIL'RD'DO'*FOR'J=1'STEP';'UNTIL'RN'DO'D(I;J);*A(I;J);' 'PROCECHE';'PROCECHE';'UNTIL'RD'DO'*FOR'J=1'STEP';'UNTIL'RN'DO'D(I;J);*A(I;J);' 'PROCECHE';'PROCECHE';'UNTIL'RD'DO'*FOR'J=1'STEP';'UNTIL'RN'DO'D(I;J);*A(I;J);' 'PROCECHE';'PROCECHE';'UNTIL'RD'DO'*FOR'J=1'STEP';'UNTIL'RN'DO'D(I;J);*A(I;J);' 'PROCECHE';'PROCECHE';'UNTIL'RD'DO'*FOR'J=1'STEP';'UNTIL'RN'DO'D(I;J);*A(I;J);' 'PROCECHE';'PROCECHE';'UNTIL'RD'DO'*FOR'J=1'STEP';'UNTIL'RN'DO'D(I;J);*A(I;J);' 'PROCECHE';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROCECHE';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROCECHE';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROCECHE';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROCECHE';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROC';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROC';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROC';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROC';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROC';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO'';' 'PROC';'PI'UNTIL'RD'DO'*FOR'I=1'STEP';'UNTIL'NN'DO''';' 'PROC';'PI'UNTIL'RD';'',''';''''';''''''''''''''''''''''</pre>	12	"PROCEDUR_D" TRANSF	( A,D,RD,	NN, KY);			
<pre>BECIN: 'INFLGGR' HX, T, J, É, 'REAL'SUN; 'ARGAY' B(1', 'RN, 1', 'RN); PROCEDURE, SHIFTBK(D); ARRAY'D; /BEGIN' INTEGER' I, J, K, H, CF, CG, CH, CJ, CK; /ARRAY'E(1, *RO, 1', 'RM), F(1', *RD, 1', 'RN); /ARRAY'E(1, *RO, 1', 'RM), F(1', *RD, 1', 'RN); /ARRAY'E(1, *RO, 1', 'RM), F(1', 'RD, 1', 'RN); /ARRAY'E(1', 'RC), 1', 'RN, 1', 'RN, 'D', 'RD, 'RN); /FOR'J=1'STEP'1'UNTIL'RD'DO'BEGIN, /FOR'J=1'STEP'1'UNTIL'RN'DO'F(I, J)=D(1, RN+J); /FOR'K=1'STEP'1'UNTIL'RN'DO'F(I, J)=D(1, KN;); /FOR'K=1'STEP'1'UNTIL'RN'DO'F(I, J)=D(1, KN;); /FOR'J=CF+CF+CK=CG; /FOR'J=CF+CF+CF+1'UNTIL'RN'DO'F(I, J)=C(1); /FOR'J=CF+CF+CF+1'UNTIL'CK'DO'D(1, J)=E(1, J-CG); /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CG;; /FOR'J=CF+CF+CF+CF+CG;; /FOR'J=CF+CF+CF+CF+CG;; /FOR'J=CF+CF+CF+CF+CG;; /FOR'J=CF+CF+CF+CF+CF+CF+CF+CG;; /FOR'J=CF+CF+CF+CF+CF+CF+CF+CF+CF+CG;; /FOR'J=CF+CF+CF+CF+CF+CF+CF+CF+CF+CF+CG;; /FOR'J=CF+CF+CF+CF+CF+CF+CF+CF+CF+CF+CG;; /FOR'J=CF+CF+CF+CF+CF+CF+CF+CF+CF+CF+CF+CG;; /FOR'J=CF+CF+CF+CF+CF+CF+CF+CF+CF+CF+CF+CF+CF+C</pre>	12	"ARRAY" A,D; "INTEX	GER' RD, NN.	, KY ;			
<pre>ADMAY' B(1*,'RN,1*,'RN); /*PROCEDURE,SHIFTBK(D);*ARRAY*D; /*PROCEDURE,SHIFTBK(D);*ARRAY*D; /*PR*I*I*STEP*I*UNTIL*RD*D0**EG[N* /*PR*I*I*STEP*I*UNTIL*RD*D0**EG[N* /*PR*J=I*STEP*I*UNTIL*RD*D0**EG[N* /*PR*J=I*STEP*I*UNTIL*RD*D0**EG[N* /*PR*J=I*STEP*I*UNTIL*RD*D0**EG[N* /*PR*J=CF=KA(HX*1;K) *CJ*CG*G A(HX*1;K)*CK*CH*GA(HX*1;K); /*FOR*J=CF=KA(HX*1;K) *CJ*CG*G A(HX*1;K)*CK*CH*GA(HX*1;K); /*FOR*J=CC*:I*STEP*I*UNTIL*RD*D0*D(I*J)*F(I*J=CG); /*FOR*J=CC::I*STEP*I*UNTIL*CK*D0*D(I*J)*F(I*J=CG); /*FOR*J=CC::I*STEP*I*UNTIL*CK*D0*D(I*J)*E(I*J=CG); /*FOR*J=CC::I*STEP*I*UNTIL*RD*D0**FOR*J=1*STEP*I*UNTIL*RN*D0*D(I*J)*A(I*J); /*FOR*J=CO: /*FOR*J=CO: /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=D; /*FOR*J=N=N; /*FOR*J=N=N; /*FOR*J=N=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*FOR*J=N; /*F</pre>		BEGIN 'INTEGER' I	HX,I,J,E;	REAL'SUM;			
<pre></pre>	15	**************************************	'.'RN);	• •	• • •	•	
<pre></pre>	· · · · · · · · · · · · · · · · · · ·	PROCEDURE SHIF	TBK ( D ) ; # A	RRAYD		a di atra t	
<pre>ARRAY*E(1,**RD,1**RM),F(1**RD,1**RN); ARRAY*E(1,**RD,1**RN); F(1**RD*1=1*STEP*1*UNTIL*R0*D0**BEGIN* FOR*J=1*STEP*1*UNTIL*R0*D0*F(1,J)=D(1,RN+J); FOR*J=1*STEP*1*UNTIL*R0*D0*F(1,J)=D(1,RN+J); Cf=CG=CH=LJ=CK=0; FOR*J=CF+FA(HX+1;K);CJ=CG+GA(HX+1;K);CK=CH+GA(HX+1;K); FOR*J=CF+FA(HX+1;K);CJ=CG+GA(HX+1;K);CK=CH+GA(HX+1;K); FOR*J=CF+I*STEP*1*UNTIL*CH*D0*D(1;J)=F(1;J=CG); FOR*J=CF+I*STEP*1*UNTIL*CK*D0*D(1;J)=F(1;J=CG); FOR*J=CF+I*STEP*1*UNTIL*RD*D0*FOR*J=1*STEP*1*UNTIL*RN*D0*D(1;J)=A(1;J); FOR*I=T*STEP*1*UNTIL*RD*D0**FOR*J=1*STEP*1*UNTIL*RN*D0*D(1;J)=A(1;J); FOR*I=T*STEP*1*UNTIL*RD*D0**FOR*J=1*STEP*1*UNTIL*RN*D0*J(1;J)=A(1;J); FOR*STEP*1*UNTIL*RD*D0**GR*J=1*STEP*1*UNTIL*RN*D0*J(1;J)=A(1;J); FOR*STEP*1*UNTIL*RD*D0**FOR*I=1*STEP*1*UNTIL*NN*D0* FOR*J=CF=STEP*1*UNTIL*RD*D0**FOR*I=1*STEP*1*UNTIL*NN*D0* O(J,I)=D(J,I)*SA(I]; FOR*J=CJ] FOR*J=CJ] FOR*J=CJ] FOR*J=CJ] FOR*J=CJ] FOR*STEP*1*UNTIL*RD*D0**FOR*I=1*STEP*1*UNTIL*NN*D0* O(J,I)=D(J,I)*SA(I]; FOR*J=CJ] FOR*J=CJ] FOR*J=CJ] FOR*J=CJ] FOR*J=CJ] FOR*J=CJ] FOR*J=CJ] FOR*J=CJ] FOR*STEP*1*UNTIL*RD*D0*FOR*I=1*STEP*1*UNTIL*NN*D0* FOR*J=1*STEP*1*UNTIL*RD*D0*FOR*I=1*STEP*1*UNTIL*NN*D0* FOR*J=1*STEP*1*UNTIL*RD*D0*FOR*I=1*STEP*1*UNTIL*NN*D0*FOR*I=1*STEP*1*UNTIL*NN*D0*FOR*I=1*STEP*1*UNTIL*NN*D0*FOR*I=1*STEP*1*UNTIL*NN*D0*FOR*I=1*STEP*1*UNTIL*NN*D0*FOR*I=1*STEP*1*UNTIL*NN*D0*FOR*I=1</pre>	17	BEGIN INTEGER	* I * J * K * H *	CF, CG, CH, CJ, CK3		•	
<pre></pre>	2	FARRAY'E(1) . PRD.	,1*+*RM),	F(1***RD,1***RN	33		•
<pre></pre>		/FOR/1=1/STEP/1	UNTILIRO	DOP BEGINA		•	
<pre>20 21 23 24 25 25 26 27 26 27 27 26 27 27 27 26 27 27 27 27 27 28 29 29 29 20 20 29 20 20 20 20 20 20 20 21 25 26 27 27 27 27 28 29 29 20 20 27 27 28 29 29 20 20 27 27 28 29 29 20 29 20 20 27 27 28 29 29 20 20 29 20 20 20 20 20 20 20 20 20 20</pre>		"FOR' J=1'STEP'1	UNTIL/RN	*DO'E(I,J)=D(I,	RN+J)1		
<pre>22 23 24 25 25 26 24 24 24 24 25 25 26 26 27 26 27 27 27 28 29 29 29 29 20 29 20 20 20 20 20 21 22 25 27 27 28 29 29 20 29 20 20 20 20 20 20 21 22 23 24 25 25 26 27 27 28 29 29 20 29 20 20 20 20 20 20 20 20 20 20</pre>	20	+FOR J=1+STEP 1	UNTIL'RN	1 DO'F(1, J)=D(1,	J)1		•
<pre>23</pre>		CF=CG=CH=CJ=CK=	03			•	
<pre>24 24 24 CH=CF+FA(HX+1,K) \$CJ=CG+G A(HX+1,K)\$CK=CH+GA(HX+1,K)\$ 25 26 27 CF=CK;CG=CJ* 27 28 29 29 29 29 29 29 29 29 20 29 29 29 29 29 20 29 29 29 29 29 29 29 29 29 29</pre>	22	FOR'K=1'STEP'1	UNTILIMA	(HX+1) / Do / BEG!	N .	•	
<pre>25 25 26 26 27 27 27 28 29 29 29 29 20 30 30 30 30 30 30 30 40 31 31 32 50 32 30 30 30 30 30 30 30 30 40 50 50 50 50 50 50 50 50 50 5</pre>	23	CH=CF+FA(Hx+1)K	) ; CJ=CG+	G A(HX+1,K);CK	CH+GA(HX+1,K)]	. •	
<pre>25 26 FOR'JIC::.1'STEP'I'UNTIL'CK'DO'D(I,J)IE(I,J-CH+CG); 27 28 28 29 29 29 29 20 29 29 29 29 20 29 29 20 29 20 20 20 20 20 20 20 20 20 20</pre>	24	FOR JECE ISTE	PP1PUNTIL	+ CH+ Do+ D(1+J)=F	(1, )-(6)3		
<pre>27 CF = CK; CG = CJ; 28 / END /; *END /; 29 / END /; 30 MAXARR(2500); 31 RAFD(B, 2, H X); 32 / FOR'I = I'STEP / I'UNTIL / RD'DO' / BEGIN/ 33 / FOR'I = RN + I'STEP / I'UNTIL / RN'DO' / BEGIN/ 34 SUM = 0; / FOR / I = I'STEP / I'UNTIL / NN'DO' / BEGIN/ 35 D(K, I) = SUM; 36 / END /; 37 / END /; 38 SHIFTBK(D); 39 SHIFTBK(D); 40 D(J, I) = D(J, I) + SA(I); 42 'END'</pre>	22	FOR JECH-11STE	P+1+UNTIL	+ CK + DO + D ( 1 + J ) = E	(I)J=CH+CG)1		
<pre>27</pre>	20	CFECK:CGECUI				•	
29 /END*3 30 HAXARR(2500)3 31 RAFD(B,2,H X)3 32 * OR*I=1*STEP*1*UNTIL*RD*DO**FOR*J=1*STEP*1*UNTIL*RN*DO*D(I,J)=A(1,J)3 32 * FOR*I=1*STEP*1*UNTIL*RD*DO**BEGIN* 33 * FOR*I=RN+1*STEP*1*UNTIL*NN*DO**BEGIN* 34 SUM=0;*FOR*J=1*STEP*1*UNTIL*RN*DO*SUM=SUM+B(1=RN,J)*A(K,J)3*END*3 35 D(K,I)=SUM; 36 * END*3 37 * END*3 38 SHIFTBK(D); 39 * FOR*J=1*STEP*1*UNTIL* RD*DO**FOR*1=1*STEP*1*UNTIL*NN*DO* 40 D(J,1)=D(J,I)*SA(I); 42 * END*	21	IENDII'ENDII					
30 HAXAR(2500)3 31 RAFD(B,2,H X)3 32 FOR'I=1'STEP'1'UNTIL'RD'DO''FOR'J=1'STEP'1'UNTIL'RN'DO'D(I,J)_A(I,J)3 33 FOR'K=1'STEP'1'UNTIL'RD'DO''BEGIN' 34 SUM=0; 'FOR'J=1'STEP'1'UNTIL'NN'DO' BEGIN' 35 D(K,I)=SUM3 36 /END'3 37 /END'3 38 SHIFTBK(D)3 40 D(J,I)=D(J,I)+SA(I)3 42 'END'	28	AEND?!					
30 RAFD(B,2,H X)3 31 RAFD(B,2,H X)3 32 FOR'II'STEP'I'UNTIL'RD'DO''FOR'JI'STEP'I'UNTIL'RN'DO'D(I,J)A(I,J)3 45 CR'IIRN+1'STEP'I'UNTIL'RN'DO'BEGIN' 34 35 SUM=0; FOR'JI'STEP'I'UNTIL'RN'DO'SUMISUM+B(I-RN,J)+A(K,J)3'END'3 35 D(K,I)ISUM3 36 FEND'3 37 FEND'3 38 SHIFTBK(D)1 39 SHIFTBK(D)1 40 D(J,I)I'STEP'I'UNTIL' RD'DO''FOR'III'STEP'I'UNTIL'NN'DO' 40 D(J,I)I'SA(I)3 42 *END'	ZY	WAYARR (2500)1					
31 32 32 FOR'II'STEP'I'UNTIL'RD'DO'FOR'JI'STEP'I'UNTIL'RN'DO'D(I,J)A(I,J) 33 FOR'KI'STEP'I'UNTIL'RD'DO'BEGIN' 34 55 55 55 56 57 57 57 57 57 57 57 57 57 57	30	RAFD(B+2+H X)					
<pre>32 *FOR*K=1*STEP*1*UNTIL*RD*DO**BEGIN* 34 35 35 35 36 37 38 38 39 39 39 40 39 40 40 40 41 42 *END* 37 40 40 40 40 40 40 40 40 40 40</pre>	31	# TOR ! 1=1 ! STEP ! 1	PUNTIL R	D'D0**FOR+J=1*S	TEP'1'UNTIL/RN/D	0'D(I,J)_A(1,J).	3
33 34 35 35 36 37 37 38 39 39 39 39 30 40 30 40 30 40 31 40 31 40 35 40 40 40 40 40 40 40 40 40 40	32	+FOR K=1 STEP 1	PUNTILPR	D'DO'BEGIN'			
37       SUM=0; *FOR*J=1*STEP*1*UNTIL*RN*DO*SUM=SUM+B(I=RN*J)*A(K*J)!*END*1         36       D(K*I)=SUM;         36       *END*1         37       *END*1         38       *END*1         39       SHIFTBK(D):         40       D(J*I)=D(J;I)+SA(I);         41       D(J*I)=D(J;I)+SA(I);         42       *END*	<del>,</del>		EP/1/UNTI	L'NN'DO'BEGIN'			
35 36 37 37 37 38 39 39 40 50 40 50 40 50 50 50 50 50 50 50 50 50 5	34	SUN=0: FOR+J=14	STEP+1+U	NTIL RN+ DO+ SUM=	SUM+B(I=RN,J)+A(	K+J}}/END*3	
36 37 38 39 39 40 40 40 41 41 42 41 42 41 42 41 42 41 42 41 42 41 42 41 41 42 41 41 42 42 42 43 43 44 44 44 44 44 44 44 44	35	D(KAI)=SUMI				•	
37 38 39 39 40 40 50R ² J=1 ² STEP ² 1 ² UNTIL ² RD ² DO ² FOR ² I=1 ² STEP ² 1 ² UNTIL ² NN ² DO ² 41 42 42 42 42 42 42 43 42 43 44 45 45 45 45 45 45 45 45 45	36	PEND 1					
38 39 39 40 50R ² J=1 ² STEP ² 1 ² UNTIL ² RD ² DO ² FOR ² I=1 ² STEP ² 1 ² UNTIL ² NN ² DO ² 40 41 42 42 42 42	3/	/FND/1				•	
40 D(J,I)=D(J,I)=SA(I); 42 ************************************	38	SHIFTBK(D)1 -					
40 D(J,1)=D(J,1)+SA(1)3 41 *END*	34	FOR J=1 STEP	1ºUNTIL!	RD'DO''FOR'1=1'	STEP 1 UNTIL NN	D0.	
41 42 'END'	40	+(1,L)D=(1,L)D=	SA(1)3				
4Z - 2410	41						
	42	ч шиш ^а					

APPENDIX 3 THICK TWISTED PLATE ENERGY DATA

The sets of numbers below define terms in the energy expressions for the shell of the form:

Where F and G are A,B or C, and the CE(IC) terms are defined in the routines COEFF1,2,3,4 as follows:

(a) COEFF1 AA, BB, CC Strain Energy Terms

(b) COEFF2 AB, AC, BC Strain Energy Terms

(c) COEFF3 AA, BB, CC Kinetic Energy Terms

(d) COFFF4 AB, AC, BC Kinetic Energy Terms

and by the CO and CN arrays defined in MAIN.

The true energy terms can be obtained by multiplying these terms by the following scale factors:



for the strain energy

for the kinetic energy

Where E is Young's modulus,  $\eta$  Poisson's ratio,  $\rho$  the density and q the length parameter defined such that the length of the shell is q times the total twist angle BI.

1. AA STRAIN ENERGY TERMS

	(a)	TERMS	FOR	WHICH	ID	H	KD	AND	JD		HD
	ID	JD	JS	JC	IC						
	0000011112	1 2 2 3 0 0 1 1 2 2 3 0 0 1	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	-3 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5 -5	$\begin{array}{c} 1 \\ 2 \\ 6 \\ 7 \\ 8 \\ 11 \\ 14 \\ 15 \\ 17 \\ 18 \\ 19 \\ 21 \\ 22 \end{array}$						
	(b)	OTHER	TERI	JS							
	ID	JD	KD	HD	JS		JC	IC			
	0 0 0 0 1 0 0 0 1	1 2 3 1 1 3 0	1 1 1 2 0 2 2 1	1 0 2 1	11110000		002201111	4 9 10 12 20 3 5 13 16			
2.	BB :	STRAIN	ENE	RGY TE	RLIS.						
	(a)	TERMS	FOR	WHICH	ID		KD	AND	JD	=	ED
	ID	JD	JS	JC	IC						
	0000000000111122	000001111200000000000000000000000000000	20222020200420220000	-57-135375313553511131	23 24 25 26 27 28 447 48 9 51 77 77 73 74 6 77						

137.
IC

36

43

64

80

3. CC STRAIN ENERGY TERMS

(a)	TERMS	FOR	WHICH	ID	=	KD	AND JD	H	HD
ID	JD	JS	JC	IC					
0	0	0	1	00					
0	0	0	-	09					
0	0	0	5	90					
0	1	2	1	94					
0	2	0	1	96					
1	0	.2	3	100					
2	0	0	5	102					
1	1	Õ	ŝ	103					
(b)	OTHER	TERM	as						
ID	JD	KD	ĦD	JS		JG	IC		
0	0	1	0	1		4	92		
0	1	1	1	1		2	95		
0	0	1	0	1		2	07		
0	2		0			4	101		
1	0	2	0	1		4	101		
0	0	2	0	0		3	91		
0	0	5	0	0		5	93		
0	5	2	0	0		3	98		
4. AB 51	PRAIN J	ENERC	Y TER	<u>IIIS</u>					
ID	JD	KD	ED	JS		JC	IC		
0	1	1	0	0		-3	5		
0	1	1	0	0		1	6		
0	1	1	0	2		-1	7		
0	1	1	0	0		_1	ġ		
0	-	2	0	0			10		
0	1	2	0	U		-	10		
0			4	0		-1	11		
0	5	1	1	0		-5	16		
0	2	1	1	0		-1	17		
0	2	1	1	2		-3	18		
0	3	1	0	0		-1	23		
0	3	1	0	0		-3	24 .		
0	3	3	0	0		1	26		
0	5	1	2	0			07		
0	2		4	0		-2	21		
1	0	0	1	0		-3	28		
1	0	0	1	0		1	· 29		
1	0	0	1	2		-1	30		
1	0	2	1	0		1	32		
1	1	0	0	0		-5	33		
1	1	0	0	0		-1	.34		
1	1	0	0	2		1	35		
		0	0	2			20		
1		U	0	2		-3	30		
1	1	0	5	5		-3	31		
1	1	5	0	0		-3	42		
1	1	2	.0	0		1	43		
1	1	2	0	2		-1	44		
1	2	0	1	0		-1	47		
1	2	0	1	2		- 3	18 .		
1	2	2	1	. 0		1	50		
'	2	4	'	0		-1	50		

	t		F	
	l		E	3

JD KD HD JS JC IC

22220000000000001111122222	111111112222333301111121111	1131000020002000211113110002	. 0002000201111002010000210020	00001111113111111111111111	1 1 1 4 0 2 2 0 6 2 4 2 2 4 4 2 0 4 0 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	54 55 57 58 1 2 3 4 9 2 3 15 90 12 25 18 91 45 6 91 2 5 3 6
5. AC SI	TD	ENERG	Y TER	<u>8.15</u> 15	TO	TC
10	.U.	AU	нIJ	12	JC	TC
0 0 1 1 1 1 1 1 2 0 0 0 0 0 1 1 1 2	12300012221122230121	1 1 0 0 2 0 0 0 2 1 0 0 0 2 0 1 1 1 0	10102010201102010101	02000200001111111111	20002402022111113111	60 63 66 77 77 77 79 91 62 64 50 73 68 78

### 6. BC STRAIN ENERGY TERMS

	ID	JD	KD	HD	JS	JC	IC	
	000000011231122200000112311112	001111112000012111001120000011121	00000022011011100211111001000201	1100022001111110102011001111102010	2200202020202020000111311111111111111	022200220220200202111111111111111111111	80 81 84 85 86 87 88 91 92 93 97 98 99 105 108 109 110 112 83 89 90 4 95 100 101 103 104 107 111	
7.	AA K	INTELC	ENE	RGY TE	राइ			
	(a)	TERIS	FOR	WHICH	ID	= KD	AND JD	= HD
	ID	JD	JS	JC	IC			
	0	0	0	-5	1			
	0	2	0	-5	2			
	1	1	0	-3	4 5			
•			U	- 5	1			
	(b)	OTHER	TER	MS				
	ID	JD	KD	HD	JS	JC	IC .	
	0	0	0	2	0	-3	2	

#### 8. BB KINEFIC ENERGY TERLS

	(a)	TERNS	FOR	WHICH	ID	= KD	AND JD	= HD
	ID	JD	JS	JC	IC			
	0 0 0 1 1 2 1	00010001	0 2 0 2 0 2 0 2 0 0	-7 -5 -7 -5 -7 -3 -3 -3 -3 -5	6 7 8 13 15 16 18 19			
	(b)	OTHER	TERI	IS				
	ID	JD	KD	HD	JS	JC	IC	
	0 0 1 0	0 0 1 0 0	1 1 2 2 2	0 0 1 0 0	1 1 1 0 0	-4 -6 -4 -3 -5	9 10 14 17 11 12	
2. (	C KI	CUETIC	ENE	GY PE	RMS			
	(a)	TERMS	FOR	WHICH	ID	= KD	AND JD	= HD
	ID	JD	JS	JC	IC			
	0 0 0 1	0 0 1 0	0 0 0	-3 1 -1 1	20 21 22 23			
10.	AB I	CINEPIC	<u>) IN</u>	ergy T	URMS	1		
	ID	JD	KD	0E	JS	JC	IC	
	0 0 1 1 0 0 1	0 1 2 1 1 0 1 2 1	1 1 1 0 2 0 0 0 1	1 0 0 0 1 0 1 0	0000001111	-3 -5 -5 -5 -5 -3 -4 -4 -4 -4 -4	2 4 6 7 8 10 1 3 5 9	•
11.	AC J	INEPIC	I HAI	SRGY TI	ERI IS	1		
	ID	JD	KD	HD	JS	JC	IC	
	1 0 0	1 2 0	0 1 1	1 0 0	0 0 0	-2 -2 0	11 12 13	

OWNER

#### 12. BC KINEFIC ENERGY TERMS

ID	JD	KD.	HD	JS	JC	IC
0	0	0	1	0	-2	14
0	0	0	1	0	-2	15
2	0	0	1	0	-4	18
1	1	1	0	0	-2	19
0	1	1	0	1	-3	16
1	0	0	1	1	-3	17

.

1430

## APPENDIX 4 REDUCTION PARTS OF THIN CYLINDER COMPUTER PROGRAM

144.

#### BCMAT

	PUBLIC	ALIDLIAIIBIIKSIHIZILIAILIAILIAIIMINIJAIJAIJAIJAIJAIJAIJAIJAIJAIJAIJAIJAIJAI
· · · · · · · · · · · · · · · · · · ·	PUBLIC PUBLIC	NA, NB, NC, ND, NN, K, CN, G/, CU, B/, CNDF (1/) CUDF (1/)
** ** **	PUBLIC	<pre>sB(200)/SS(200)/SBDP(1)/SCDP(1)</pre>
- <u>27 - 27 - 27 - 2</u> - 27 - 27 - 27 - 27 -	PUBLIC	1X(4,4)/FA(40)/FB(30)/GA(40)/GB(30)/NMA(4)/GB0P(1)
. 6	PUBLIC	NMB(4),NBDP(1)
7	PUBLIC	IXDP(2),FADP(1),FBDP(1),GADP(1),NMDP(1)
. 8	REAL K	Sin ang panganan na ang panganan kanan na ang panganan na ang panganan na ang panganan na ang panganan ang pan Ang panganan na ang panganan na
·	INTEGE	R R , R H , R N , R X , F A , F B , G A <u>Contraction of the second contraction</u> of the second state of the second secon
10	INTEGE	
11 (11 (11 (11 (11 (11 (11 (11 (11 (11	ARCEINA	LIAININN/J/ NIEVER NY ARAA AMAMPIANA AMAMPIANA AMAMPIANA AMAMPIANA AMAMPIANA AMAMPIANA AMAMPIANA AMAMPIANA AMA Amampiana amampiana am
13	PEAL PEINS	n kana dalam da ang kana na na sa
14	PREAL AX, B	
	INTEGERIN	AB》NABCI (11)에 온 14년에 대한 방법에 대한 방법에 가지 않는 것 같은 것 같은 것 같은 것 같은 것 같이 가지 않는 것 같은 것 같은 것 같이 있는 것 같이 있는 것 같이 있는 것 같이
16	INTEGER I	• J • K • H • CC • M × • K × • H × 8
17	MAXARR(250	📭 the stand of the second and the second s
10	M=N*//22	n an
· · · · · · · · · · · · · · · · · · ·	UDITEA(SEA	5 C + 7 ( + 5 C F   + ) + ) 1
21	WRITEA(85)	\$6, / ( / \$\$F[ / } / }]
22	BEGIN	
23	JARRAY A(1	
24	NAB=NA+NB;	NABC=NAB+NC3
25	MONITOS	
20	NABENADEZ-	na Nazotna:Mzotn:
28	NUI L(APRXP	2 NN }
29	*FoR*1=1*5	TEP'1'UNTIL'M'DO''FOR'JE1'STEP'1'UNTIL'N+1'DO''BEGIN'
30	AX=AL++(1-	2) J. C.
31	CC=0;	
32	BX=BL++(J=	】}}NX3{J♥}}*N+ },
. 33	K=(1+1)*/*	ζζηματική με τη μεταγραφική με τη μεταγραφική του μεταγραφική του διατογραφική του του του του του του του του Αφτήρημα 2 από το 2 από του
34	ATER DECEN	TINENTTOVIVTOVIJA – po obječno u tekonomický se
35	A(CC+.1+NA)	*ETA*J *AX*AL*B 13
37	84++++1F+J	'E0'1'OR'1'E0'1'THEN''GOTO/C33
38	ACCC+J,NA+	Mx=M) = { [= ] } + Ax + A [ }
39	C3+++1F+J	LEP2PTHENPGOTOPES
40	ACCC+J, NAB	* NX } # AX # AL }
		*N*14 •17 HENJ\$ COVO\$451
42	*1E*1*EQ*1	PORIJEGI ITTENI GOTOPBII
-INFERT 44	-XH (CC+J) HX	H)*(1-1)*AX*A12
45	B2++++1F+J	'EQ'N'THEN'' GOTO'ASI
46	ALCC+J, NA+	MX)=J+AX+AL#B[].collocates of the Addression s <u>inctuments</u> of this is a second second second second second second
47	A51+, CC=CC	NI CARACTERISTICS AND
48	PIF JPGEPN	THENY'GUTO'CIJ (1980) - Provinský do počel manna z promínik sme provinské statem v state se state se state se Tiep Yabilis
47	AIC+KJMA/	1 HFN// GOTO/C11
51	A(rC+1)//2	*NA+MX)=ETA+(1-1)+BX+BL+A13
52	C1++++1F+	LE' 2' THEN'' GOTO'ALS
53	AICC+K, NAB	+HX)=ETA+BX3
54	A1*+*CC=CC	+H'/'2;
55	CC=CC=13	an an an an ann an Anna an Anna An Anna Anna
56	IF JGE'N	
57	*1F-1/20/3	ANX 38 ( PO ) SEX SELENT STATE OF A CONTRACT OF
<u></u>	RIJAJCCI	K,NA+MX)=J+BX+BL
60 K	A2/ . / / END	
61	BEGIN'	
62	FREAL SUN	n han di sedi di seriente della d Nel 1997 e della
63	ARRAY'CI	(1°•°RX,1°•°NN)3
64	PROCEDURE	SCALE(A,RI)} ARRAY AJ'INTEGER'RI
65	BEGIN	TEGER 1JJJKJ KATARRAT JSKI TOTRIJJ.
00 <u>.</u> 00 <u>.</u>		) FL. 1. AUITT. KI. AA., DEATU, Torre in How and and and a second source in the second s
68	/FoRf.is1/	TEP'1'UNTIL'NN'DO''BEGIN'
69	AX=ABS(A()	<pre>(FJ)}}</pre>
70	IF'AX'GT	SU'IT THEN SUM AXS
71	'END';	
72	JS(1)#1/5	JMJ 'END'J
73	/FOR*1=1*	SIEP'I'UNTIL'RI'DO''FOR'J#1'STEP'I'UNTIL'NN'DO'
74 - 74 - 74 - 74 - 74 - 74 - 74 - 74 -	ACIDIANS	) ( / * ^ ) [ / U / ) <u>An an an</u>
72	NARSNAR42	M:NN#2+NN:M=M*/*2;
77	*FOR*1=1*	STEP' I' UNTIL'NN' DO'' BEGIN!
78	SUN#0;	
	*FOR*J=1*	STEP 1 / UNTIL 'N+1 'DO'SUH=SUM+A(J) 1)+BL++(J-1)1
80	275 A(N+1,1)=	SUMJYENDYJ – Provenské provenské provenské provenské provenské provenské provenské provenské provenské provensk

	81		+FOR INA+1+STEF'1'UNTIL'NN'DO''BEGIN
	82		A(2*N+H+1+1)]=A(N+1+1)}A(N+1+1)=O}*END*3
	83		+FOR*1=1*STEP*1*UNTIL*NA*DO*A(2+N+H+1+1)+03
	84		+FOR+ J=1+STEP+1+UNTIL+RX+DO++BEGIN+
	85		FOR I I I STEP 2' UNTIL'NA'DO''BEGIN
	86	2	K=1*/2+12C(J,K)=A(J,1)2D(J,K)=A(J,1+1)2*END*2
	87	1.1.1	FOR' ISNA+1'STEP'2'UNTIL'NAB'DO''BEGIN'
	88		K=1*//2+15C(J,K)=A(J,1+1);D(J,K)=A(J,1);PEND*5
	89		FORPINAB+1'STEP'2'UNTIL'NN'DO'BEGIN
	90		$K = 1^{2}/2 + 1 + C(J + K) = A(J + 1) + D(J + K) = A(J + 1) + 1 + 1 + 2^{2} = ND^{2}$
	91		/FND/ t
	92		NN = NN / / 21
승규들은	03		FOR + = 1 + STEP 1 + UNT 11 + RX + DO + FOR + J= 1 + STEP 1 + UNT 11 + NN + DO +
	94		/BEGIN/C(1/1)=C(1/1)+SS(1)1D(1/1)+SB(1)+SB(1)1/FND/1
• • • •	95		
10.0 m 14	96		CHATCOLC, RX, NNA 1 EMATPR(D, RX, NNA 4)1
	07	1 	$WRITE_A(AS, C, *(*FLBC*)*) WRITE_A(AS, D, *(*FLBC*)*) F$
ri where is r	0.8	· • · · ·	
1 <u>1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 </u>	60		
		and and and a	
	100	1. A	
	141	- · · -	- ՀՀԱՅԱՆԴ ԳԱՅԴԱՅԴ ԳԱՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅՅ

# MAIN For use with BCRED

DJ

<pre> 1</pre>			· · · · ·		1	
2       PUBLIC NA, NB, NC, ND, NR, RC(6), CO(6), CNDP(1), CODP(1)         3       PUBLIC RM, RN, RX, AA         4       PUBLIC SS(200), SS(200), SBDP(1), SSDP(1)         5       PUBLIC IX(4,4), FA(2,40), GA(2,40), NMA(4), FADP(2), IXDP(2), GADP(2)         6       PUBLIC MA(2), MADP(1)         7       PUBLIC NMB(4), NBDP(1)         8       PUBLIC NMB(4), NBDP(1)         9       REAL KS         10       INTEGER R, RM, RX, FA, GA         11       >BEGIN#         12       /INTEGER R, RM, RX, FA, GA         13       SUPDV1(CN, CNDP, 6) IS UPDV1(CO, CODP, 6) IS         14       SUPDV1(NMB, MBDP, 4) I         15       SUPDV1(SB, SBDP, 200) IS UPDV1(SS, SSDP, 200) I         16       SUPDV1(IX, IXDP, 4, 4) IS UPDV2(FA, FADP, 2, 40) I         17       SUPDV2(GA, GADP, 2, 4) IS UPDV1(NMA, NMDP, 4) I         18       SUPDV2(GA, GADP, 2, 40) IS UPDV1(NMA, NMDP, 4) I         19       START*, * ININT(M) I* IF* M* EQ* 99* THEN* STOP I         20       ININT(M) I         21       NA = NB=NC = M* N= MI         22       NN = 3* M* N = 3* MI         23       R=NAJ         24       RX = RM = 2* M + 2* NIS         25       RN=NN=RMI         26       INREAL(AL, BL, ETA,	ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا ا	1 PUBLIC	: AL>BL>AI>BI>KS+H	TZ,ETA,ETAZ,ETA	3 HT MANA JHA JNA J	A JB JC
3       PUBLIC RM, Rh, RX, AA         4       PUBLIC SB(200), SS(200), SBDP(1), SSDP(1)         5       PUBLIC IX(4,4), FA(2,40), GA(2,40), NMA(4), FADP(2), IXDP(2), GADP(2)         6       PUBLIC MA(2), MADP(1)         7       PUBLIC NMB(4), NBDP(1)         8       PUBLIC NMB(4), NBDP(1)         9       REAL KS         10       INTEGER R, RM, RN, RX, FA, GA         11       / BEGIN#         12       / INTEGER / I, J, K3         13       SUPDV1 (CN, GNDP, 6) I SUPDV1 (CO, CODP, 6) I         14       SUPDV1 (NMB, NBDP, 4) J         15       SUPDV1 (SS, SSDP, 200) I         16       SUPDV1 (SS, SSDP, 200) I         17       SUPDV1 (SS, SSDP, 200) I         18       SUPDV2 (TX, IXDP, 4, 4) I SUPDV2 (FA, FADP, 2, 40) I         19       START+, * ININT(M) I* IF * M* EQ* 99* THEN* STOPI         20       ININT(N) I* IF * M* EQ* 99* THEN* STOPI         21       NA=NBENC=M***I         22       NN=3 * M* N=3 * MI         23       R=AJ         24       RX=RM=2 * M+2 * NI         25       RN=NN=RMI         26       INREAL(AL, BL, ETA, HT, KS) I		2 PUBLIC	NA, NB, NC, ND, NN, R	CN(6),CO(6),CN	DP(1),CODP(1)	
<pre>4</pre>		3 PUBLIC	: RM+RN+RX,AA 🔅 👘	e e marine a a construire e const		
5       PUBLIC IX(4,4),FA(2,40),GA(2,40),NMA(4),FADP(2),IXDP(2),GADP(2);         6       PUBLIC MA(2),MADP(1)         7       PUBLIC MMDP(1)         8       PUBLIC MB(4),NBDP(1)         9       REAL KS         10       INTEGER R,RM,RN,RX,FA,GA         11       /BEGIN*         12       /INTEGER/I*J,JKI         13       SUPDV1(CN,CNDP,6);SUPDV1(CO,CODP,6);         14       SUPDV1(NMB,NBDP,4);         15       SUPDV1(SS,SSDP,200);SUPDV1(SS,SSDP,200);         16       SUPDV2(IX,IXDP,4+4);SUPDV2(FA,FADP,2,40);         17       SUPDV2(IX,IXDP,4+4);SUPDV2(FA,FADP,2,40);         18       SUPDV2(GA,GADP,2+40);SUPDV1(NMA,NMDP,4);         19       START*, *ININT(M);*IF*M*EQ*99*THEN*STOP;         20       ININT(N);         21       NA:NB:NC:M*N=M;         22       NN:3*HN=3*M;         23       R=NA;         24       RX:RM:2*H*2*N;         25       RN:NN=RM;         26       INREAL(AL,BL,ETA,HT,KS);	1	4 PUBLIC	SB(200), SS(200),	SBDP(1), SSDP(1)	· · · · · · · · · · · · · · · · · · ·	
6 PUBLIC MA(2), MADP(1) 7 PUBLIC NMDP(1) 8 PUBLIC NMDP(1) 9 REAL KS 10 INTEGER R, RM, RN, RX, FA, GA 11 'BEGIN' 12 'INTEGER R, RM, RN, RX, FA, GA 11 'BEGIN' 12 'INTEGER R, RM, RN, RX, FA, GA 13 SUPDV1(CN, CNDP, 6) I SUPDV1(CO, CODP, 6) I 14 SUPDV1(NMB, NBDP, 4) I 15 SUPDV1(NMB, NBDP, 4) I 15 SUPDV1(NMA, MADP, 2) I 16 SUPDV1(MA, MADP, 2) I 17 SUPDV2(IX, IXDP, 4, 4) I SUPDV2(FA, FADP, 2, 40) I 18 SUPDV2(GA, GADP, 2, 40) I SUPDV1(NMA, NMDP, 4) I 19 START, 4, 1NINT(M) I / IF M' EQ' 99' THÈN' STOPI 20 ININT(N) I 21 NA = NB = NC = MM = MI 22 NN = 3 + M + N = 3 + MI 23 R = NA J 24 RX = RM = 2 + M + 2 + N I 25 RN = NN = RM = 2 + M + Z + N I 26 INREAL(AL, BL, ETA, HT, KS) I 27 VIENTIAL CALL A BL,		5 PUBLIC	[X(4,4),FA(2,40)	GA(2)40),NMA(4	J.FADF(2), IXDF(2	J-GADF(2)
7       PUBLIC NMDP(1)         8       PUBLIC NMB(4),NBDP(1)         9       REAL KS         10       INTEGER R,RM,RN,RX,FA,GA         11       *BEGIN*         12       *INTEGER R,RM,RN,RX,FA,GA         11       *BEGIN*         12       *INTEGER *I,J,Ki         13       SUPDV1(CN,CNDP>6):SUPDV1(CO,CODP.6):         14       SUPDV1(NMB,NBDP,4):         15       SUPDV1(SB,SBDP,200):SUPDV1(SS,SSDP,200):         16       SUPDV1(MA,MADP.2):         17       SUPDV2(IX,IXDP.4+4):SUPDV2(FA,FADP.2,40):         18       SUPDV2(GA,GADP.2+40):SUPDV1(NMA,NMDP,4):         19       START.*, *ININT(M):*IF*M*EQ*99*THEN*STOP:         20       ININT(N):         21       NA=NB=NC=M*N=M:         22       NN=3*M:         23       R=NA;         24       RX=RM=2*M+2*N:         25       RN=NN=RM;         26       INREAL(AL,BL,ETA,HT,KS);	<u> </u>	6 PUBLIC	MA(2) + MADP(1)	. *		
8       PUBLIC NMB(4), NBDP(1)         9       REAL KS         10       INTEGER R, RM, RN, RX, FA, GA         11       /BEGIN ⁴ 12       /INTEGER 1, J, KI         13       SUPDV1(CN, CNDP, 6); SUPDV1(CO, CODP, 6);         14       SUPDV1(NMB, NBDP, 4);         15       SUPDV1(SB, SBDP, 200); SUPDV1(SS, SSDP, 200);         16       SUPDV1(MA, MADP, 2);         17       SUPDV2(IX, IXDP, 4, 4); SUPDV2(FA, FADP, 2, 40);         18       SUPDV2(IX, IXDP, 4, 4); SUPDV1(NMA, NMDP, 4);         19       START, * ININT(M); IF*M*EQ*99*THEN*STOP;         20       ININT(N);         21       NA=NB=NC=M*N=M;         22       NN=3*M*N=3*M;         23       R=NA;         24       RX=RM=2*M*2*N;         25       RN=NM;         26       INREAL(AL, BL, ETA, HT, KS);	Ref. (An April 1	7 PUBLIC	NMDP(1)	mara - tarini tanimi		
9       REAL KS         10       INTEGER R, RM, RN, RX, FA, GA         11       'BEGIN'         12       'INTEGER' I, J, KJ         13       SUPDV1(CN, CNDP, 6); SUPDV1(CO, CODP, 6);         14       SUPDV1(NMB, NBDP, 4);         15       SUPDV1(SB, SBDP, 200); SUPDV1(SS, SSDP, 200);         16       SUPDV2(IX, IXDP, 4, 4); SUPDV2(FA, FADP, 2, 40);         17       SUPDV2(GA, GADP, 2, 40); SUPDV1(NMA, NMDP, 4);         18       SUPDV2(GA, GADP, 2, 40); SUPDV1(NMA, NMDP, 4);         19       START, * ININT(M); IF*M*EQ* 99* THEN* STOP;         20       ININT(N);         21       NA=NB=NC=M*N=M;         22       NN=3#MN=A3*M;         23       R=NA;         24       RX=RM=2+M*2+N;         25       RN=NN=RM;         26       INREAL(AL, BL, ETA, HT, KS);	المار مشتر <del>المتحاضية (</del>	8 PUBLIC	NMB(4),NBDP(1)			
10 INTEGER R, RM, RN, RX, FA, GA 11 *BEGIN* 12 *INTEGER*I*J>K3 13 SUPDV1(CN*CNDP*6)3SUPDV1(CO*CODP*6)3 14 SUPDV1(NMB*NBDP*4)3 15 SUPDV1(SB*SBDP*2003SUPDV1(SS*SSDP*20033 16 SUPDV1(MA*MADP*2)3 17 SUPDV2(IX*IXDP*4*4)3SUPDV2(FA*FADP*2,4033 18 SUPDV2(GA*GADP*2*40)3SUPDV1(NMA*NMDP*4)3 19 START***ININT(M)3*IF*M*EG*99*THEN*STOP3 20 ININT(N)3 21 NA=NB=NC=M*N=M3 22 NN=3*M*N=3*M3 23 R=NA3 24 RX=RM=2*M*2*N3 25 RN=NN=RM3 26 INREAL(AL*BL*ETA*HT*KS)3 27 INREAL(AL*BL*ETA*HT*KS)3 26 INREAL(AL*BL*ETA*HT*KS)3	1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	9 REAL K	S S			
<pre>11</pre>		0 INTEGE	R RORMORNORXOFADG	A	an a	nen den anne an an i anne anne an
<pre>12 / INTEGER/I/J/K3 13 SUPDV1(CN, CNDP, 6) J SUPDV1(CO, CODP, 6) 3 14 SUPDV1(NMB, NBDP, 4) 3 15 SUPDV1(SB, SBDP, 200) J SUPDV1(SS, SSDP, 200) 3 16 SUPDV1(MA, MADP, 2) 3 17 SUPDV2(IX, IXDP, 4, 4) 3 SUPDV2(FA, FADP, 2, 40) 3 18 SUPDV2(GA, GADP, 2, 40) 3 SUPDV1(NMA, NMDP, 4) 3 19 START, 0, ININT(H) 3 / IF/M/EQ²99' THEN/STOP3 20 ININT(N) 3 21 NA=NB=NC=M*N=M3 22 NN=3*M*N=3*M3 23 R=NA3 24 RX=RM=2*M*2*N3 25 RN=NN=RM3 26 INREAL(AL, BL, ETA, HT, KS) 3 26 INREAL(AL, BL, ETA, HT, KS) 3 27 INF (AL, CAL, BL, ETA, HT, KS) 3 28 SUPDV2(GA, GADP, 2, 40) 3 SUPDV2(GA, GADP, 2, 40) 3 29 START, 0, ININT(H) 3 SUPDV2(FA, FADP, 2, 40) 3 20 ININT(N) 3 21 NA=NB=NC=M*N=M3 22 NN=3*M*N=3*M3 23 R=NA3 24 RX=RM=2*M*2*N3 25 RN=NN=RM3 26 INREAL(AL, BL, ETA, HT, KS) 3 27 INREAL(AL, AL, BL, ETA, HT, KS) 3 28 SUPDV2(SA, SADP, 2, SA</pre>		1. BEGINA	n transformation and the second	· · · · · · · · · · · · · · · · · · ·		
<pre>13 SUPDV1(CN, CNDP, 6); SUPDV1(CO, CODP, 6); 14 SUPDV1(NMB, NBDP, 4); 15 SUPDV1(SB, SBDP, 200); SUPDV1(SS, SSDP, 200); 16 SUPDV1(MA, MADP, 2); 17 SUPDV2(IX, IXDP, 4, 4); SUPDV2(FA, FADP, 2, 40); 18 SUPDV2(GA, GADP, 2, 40); SUPDV1(NMA, NMDP, 4); 19 START, *, ININT(H); IF, M, EQ*99, THEN, STOP; 20 ININT(N); 21 NA=NB=NC=M*N=M; 22 NN=3*M*N=-3*M; 23 R=NA; 24 RX=RM=2=M+2=N; 25 RN=NN=RM; 26 INREAL(AL, BL, ETA, HT, KS); 26 INREAL(AL, BL, ETA, HT, KS); 27 SUPDV: S</pre>	1	2 / INTEGER / I	JaK3			· · · · · · · · · · · · · · · · · · ·
<pre>14 SUPDV1(NMB,NBDP,4); 15 SUPDV1(SB,SBDP,200);SUPDV1(SS,SSDP,200); 16 SUPDV1(MA,MADP,2); 17 SUPDV2(IX,IXDP,4,4);SUPDV2(FA,FADP,2,40); 18 SUPDV2(GA,GADP,2,40);SUPDV1(NMA,NMDP,4); 19 START,*,*ININT(H);*IF*M*EQ*99*THEN*STOP; 20 ININT(N); 21 NA=NB=NC=M*N=M; 22 NN=3*M*N=3*M; 23 R=NA; 24 RX=RM=2+M+2+N; 25 RN=NN=RM; 26 INREAL(AL,BL,ETA,HT,KS);</pre>	i	3 SUPDVICCNA	CNDP+6); SUPDV1(CO	CODP+611		an and a second s
15       SUPDV1(SB,SBDP,200)JSUPDV1(SS,SSDP,200)J         16       SUPDV1(MA,MADP,2)J         17       SUPDV2(IX,IXDP,4,4)JSUPDV2(FA,FADP,2,40)J         18       SUPDV2(GA,GADP,2,40)JSUPDV1(NMA,NMDP,4)J         19       START*•*ININT(M)J*IF*M*EQ*99*THEN*STOPJ         20       ININT(N)J         21       NA=NB=NC=M*N=M#         22       NN=3*M*N=3*MJ         23       R=NAJ         24       RX=RM=2+M+2+NJ         25       RN=NN=RMJ         26       INREAL(AL,BL>ETA,HT,KS)J	1. · · · · · · · · · · · · · · · · · · ·		3+NBDP+412			
16       SUPDV1(MA, MADP,2);         17       SUPDV2(IX, IXDP,4,4); SUPDV2(FA, FADP,2,40);         18       SUPDV2(GA, GADP,2,40); SUPDV1(NMA, NMDP,4);         19       START, * ININT(M); IF*M*EQ*99*THEN*STOP;         20       ININT(N);         21       NA=NB=NC=M*N=M;         22       NN=3*M*N=3*M;         23       R=NA;         24       RX=RM=2+M+2+N;         25       RN=NN=RM;         26       INREAL(AL, BL, ETA, HT, KS);	F PP i	5 SUPDVI(SB)	SBDP,200) JSUPDV1(	SS.SSDP.20011	1.2. 1997년 1월 1992년 1월 19일7년 1211년 13일 1991년 1991년 1월 1992년 18일7년 18일7년 18일7년 18일7년 18일7년 18일7년 18일7년 18일7년 18일	
17       SUPDV2(IX,IXDP+4+4); SUPDV2(FA,FADP,2,40);         18       SUPDV2(GA,GADP,2+40); SUPDV1(NMA,NMDP,4);         19       START+0*ININT(M);*IF*M*EQ*99*THEN*STOP;         20       ININT(N);         21       NA=NB=NC=M*N=M;         22       NN=3*M*N=3*M;         23       R=NA;         24       RX=RM=2+M+2+N;         25       RN=NN=RM;         26       INREAL(AL,BL>ETA>HT>KS);	· · · · · · · · · · · · · · · · · · ·	6 SUPDVICMA	HADP+2)1			a serie and the series of the
18       SUPDV2(GA,GADP,2,40);SUPDV1(NMA,NMDP,4);         19       START,* *ININT(M);*IF*M*EQ*99*THEN*STOP;         20       ININT(N);         21       NA=NB=NC=M*N=M;         22       NN=3*M*N=3*M;         23       R=NA;         24       RX=RM=2*M*2*N;         25       RN=NN=RM;         26       INREAL(AL,BL>ETA)HT>KS);		7 SUPDV2(IX)	1xDP+4+4)1SUPDV2(	FA, FADP, 2, 4011		
19       START*•*ININT(H)3*IF*M*EQ*99*THEN*STOP;         20       ININT(N);         21       NA=NB=NC=M*N=M;         22       NN=3*H*N=3*M;         23       R=NA;         24       RX=RM=2*M*2*N;         25       RN=NN=RM;         26       INREAL(AL, BL, ETA, HT, KS);		8 SUPDV2(GA	GADP+2+40)ISUPDV1	(NMA+NMDP+4)3		
20 JNINT(N); 21 NA=NB=NC=M*N=M; 22 NN=3*M*N=3*M; 23 R=NA; 24 RX=RM=2*M+2*N; 25 RN=NN=RM; 26 INREAL(AL,BL,ETA,HT,KS);		9 STARTA-PIN	INT(N) IF TEPMPEOPO	91 THENISTOP:		
21 NA=NB=NC=M*N•M3 22 NN=3*M*N-3*M3 23 R=NA3 24 RX=RM=2*M+2*N3 25 RN=NN-RM3 26 INREAL(AL,BL,ETA,HT,KS)3	2					anna anna anna a faoinn a faoinn an hann an thar ann ann ann ann ann ann ann ann ann a
22 NN=3+M+N-3+M; 23 R=NA; 24 RX=RM=2+M+2+N; 25 RN=NN-RM; 26 INREAL(AL,BL,ETA,HT,KS);	- 2	1 NAENBENCER	4 <b>₩NwM</b> ∄ -			(a) appropriate and a second s second second secon second second sec
23 R=NAJ 24 RX=RM=2+M+2+N; 25 RN=NN-RM; 26 INREAL(AL,BL,ETA,HT,KS);	2	2 NN=3+M+N=3	3 19 - 19 - 1100 to 10 - 10 - 1000 to 20 3 <b>4 M 1</b>	alan mendu mula suker sida melanda. T		ningeneringen auf gestellten. Annen in seinen säterning som en som
24 RX=RM=2+M+2+N; 25 RN=RM; 26 INREAL(AL,BL,ETA,HT,KS);	T 25 2	3 DENAL		and a second	e e de la desta de la desta La desta de la d	na an an ann an an an an an an an an an
25 RN=NN-RH; 26 INREAL(AL,BL,ETA,HT,KS);	2		+2+N1			a na Tana ang bang na ka si kana ang sara sa
26 INREAL(AL, BL, ETA, HT, KS);		5 PN=NN_PM+				
	2	6 INREALCAL	BIAFTAAHTAKSII		· · · · · · · · · · · ·	a antipatria de tatos de la seconda de la Tenenda de la seconda de la
THE ALL TIMPNEATTIPHINTEMAZIYSPAPENTATINTENAZIKSPALESSULT. THE STATEMATEMATEMATIC	2	7 I INFS(4)1	IPDINT(N.2))SPACES	COLIPPINTEN-2	LSPACESCALL	
$28 \qquad FPRINT(A \mid A) \downarrow SPACES(3) \downarrow FPRINT(A \mid A) \downarrow SPACES(3) \downarrow FPRINT(FTA \mid A) \downarrow$	- HTEET - E - E - E - E - E - E - E - E - E	8 FPRINT(AL	4)1SPACES(3)1FPRI	NT(RIAA)1 CPACES	S(3) IFPRINT/FTA.	a den eren er en en en en er en er

<pre>29 SPACEs(3):EPRINT(HT, 4):LINES(4): 30 AL=0.5=AL: 31 AI=1/AL:JAL=1:IBI=1/BL:BL=1: 32 HT=0.5=HT: 33 ETA2=TA3ETA;ETA3ETA5ETA2:HT2=HT+HT: 34 CO(1)=1=ETA3ETA5ETA2:HT2=HT+HT: 35 NMA(1)=1A;NA(2)=NB:NMA(3)=NC:NMA(4)=ND: 36 NMA(1)=0:NMA(2)=NB:NMA(3)=NC:NMA(4)=ND: 36 NMA(1)=0:NMA(2)=NB:NMA(3)=NC:NMA(4)=ND: 36 NMA(1)=0:NMA(2)=NB:NMA(3)=NC:NMA(4)=ND: 36 NMA(1)=0:NMA(2)=NB:NMA(3)=NC:NMA(4)=ND: 37 IX(1)=10:IX(2)=1:IX(1)=IX(3)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=2:IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(1)=IX(</pre>			146
<pre>29 SPACEs(3):EPRINT(HT.4):LINES(4): 31 Al=:/AL;AL=:ER:INE:(A]:EDRINT(HT.4):LINES(4): 31 AI=:/AL;AL=:EA;ETA;ETA;ETA;ETA;ETA;ETA;ETA;ETA;ETA;E</pre>		2	
30       AL=0.55AL;         31       Al=0.4L; 12Bl=1/BL; BL=11/3         32       HT=0.55AL;         33       ETA25TA4ETA; ETA3EETA4ETA2; HT2*HT*HT;         34       C0(1)=1-ETA; CN(1)=1/3;         35       NAA(1; TA; TA3EETA4ETA2; HT2*HT*HT;         36       NBA(1; TA; TA3EETA4ETA2; HT2*HT*HT;         37       IX(1:1)=0.1X(2:2)=RB; NMA(3)=RMA(1)+MMA(2);         38       NAA(1; TA3EACA; TA3EETA4ETA2; HT2*HT*HT;         39       NAA(1; TA3EACA; TA3EETA4; TA3EETA4; TA3EETA4; TA3EACA; TA3EACA		29	SPACES(3) FERINT(HT, 4) FLINES(4) FOR A CONTRACT STATES (4) FOR A CONTRACT STATES FOR A CONTRACT STATES (4) FOR A CONTRACT STATES FO
<pre>31 ALE/ALAC 1/31/31 33 ETA2E TAAE TA3E TA3E TAAE TA2JHT22HT+HT1 34 ETA2E TAAE TA3E TA3E TAAE TA2JHT22HT+HT1 35 NMA(1)=NAINAA(2)=NBINMA(3)=NA(1)=NMA(2) 36 NMB(1)=01NB(2)=NMA(1)INMB(3)=NMA(1)=NMA(2)] 37 IX(1,1)=01IX(2,2)=11X(3,3)=21IX(1,2)=31IX(1,3)=41IX(2,3)=51 38 /70R1=1,2'00'FOR'J=1+IYSTEP'IUNT1L'37D0'IX(J,1)=IX(1,J)] 39 NH2/HJ 40 UAJB=UGEHH 41 ININT(MA (1))3 42 /FOR1=1'STEP'IUNT1L'MA(1)'D0'ININT(FA(1+1))3 43 /FOR1=1'STEP'IUNT1L'MA(1)'D0'ININT(FA(1+1))3 44 ININT(MA(2))3 45 /FOR1=1'STEP'I'UNT1L'MA(2)'D0'ININT(FA(2,1))3 46 /FOR1=1'STEP'I'UNT1L'MA(2)'D0'ININT(FA(2,1))3 47 LINES(4)3 49 LINES(1)3 51 LINES(1)3 52 /FOR1=1'STEP'I'UNT1L'MA(1)'D0'IPRINT(FA(1+1)'A)33 53 LINES(1)3 54 /FOR1=1'STEP'I'UNT1L'MA(1)'D0'IPRINT(FA(2,1)'A)33 55 LINES(1)3 56 /FOR1=1'STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 57 LINES(1)3 58 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 59 LINES(1)3 50 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 51 LINES(1)3 52 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 53 LINES(1)3 54 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 55 LINES(1)3 56 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 57 LINES(1)3 58 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 59 LINES(1)3 50 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 51 LINES(1)3 52 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 53 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 54 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 55 / LINES(1)3 56 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 57 / LINES(1)3 58 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 59 / LINES(1)3 50 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 51 / LINES(1)3 52 / FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 53 / LINES(1)3 54 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 55 / LINES(1)3 56 /FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 57 / LINES(1)3 58 / FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 59 / LINES(1)3 50 / FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)33 50 / LINES(1)3 50 / LINES(1)3 50 / FOR1=1STEP'I'UNT1L'MA(2)'D0'IPRINT(FA(2,1)'A)</pre>		30	, na standar se
<pre>H1205901;4 H1205901;4 H1205901;4 C(1);4;=ETA;E(1);2;A3;ETA*ETA2;HT2*HT*HT3 C(1);4;=ETA;E(1);1/3;A3;A(1);AMA(2);MD2 MH(1);5;NMB(2)=NMA(1);MMB(3)=NMA(1);AMA(2); MH(1);5;NMB(2)=NMA(1);MMB(3)=NMA(1);AMA(2); FOR;11;2;OO'FOR'J=1;1;X(5;A3);2;1;X(1,2);3;1;X(1,2);5; FOR;11;2;EP+1;UNT1L'MA(1);DO'ININT(FA(1,1);); H200;FOR;11;S;EP+1;UNT1L'MA(1);DO'ININT(FA(1,1)); FOR;11;S;EP+1;UNT1L'MA(1);DO'ININT(FA(1,1)); FOR;11;S;EP+1;UNT1L'MA(1);DO'ININT(FA(1,1)); FOR;11;S;EP+1;UNT1L'MA(2);DO'ININT(FA(1,1)); FOR;11;S;EP+1;UNT1L'MA(2);DO'ININT(FA(1,1)); FOR;11;S;EP+1;UNT1L'MA(1);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(1);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(1);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(1);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(1);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(1);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(1);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(1,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;11;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;21;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;21;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;21;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;21;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;21;S;EP+1;UNT1L'MA(2);DO'IPRINT(FA(2,1);3); FOR;21;S;EP+1;UNT1L'MA(</pre>		31	Alti/AL;AL-I;BI=I/OL;BL=I;
<pre>34 C(1) = 1 - ETA;CN(1) = 1/3; 35 NHA(1) = NA;MyA(2) = NB;NHA(3) = NC;NHA(4) = ND; 36 NHB(1) = C(NHB(2) = NHA(1) = NHA(1) = NHA(2) =</pre>		32	HIEV050410
<pre>35 NMA(1)=N, N, N(2)=NB, NMA(3)=NC; NMA(4)=ND; 36 NMB(1)=O, NMB(2)=NMA(1); NMB(3)=NMA(1)+NMA(2); 37 TX(1,1)=O(1X(2,2)=1]X(3,3)=2]X(1,2)=3]X(1,2)=4] TX(2,3)=5] 38 PDR*1=1,2*DO*FOR*J=1; YSTEP*1*UNT1L*3*DO*TX(1,1); 39 M=2*N; 40 JA=JB=JC=N3 41 ININT(MA(1)); 41 ININT(MA(1)); 42 PFOR*1=1*STEP*1*UNT1L*MA(1)*DO*ININT(FA(1,1)); 43 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1 NINT(FA(2,1)); 44 ININT(MA(2)); 45 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1 NINT(FA(2,1)); 46 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1 NINT(FA(2,1)); 47 LINES(1); 48 IPRINT(MA(1),3); 49 LINES(1); 50 PFOR*1=1*STEP*1*UNT1L*MA(1)*DO*1PRINT(FA(1,1)*3); 51 LINES(1); 52 PFOR*1=1*STEP*1*UNT1L*MA(1)*DO*1PRINT(FA(1,1)*3); 53 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 56 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 57 LINES(1); 58 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 59 LINES(1); 50 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 51 LINES(1); 53 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 55 LINES(1); 56 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 57 LINES(1); 58 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 59 LINES(1); 50 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 51 LINES(1); 52 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 53 LINES(1); 54 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 55 LINES(1); 56 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 57 LINES(1); 58 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 59 LINES(1); 50 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 51 LINES(1); 52 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 53 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 54 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 55 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 56 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 57 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 57 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 58 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 59 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 50 PFOR*1=1*STEP*1*UNT1L*MA(2)*DO*1PRINT(FA(2,1)*3); 50 PFOR*1=1*STEP*1*UNT1*</pre>		5.55 34	$\mathbf{C}(\mathbf{A}) = \mathbf{C} \mathbf{A} \cdot \mathbf{C} \mathbf{N}(\mathbf{A}) = 1/3$
<pre>36 MMB(1)=0;NMB(2)=NMA(1)JNMB(3)=NMA(1)+NMA(2); 37 IX(1,)]=0;IX(2,2)=1;IX(3,3)=2;IX(1,2)=3;IX(1,3)=4;IX(2,3)=5; 37 MZ(1,)]=0;IX(2,2)=1;IX(3,3)=2;IX(1,2)=3;IX(1,3)=4;IX(2,3)=5; 39 MZ(1,3)=0;IX(1,3)=0;IX(1,3)=0;IX(1,3); 39 MZ(1,3)=0;IX(1,3)=0;IX(1,3)=0;IX(1,3); 41 ININT(MA(1)); 42 PFOR;II:STEP:I/UNTIL/MA(1)/DO'ININT(FA(1,1)); 43 PFOR;II:STEP:I/UNTIL/MA(2)/DO'ININT(FA(2,1)); 44 ININT(MA(2)); 45 PFOR;II:STEP:I/UNTIL/MA(2)/DO'ININT(FA(2,1)); 46 PFOR;II:STEP:I/UNTIL/MA(2)/DO'ININT(FA(2,1)); 47 LINES(4); 48 IPRINT(MA(1),3); 49 LINES(1); 50 PFOR;II:STEP:I/UNTIL/MA(1)/DO'IPRINT(FA(1,1),3); 51 LINES(1); 52 PFOR;II:STEP:I/UNTIL/MA(1)/DO'IPRINT(FA(1,1),3); 53 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 55 LINES(1); 56 PFOR;II:STEP:I/UNTIL/MA(2)/DO'IPRINT(FA(2,1),3); 57 CINES(1); 58 PFOR;II:STEP:I/UNTIL/MA(2)/DO'IPRINT(FA(2,1),3); 59 LINES(1); 59 LINES(1); 50 PFOR;II:STEP:I/UNTIL/MA(2)/DO'IPRINT(FA(2,1),3); 51 LINES(1); 53 LINES(1); 54 CINES(1); 55 LINES(1); 55 LINES(1); 56 PFOR;II:STEP:I/UNTIL/MA(2)/DO'IPRINT(FA(2,1),3); 57 LINES(1); 58 PFOR;II:STEP:I/UNTIL/MA(2)/DO'IPRINT(FA(2,1),3); 59 LINES(4); 59 LINES(4); 50 PIENE; 50 PIENE; 50 PIENE; 50 PIENE; 51 LINES(4); 52 PIENE; 53 LINES(4); 54 PIENE; 54 PIENE; 55 PIENE; 55 PIENE; 55 PIENE; 55 PIENE; 56 PIENE; 57 PIENE; 5</pre>		35	NMA(1)=NA;NMA(2)=NB;NMA(3)=NC;NMA(4)=ND;
<pre>37 IX(1,1)=011X(2,2)=11X(1,3)=21X(1,2)=31IX(1,3)=41IX(2,3)=51 38</pre>			NMB(1)=0;NMB(2)=NMA(1)JNMB(3)=NMA(1)+NMA(2);
<pre>38</pre>		• 37	1×(1,1)=0;1×(2,2)=1;1×(3,3)=2;1×(1,2)=3;1×(1,3)=4;1×(2,3)=5;
<pre>39 M=2*M; 40 JA=JB=JC=M4 41 ININT(AA (1)); 42 *FOR*I=1*STEP*I*UNTIL*MA(1)*D0*ININT(FA(1,1)); 43 *FOR*I=1*STEP*I*UNTIL*MA(1)*D0*ININT(FA(2,1)); 44 ININT(HA(2)); 45 * FOR*I= 1*STEP*I*UNTIL*MA(2)*D0*ININT(FA(2,1)); 46 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*ININT(FA(2,1)); 47 LINES(4); 48 IPRINT(MA(1),3); 49 LINES(1); 50 *FOR*I=1*STEP*I*UNTIL*MA(1)*D0*IPRINT(FA(1,1)*3); 51 LINES(1); 52 *FOR*I=1*STEP*I*UNTIL*MA(1)*D0*IPRINT(FA(1,1)*3); 53 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 56 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 57 LINES(1); 58 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 59 LINES(1); 50 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 51 LINES(1); 52 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 53 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 56 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 57 LINES(1); 58 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 59 LINES(1); 50 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 51 LINES(1); 52 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 53 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 55 LINES(1); 56 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 57 LINES(1); 58 *FOR*I=1*STEP*I*UNTIL*MA(2)*D0*IPRINT(FA(2,1)*3); 59 LINES(4); 60 CLEAR; 61 CMAIN(2); 62 *END*</pre>		38	For, I=1'5, Do', Lou, 1=1+1, 2LEb, 1, ANULTIT, 3, DO, IX(1'1)=IX(1'1')
<pre>40</pre>		39	
<pre>41 INT(MA (1)); 42 *FOR* [=1'STEP*]*UNTIL*MA( 1)*DO*ININT(FA(1*1)); 43 *FOR* [=1'STEP*]*UNTIL*MA( 1)*DO*ININT(GA(1*1)); 44 ININT(MA(2)); 45 * FOR* [= 1'STEP*]*UNTIL*MA(2)*DO*ININT(GA(2*1)); 46 *FOR* [= 1'STEP*]*UNTIL*MA(2)*DO*ININT(GA(2*1)); 47 LINES(4); 49 LINES(1); 49 LINES(1); 50 *FOR* [=1'STEP*]*UNTIL*MA(1)*DO*IPRINT(FA(1*1)*3); 51 LINES(1); 52 *FOR* [=1'STEP*]*UNTIL*MA(1)*DO*IPRINT(FA(1*1)*3); 53 LINES(1); 54 IPRINT(MA(2)*3); 55 LINES(1); 56 *FOR* [=1'STEP*]*UNTIL*MA(2)*DO*IPRINT(FA(2*1)*3); 57 LINES(1); 58 *FOR* [=1'STEP*]*UNTIL*MA(2)*DO*IPRINT(FA(2*1)*3); 59 LINES(1); 59 LINES(1); 50 CLEAR; 61 CHAIN(2); 62 *END*</pre>		물로 1 40	u seren ( JA=JB=JC=M) ( Alter Elektron of Alter Elektron and a standard and Alter Electron standard attraction and a standard attraction and a standard attraction at
<pre>43</pre>		- #1 - #2807211 - 1-41	ININT(MA (1));
<pre>44 ININT(HA(2)); 45 FOR*1= 1*STEP*1*UNTIL*MA(2)*DO*1 NINT(FA(2,1)); 46 FOR*1= 1*STEP*1*UNTIL*MA(2)*DO*1NINT(FA(2,1)); 47 LINES(4); 48 IPRINT(MA(1),3); 49 LINES(1); 50 FOR*1=1*STEP*1*UNTIL*MA(1)*DO*1PRINT(FA(1,1)*3); 51 LINES(1); 52 FOR*1=1*STEP*1*UNTIL*MA(1)*DO*1PRINT(GA(1,1)*3); 53 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 56 FOR*1=1*STEP*1*UNTIL*MA(2)*DO*1PRINT(FA(2,1),3); 57 LINES(1); 58 FOR*1=1*STEP*1*UNTIL*MA(2)*DO*1PRINT(FA(2,1)*3); 59 LINES(4); 60 CLEAR; 61 CHAIN(2); 62 *END*</pre>		44	FOR ITISTEPSTONIL MAY 17 DO INTITATION
<pre>45</pre>			
<pre>46</pre>		- EV 10 45	FORVIE 1 STEP/1'UNTIL'MA(2)'DO'I NINT(FA(2)I))
<pre>47 LINES(4); 48 [PRINT(MA(1),3); 49 LINES(1); 50 FOR*1=1*STEP*1*UNTIL*MA(1)*D0*IPRINT(FA(1+1)*3); 51 LINES(1); 52 FOR*1=1*STEP*1*UNTIL*MA(1)*D0*IPRINT(GA(1+1)*3); 53 LINES(1); 54 [PRINT(MA(2),3); 55 LINES(1); 56 FOR*1=1*STEP*1*UNTIL*MA(2)*D0*IPRINT(FA(2+1)*3); 57 LINES(1); 58 FOR*1=1*STEP*1*UNTIL*MA(2)*D0*IPRINT(GA(2+1)*3); 59 LINES(4); 60 CLEAR; 61 CHAIN(2); 62 FND*</pre>		ETERSTOR 46	/FOR/1: 1'STEP'1'UNTIL'MA (2)'DO'ININT(GA(2,1));
<pre>48</pre>		47	· LINES(4): 174 5 - 147 - 147 - 1
49 LINES(1); 50 FOR'1=1'STEP'1'UNTIL'MA(1)'DO'IPRINT(FA(1)]'J)]; 51 LINES(1); 53 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 56 FOR'1=1'STEP'1'UNTIL'MA(2)'DO'IPRINT(FA(2,1),3); 57 LINES(1); 58 FOR'1=1'STEP'1'UNTIL'MA(2)'DO'IPRINT(GA(2,1),3); 59 LINES(4); 60 CLEAR; 61 CHAIN(2); 62 FON'		48	PRINT(MA(1),3):
50		49	) LINES(1);
51 LINES(1); 52 /FOR/I=1/STEP/I/UNTIL/MA(1)/DO'IPRINT(GA(1/1)/3); 53 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 56 /FOR/I=1/STEP/I/UNTIL/MA(2)/DO'IPRINT(FA(2/1)/3); 57 LINES(1); 58 /FOR/I=1/STEP/I/UNTIL/MA(2)/DO'IPRINT(GA(2/1)/3); 59 LINES(4); 60 CLEAR; 61 CHAIN(2); 62 /END/		문문 50	FOR 1=1. STEP 1. UNTIL MA(1) DO IPRINT(FA(1,1),3) Moved by the second state of the sec
52 /FOR/I=1/SIEP/I/UNTIL/MA(I)/DU/IFRINT(GA(I)/I//3/) 53 LINES(1); 55 LINES(1); 56 /FOR/I=1/STEP/I/UNTIL/MA(2)/DO/IPRINT(FA(2,1),3); 57 LINES(1); 58 /FOR/I=1/STEP/I/UNTIL/MA(2)/DO/IPRINT(GA(2,1),3); 59 LINES(4); 60 CLEAR; 61 CHAIN(2); 62 /END/		51	LINES(1); Charles and the second s
<pre>&gt;3 LINES(1); 54 IPRINT(MA(2),3); 55 LINES(1); 56 /FOR*1=1*STEP*1*UNTIL*MA(2)*DO*IPRINT(FA(2,1),3); 57 LINES(1); 58 /FOR*1=1*STEP*1*UNTIL*MA(2)*DO*IPRINT(GA(2,1),3); 59 LINES(4); 60 CLEAR; 61 CHAIN(2); 62 /END*</pre>		- <del>2</del> 25. (* ) 52	The start of the s
55       LINES(1);         56       +FOR*1=1*STEP*1*UNTIL*MA(2)*DO*IPRINT(FA(2,1)*3);         57       LINES(1);         58       +FOR*1=1*STEP*1*UNTIL*MA(2)*DO*IPRINT(GA(2*1)*3);         59       LINES(4);         60       CLEAR;         61       CHAIN(2);         42       +END*		- B.1.	
56 FOR'I=1'STEP'I'UNTIL'MA(2)'DO'IPRINT(FA(2,1),3); 57 LINES(1); 58 FOR'I=1'STEP'I'UNTIL'MA(2)'DO'IPRINT(GA(2,1),3); 59 LINES(4); 60 CLEAR; 61 CHAIN(2); 62 FND'			
57 58 507 1 = 1 + STEP + 1 + UNTIL + MA(2) + DO + IPRINT(GA(2+1)+3); 59 60 60 61 61 61 62 7 END + 62 62 7 END +		50	FOR/1=1/STEP/1/UNT1L/MA(2)/DO/IPRINT(FA(2,1),3)
58 FOR*1±1*STEP*1*UNT1L*MA(2)*DO*IPRINT(GA(2*1)*3); 59 LINES(4); 60 CLEAR; 61 CHAIN(2); *END* *	-	57	LINES(1);
59 60 CLEARJ 61 CHAIN(2); 62 PEND •	••	- gro p io 76 <b>* 58</b>	FOR ITISSTEP'I' UNTIL' MA(2)'DO'IPRINT(GA(2)I) 3); And
60 CLEARJ 61 CHAIN(2)J • •		-	LINES(4); second s
<pre>61 CHAIN(2); // Particular and part and and and and and and and and and and</pre>		60	D C CLEARI CONTRACTOR STATE AND A CONTRACTOR STATE AND A CONTRACTOR AND A C
<ul> <li>A second s</li></ul>		- TT	CHAIN(2); CHAIN(2); Chain(Chain) and Chain
		- E	2 - La Contendada en la co
		1 A 7	· "你们,我们们们都是我们就是我们的,我们们的你们,我们们的你们,我们们的你们。"● 我们们们们的你们的,我们们们们就是我们的你们,我们就不能能能。" "我们们就是我们就是我们们们们们们们们们们们们们们们们们们们们们们们们们们们们们
에는 것 같은 것 같은 것 같은 것 같은 바깥에 가지 않는 것 같은 것 같		* *	이 사실 수 있다. 이 가지 않는 것은 이 가지 않는 것은 것을 하는 것을 하는 것을 하는 것을 가지 않는 것을 하는 것을 같은 것은 것은 것을 하는 것은 것을 하는 것을 수 있는 것을 수 있는 것을 하는 것을 하는 것을 하는 것을 하는 것을 하는 것을 하는 것

5.3.

•

	,我们们就是你有这些人,我们就是你的你,我们就是你们的你的?""你们,我们就是你的你的你,我们就是你的你,我们不是你的你的?""你们,你们不是你们的?""你们,你 你们,你们们我们们就是你们就是你们的你?""你们,你们就是你们的是我们的是我们的是你的你的,我们就是你们我们就不是你的你们的?""你们,你们们们们们们,你们们们们
1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	
1	
1 A	
- 11 J 64	
	an ange shipping PUBLIC AL, BL, AI, BI, KS, HT2, ETA, ETA2, ETA3, HT0, N, N, JN, JN, JA, JB, JC
2	PUBLIC NA, NB, NC, ND, NN, R, CN(6), CO(6), CNDP(1), CODP(1)
	PUBLIC RM, RN, RX
	United and August PUBLIC SB(200) SS(200) SBDP(1) SSDP(1) And August Aug
1	PUBLIC IX(4,4), FA(2,40), GA(2,40), NMA(4), FAUP(2), IXDP(2), GAUP(2)
	Distance of the PUBLIC MAC() MADP() I and a set desire in solid data in the second
1	CALLER FUELLE NHOP (1) - CALLER AND
و الأحد المبيد الما درية. م الا	
and an and the second sec	KLAL NO SHE AND A SHE
22.2% (Sec. 14	
الة ( المعمد بتعليل ). 11: >	
• • • • • • • • • • • • • • • • • • •	
	INTEGEN MOINTER
16	MAXAR9(2500);
17	/ BEGIN/
	REAL AX, BX; CALLER CONTRACTOR CONT
19	· INTEGER·I·J·K·H ·C C·MX;
20	PROCEDURE'SHIFTH(B); ARRAYEBI
	BEGIN, BEGIN
21	2 cl / INTEGER / I · J · K · H · CF · CG · CH · CJ · CK 3 <u>cl / 20 cl / </u>
2;	J /ARRAY/A(1 ² • ² RX, 1 ² • ² NN);
	RAFD (A) 3) HX ) 3
2	CF=0;CG=0;CH=0;CJ=0;CK=0;
2	FOR KEISSEPTI UNTIL MA(HX+1) DO BEGIN
	CH=CF+FA(HA+1)KJ JCJ=CG+GA(HA+1)KJJCK=CH+GA(HA+1)KJJ
100000	
	For J=1'STFP'1'UNT11'RX'DO'B(J/1-CH+CG)=A(J/1)'
3	cF=CK:cG=CJi
3	3 / END / :
3	4. WATD(A, 3, 3);
3	5 / END';
	and the second

	36 1. NAB=NA+NB:
i i i i i i i i i i i i i i i i i i i	37 BEGIN'TINTEGER'IL'ARRAYTA(1'+'R+1'+R)
	38 FOR 120, 12, 13, 1, 2, 14, 3, 15, 4, 16, 17, 5, 6, 18, 19, 7, 8, 20, 9, 21, 10, 22, 23, 11
	_39 OO' BEGIN
- <u>1987 - 19</u> 81 - 19	41 KATDIA.1.1):
	42 'END'; 'END';
	43 READAR(85, \$\$, ? ( 'SSFL ? ) ' ) ;
	44
	45 'BEGIN' ARRAY'A, B(I', 'RX, I', 'NN);
	PAGE READAR(85)A)*(*FLBU)*33
	$48 \qquad FMATPP(CA, RM, NN) 4) = FMATPP(B, RM, NN) 4) =$
	49 WATD(A) 3, 0):
	50 WATD(B, 3, 1);
a second a second	51
	52 FOR HX=0, 1, DO, BEGIN, and the termination of the first state of the second state
	53 BESIN
	24 CARNAY D(1) TRUET RULE TRUET TANDA TANAN AND A TANÀNA ANA ANA ANA ANA ANA ANA ANA ANA AN
	56 / BFGIN' ARRAY'A(1' • ' RM / 1' • ' RM )]
	57 / ARRAY C (1' + RH) I' + RH) I
· · · · · · · · · · · · · · · · · · ·	53 MATCOP(C, B, RH, RH): Martin Mar
	59 HATINV(A, D, RH) INATCOP(B, A, RM, RM)
en transformer og som en s	60 NATMUL(A, B) C) RH, RM) JENATPR(A, KM, RM, 4) J
	61
	-νε «ματροπορηγητη αναλαματικά του
<u>,</u>	64 'BEGIN''ARRAY'B(1' * 'RM)1' * 'NN)1
	65 RAFD(B,3,3);
	66
•• • • • • • •	67, p. 18 RAFD(A,3,41). Lug statistickersnikeling kreist er ver statisticker Hänsensliker Berliker Lug pros
🖌 ya set i ke	63
and a second	69 C([,J)=0; 'FOR'K=1'SIEP-1'UNIL'RM'DU'
1	
	72 'END': 'END':
	73 'END'I
i sur statist	74 END'S action were a many and an an an an an an an and a strong were and an an an and a strong to a
이는 소리에서 이	175 yr yn BEGIN TRY RY R
	76 PROCEDURE FULL(EF,R); ARRAY EF; INTEGER R;
	177 Server ABEGIN/YINI BEKYINJA Memori da servera da servera da servera da servera da servera da servera da se
121.111.22	80 · END · :
	81 PROCEDURE VECCOP(SA, SS, X); ARRAY SA, SSJ INTEGER X)
	182BEGIN**INTEGER*II*
بنبية المحاسبين ال	83 / FOR/I=1/STEP/1/UNTIL/X/DU'SA(I)=SS(I))/END'3
	84 PRUCEDURE' SHITTIGAJKAJNNJKMJKNJJ'ARKAT'AJ INTEGEK KAJNNJKMJKNJ
و به شو بدر سینٹسی آراز ان ا	
· · · · · · · · · · · · · · · · · · ·	67 / ARRAY B(1', 'RN, 1', 'RM);
	88 CF=0;CG=0;CH=0;CK=0;
	89 FOR'K=1'STEP'I'UNTIL'HA(HX+1)'DO''BEGIN
	[90]CH=CF+FA(HX+1,K);CJ=CG+G_A(HX+1,K);CK=CH+GA(HX+1,K);
in Line -	191 Hor' FOR' I=CF+1' STEP' I' UNTIL' CH' DO
	_ 92 /FOR'Jz1'STEP'I'UNTIL'RX'DO'A(J)I-CG)=A(J)I}
1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	93 // OK/I=CH+I/SIEP/I/UNIL/CK/D0/ 04 // COR/I=CH+I/SIEP/I/UNIL/CK/D0/
· · · · · · · · · · · · · · · · · · ·	24
	96 · FRDI:
	97 /FOR'1=RN+1'STEP'1'UNTIL'NN'DO'
a partina anti-a	98 /FOR'J=1'STEP'1'UNTIL'RX'DO'A(J,1)=B(J,1=RN);
	e 99 . H. A. PENDY: C. A. D. B. C. M. B.
	100 PROCEDURE' SHIFTY(A, H); 'ARRAY'A; 'INTEGER'H;
المنتخب والمحيد	101 · BEGIN' BEGIN'
- 	102 INTEGER'I, J,K,CF,CG,CH,CJ,CK;
	103 /ARRAY/B(I'+'RM/I'+'R/J second in the second se
	LCTUALGEUACH-UALA-UALA-UALA-UA
•	106 CH=CF+FA(HX+1)K) JCJ=CG+G A(HX+1)K)JCK=CH+GA(HX+1)K)1
يەر چەرمى بىمەرىيە . مەر	107 /FOR'1=CF+1'STEP'1'UNTIL'CH'00'
• • • •	108 /FOR'J=1'STEP'1'UNTI_'NMA(H)'DO'A(1-CG)J)=A(1)J}
	109 /FOR'1=CH+1'STEP'1'UNT1L'CK'DO'
e to a good a	110 /FOR' J=1' STEP' 1' UNTIL' NMA(H)' DO' B(I-CH+CG, J)=A(I, J):
	111 CF=CK;CG=CJ;
	113 FORPINDUALISTEDIIPUNTICANNICAL
	114 FORMULT VITE TO STEP 1/ UNTIL TONT LET MY DUT AND
	115 /END'1
	116 MONITO;
	117 /FOR'HX=0,1/DO'/BEGIN/
	118 . IF'HX'EQ'O'THEN'VECCOP(SA,SS,NN)'ELSE'VECCOP(SA,SB,NN)3
العديدية : محمد بالبيد العرب :	119 BEGIN' BEGIN'
	120 PINFEGERJKAJKJLJ'ARRAY'C(1'+'NN)1'+'R)J
	122 FOR/KX20/J'DU''BEGIN/
<b></b>	123 /FOR/1+1.2/3/00//BEGIN/
· · · · · · · · · · · · · · · · · · ·	「金田子」「「「「「金田子」」「金田子子」「金田子」「金田子」「「「「金田」」」「「金田」」」「「金田」」「金田子子子」「「「金田」」「「金田」」」」「「金田」」」

										1999 - A.	148.
			•		•		•				
	•										
•											
		125	· · · · ·	RAFDLASIS	12+HX+2+	KX+IX(K)	.)); FI SE//1E/I	FOT KT THE	N •		an a the second s
		127	ا مالغد بنده . این محمد از در	#BEGINATE	ANS(B)A)	R.R. HATC	OP (A.B.R.I	R); PEND ;	· · · · · · · · · · · · · · · · · · ·	·····	
	بالمتصححة	128	alle o	*FoR*1=1*	STEPIIIU	NTIL'NHA	L) DO' FOR	RAJE1 STEP	'I'UNTIL'	NMA (K) DO	
	-	130		FENDAT			- NHO(L)/-3/			بدارمتها ففتقك وور	
		131		SHIFTV(C.	K);						an a
	·	132		#BEGIN##R	LAL SUMI	*INTEGER: **R)*B()	1,JJ	RN);		ali di setta di secono di secon	n shing in the second
		134		RAFD(R.2.	HX33						-
		135		*FOR*1=1*	SIEPALAU	INTIL'RN'C	00"FOR"J=; [P/1/UNTT1]	INNEDO	NTIL'NMAG	K) DO BE	GIN"
	• · · <del>- ·</del>	137	• • • •	SUH=C(1)3	(L-RN+1)	+C(L,J))	(I-J)=SUM	FEND -			
	~	138		WATD CA. 3.	KX+K-1);		nang artista ang disalikan di salam da kan -			tag i agama ara ann ann feir i.	. An
		139.		/END/:				· · · · · · · · · · · · · · · · · · ·			a an
		.141		PEND							
		142		PENDI: and							. <del></del>
	· · · · · · · ·	144	· · ·	REAL SUN	;;						n kan manan kan kan kan kan kan kan kan kan kan
		145		/INTEGER/	KI:	DBANAN					
	مورید بر <u>مشید ،</u>	147	a literation,	/FU ^{R®} KI=C	STEPII	UNTILALL	O BEGIN	NEN / / Thurs (2772)	,	. <u>Attan</u> an ita	in a final af a la companya da serie de la companya da serie de la companya da serie de la companya da serie d
		148		"BEGIN"	RRAY CLI	** RN . 1 .	• ? R } }	line de la service	a shi ta ka i	an thair an a	an a
	t in a second for	149	· · · · ·	*F0 ^{K*} K=0;	. 3*K 1+K 13	BFCIN	द्वार्य-स्टब्स्ट्रियः				د مېر در مېرمانو و اور ماردو درېسو وي. • •
		151		'FOR' 1=1	STEP-1-1	NTIL'RN'	00''FOR'J=	1'STEP-1-0	NTIL'NMA	(K+1)+D0+	
		152		ACIJUANNE	s(K+1))=( 		لأعدعت فتعدفه	• "	بمعدية بمعديات	alan seriet	<ul> <li>A set and a set and a set of the set of th</li></ul>
		155	· · ·	SHIFTH(A	, RN, NN, RI	4, RN ) ;		nije se			
	• •	155		BEGIN"	ARRAY'B(	1***RM+1*	**RN3+C(1*	.* RN#1***F	N ) ;		جورج بالمحتم أحبين
		. 150	¥يمور تاريخ ب	RAFD(B+2)	, HX)1	JNTIL PRNP	DOPPFOR'Ja	1+STEP+1+4	NTIL'RNA	DOPPBEGIN ⁴	9 48 9 14 9 11 - 12 4 CR 2 1 - 1 1
	nen anderen er	158		SUMPACIA	J) FORI	=RN+1"ST	EP 1'UNTI	LPAN DO			(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
	· •	.159		SUM=SUM+/	A(I,K)*8) /STFP/=1	(K=RNJJ); 'UN stiltr	C(],J)=SUM N=1'00''FO	]/LND/] R/J±1+1/S1	EP I UNT	11 * RN/ DO*-	
		161		C(J+1)=C	(1+1):						
	<u></u>	162-	*2444-27	HAXARR(5)	000) <u>;</u>			<u> </u>			
	an a	164	 	MAXARR(2)	500)::	••••••••••••••••••••••••••••••••••••••					anda ana ana ana ana ang manakana damatana ata da
		165	سسب سرأ دو	-*END*;							
	. 1	160	in i sent net e nen	*END*]							
		168		·END ;	a ang ang ang ang ang ang ang ang ang an				a de <b>Tim</b> te de la consta	aliene ra solili	se tala a can bara
	•	169	· · · · ···	/END/; 	an a					a gant canana lu	
		171		CHAIN(5)	;	····		· · · · · · · · ·			
		172		PEND?		unti Allinadi. N			ireall marada		arata data suttata sa di ta
					•					•	· · · ·
		•									•
									,		
	•				•						: .
											· •

#### APFFUDIX 5. INFUT DATA FOR COMPUTER FROCRAMS

The following input data is required for the computer programs described in this thesis.

11-730

 M = Order of torms in displacements. Must be an odd number for symmetry solutions.

2. N = Order of 
$$\beta$$
 terms in displacements.

- 3. BI = Total twist in radians. (Not required for Cylindrical Shell)
- 4. AL = Width of Shell
- 5. BL = Length of Shell
- 6. ETA = Poisson's Ratio
- 7. HT = Thickness of Shell
- 8. KS = Transverse Shear Factor = 1.0 in all cases
- 9. E = Young's Modulus

10. RO = Density

- 11. RD = Number of Modes Required
- 12. NX = Number of sections in  $\ll$  direction required in mode shapes print out.
  - 13. NT = Number of sections in  $\beta$  direction required in mode shapes print out.

APF	INDIX 6. LIBRARY PROCEDURES USED IN COMPUTER PROGRAM							
1.	INREAL (X, Y, Z,)							
	Reads numbers from data into real locations X, Y, Z, etc.							
2.	ININT (I, J, K)							
•	Reads numbers from data into integer locations I, J, K, etc.							
3.	EPRINT (X, I)							
	Prints real value X to I decimal places.							
4.	IPRINT (M, I)							
	Prints integer value M to I significant figures.							
5.	LINES (I)							
	Outputs I line feeds.							
6.	SPACES (I)							
	Output I space characters.							
7.	TEXT ( '('S'TRING')')							
tan Tan Tan	Outputs characters STRING							
8.	EMATFR (A, M, N, I)							
	Outputs real array A(1:M, 1:N) row by row to I decimal places							
9.	EVECFR (A, M, I)							
	Outputs real array A(1:M) to I decimal places.							
10.	WATD (A, I, J)							
	Writes array A to position J on disc unit I							
11.	RAFD (A, I, J)							
	Reads array A from position J on disc unit I.							
12.	MAXARR (M)							
-	Defines maximum size of arrays to be used in WATD and RAFD in order to							
	define size of each area J on the disc.							
13.	WRITEA (I, A, '('NAME')')							
	Writes array A to present write position of magnetic tope unit I, and gives							
	it the identifier NAME.							
14.	READAR (I, A, '('NAME')')							
	Reads next array with identifier NAME on magnetic tape unit I into array A.							

15. CLOSHT (I)

Reallocates magnetic tape unit I

16. MONITO

Cutputs computer processor time used by program, and real time.

17. STOP

Clears output buffers and stops execution of the program.

18. CLEAR

Clears output buffers

19. CHAIN (I)

Jumps from present chain to CHAIN I.

As the program is written in Algol, but the intermodular transfer of information is by way of Fortran PUBLIC variables, extra information is required to define the lower bounds of all arrays. This is performed in the program by the following routines.

20. SUPDV1 (A, ADOPE, M)

Sets up dope vector ADOPE (1:1) for array A(1:M)

21. SUPDV2 (A, ADOPE, M, N)

Sets up dope vector ADOPE (1:2) for array A(1:M, 1:N)

22. SUPDW3 (A, M, N, P, Q, I, J)

Sets up information in last three locations of array A(M:N, P:Q, I:J). Thus in the FUBLIC declaration the array A must be declared 3 elements larger than actually required.

These are the procedures used for this purpose in the program. Routines exist to perform the same operations for arrays of dimensions one to six. In the following procedures the notation A(M, N) refers to an array of size A(1:M, 1:N)

23. NULL (A, M, N)

Nulls matrix A(M,N)

24. TRANS (A, B, M, N)  
$$A(M, N) = B^{T}(N, M)$$

25. MATHUL (A, B, C, M, N, P)  $A(H, P) = B(H, N)^* C(N, P)$ 

28. MATDIV (A, B, M, N)  
A(M, N) = 
$$B^{-1}(M, M) * A(M, N)$$

Calculates eigenvalues of symmetric array A(M, M) by Householder's method. Stores eigenvalues in array E(M). If  $K \neq 2$  it stores eigenvectors in rows of A, calculating these by inverse iteration.

Calculates eigenvalues and eigenvectors of array A(M, M) by HQR method and inverse iteration. It stores real and imaginary parts of eigenvalues in  $ER(\overline{M})$  and EI(M) respectively, and real and imaginary parts of eigenvectors in rows of A(M, M) and B(M, M) respectively. The integers X and L define normalisation of real and complex eigenvectors respectively as follows

- 1 The Manhattan Norm
- 2 The Euclidean Norm
- 3 The Maximum Norm